A Healthy Toronto By Design Report **Road to Health:**Improving Walking and Cycling in Toronto

April 2012

416.338.7600 toronto.ca/health

M TORONTO Public Health

Acknowledgements

Reference

Toronto Public Health, Road to Health: Improving Walking and Cycling in Toronto. April 2012.

Prepared by:

IndEco Strategic Consulting Inc. in collaboration with Toronto Public Health and City of Toronto, Transportation Services

Prepared For:

Toronto Public Health, Healthy Public Policy Directorate

Project Team:

Deborah Lightman, IndEco Meghan Winters, Simon Fraser University David Heeney, IndEco Carol Mee, Toronto Public Health Shawn Chirrey, Toronto Public Health Monica Campbell, Toronto Public Health Rosie Mishaiel, Toronto Public Health

Project Advisory Committee:

Kathryn MacKay, Ontario Medical Association Matthew Mayer, Heart and Stroke Foundation of Ontario Catriona Delaney, YMCA of Greater Toronto Christina Bouchard, Transportation Services, City of Toronto, Janet Lo, Transportation Services, City of Toronto

Additional Advisors:

Fiona Chapman, Transportation Services, City of Toronto Daniel Egan, Transportation Services, City of Toronto Linda Ferguson, Toronto Public Health Rita Paul-Sen Gupta, Toronto Public Health Nancy Smith Lea and Ryan Whitney, Toronto Centre for Active Transportation Jackie Leroux, Toronto Public Health Ronald Macfarlane, Toronto Public Health

Toronto Public Health Healthy Public Policy 277 Victoria St., 7th Floor Toronto, Ontario, Canada M5B 1W2 (416)392-6788 E-mail: publichealth@toronto.ca

Report available at: http://www.toronto.ca/health

April 2012

About the Healthy Toronto By Design Report Series

Healthy Toronto By Design was released by Toronto Public Health in October 2011 and was the first in a series of reports on how local communities shape the health of their residents. The report noted that healthy cities are cities that are liveable, prosperous and sustainable. They are cities with high quality built and natural environments, public transit, housing, culture, education, food and health care. Healthy cities don't just happen. They result from creative vision, strategic decision-making and thoughtful implementation that respects the needs and challenges of all residents. They happen by design – through intentional investment and provision of infrastructure, programs and services with health in mind.

This report is one of a series which explore what makes a healthy city. Visit Toronto Public Health's website at http://www.toronto.ca/health for a list of reports in the series. Some of the topic areas in the series include the following:

- Road to Health this report synthesizes evidence on health benefits and risks associated with walking, cycling and physical activity related to the use of public transit, as well as economic assessments and specific strategies to increase the use and safety of active transportation in Toronto.
- The Walkable City this report summarizes the findings of a Residential Preferences Survey that gauges public demand for walkable versus more auto-oriented neighbourhoods, and links this information with travel choices, physical activity levels and body weight.
- Creating Healthy Built Environments Highlights of Best Practices in Toronto this report showcases examples of innovative practices and policies across city government in Toronto that promote healthy built environments.
- Enabling Healthier Neighbourhoods through Land Use Planning this report synthesizes zoning barriers and opportunities to promote healthy neighbourhoods, particularly in clusters of residential apartment towers in low income areas and inner suburbs of Toronto.
- Health Impact Assessment Software Tool a software tool has been developed to assist
 policy and decision-makers understand how different approaches to neighbourhood
 design might impact health-related outcomes such as physical activity levels, body
 weight and greenhouse gas emissions. A technical report synthesizes information on
 the development of the tool and results of pilot testing.

Table of Contents

Introduction: The critical nexus of transportation and health.	Executive Summary	4
Chapter 1. Active transportation: A review of the evidence 10 Active transportation as a source of physical activity 10 Individual health benefits of physical activity from active transportation 11 Population health benefits of reduced motor vehicle travel 14 Comparing health benefits, risks and costs 15 Transportation system benefits. 16 Social, economic, and environmental benefits 18 Chapter 2. Active transportation in Toronto 21 Walking and cycling in the City of Toronto. 21 Quantifying the health benefits of active transportation in Toronto 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto 26 The value of active transportation in Toronto 26 The value of increasing active transportation in Toronto 28 Chapter 3. Health risks of walking and cycling in Toronto 28 Collisions, injuries, and fatalities 32 Collisions, injuries, and fatalities 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Tor	Introduction: The critical nexus of transportation and health	8
Active transportation as a source of physical activity 10 Individual health benefits of physical activity from active transportation 11 Population health benefits of physical activity from active transportation 13 Population health benefits, risks and costs 14 Comparing health benefits, risks and costs 15 Transportation system benefits. 16 Social, economic, and environmental benefits. 18 Chapter 2. Active transportation in Toronto 21 Walking and cycling in the City of Toronto. 21 Quantifying the health benefits of active transportation in Toronto 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto 26 The value of active transportation in Toronto 26 The value of active transportation in Toronto 28 Collisions, injuries, and fatalities 32 Collisions, injuries and fatalities 32 Collisions, injuries, and fatalities 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation in Toronto 47	Chapter 1. Active transportation: A review of the evidence	.10
Individual health benefits of physical activity from active transportation 11 Population health benefits of physical activity from active transportation 13 Population health benefits, risks and costs 15 Transportation system benefits 16 Social, economic, and environmental benefits 18 Chapter 2. Active transportation in Toronto 21 Walking and cycling in the City of Toronto. 21 Quantifying the health benefits of active transportation in Toronto 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto 26 The value of active transportation in Toronto 28 Chapter 3. Health risks of walking and cycling in Toronto 28 Collisions, injuries, and fatalities 32 Costs of collisions, injuries, and fatalities 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Toronto's actions on active transportation 47 Making active transportation more attractive	Active transportation as a source of physical activity	.10
Population health benefits of physical activity from active transportation. 13 Population health benefits of reduced motor vehicle travel 14 Comparing health benefits, risks and costs 15 Transportation system benefits 16 Social, economic, and environmental benefits 18 Chapter 2. Active transportation in Toronto 21 Walking and cycling in the City of Toronto 21 Quantifying the health benefits of active transportation in Toronto 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto 26 The value of active transportation in Toronto 28 Chapter 3. Health risks of walking and cycling in Toronto 28 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 32 Costs of collisions, injuries, and fatalities 39 Walking and cycling in Toronto's core versus suburbs. 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Improving active transportation safer 49 Making active transportation more accessible 58 </td <td>Individual health benefits of physical activity from active transportation</td> <td>.11</td>	Individual health benefits of physical activity from active transportation	.11
Population health benefits of reduced motor vehicle travel 14 Comparing health benefits, risks and costs 15 Transportation system benefits 16 Social, economic, and environmental benefits 18 Chapter 2. Active transportation in Toronto 21 Walking and cycling in the City of Toronto 21 Quantifying the health benefits of active transportation in Toronto 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto 26 The value of active transportation in Toronto 26 The value of increasing active transportation in Toronto 28 Chapter 3. Health risks of walking and cycling in Toronto 32 Collisions, injuries and fatalities 32 Collsions, injuries, and fatalities 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation more attractive 58 Enabling active transportation more attractive 58 <t< td=""><td>Population health benefits of physical activity from active transportation</td><td>.13</td></t<>	Population health benefits of physical activity from active transportation	.13
Comparing nearing benefits, fisks and costs 15 Transportation system benefits 16 Social, economic, and environmental benefits. 18 Chapter 2. Active transportation in Toronto 21 Walking and cycling in the City of Toronto. 21 Quantifying the health benefits of active transportation in Toronto 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto 26 The value of increasing active transportation in Toronto 26 The value of increasing active transportation in Toronto 28 Chapter 3. Health risks of walking and cycling in Toronto 28 Collisions, injuries, and fatalities 32 Costs of collisions, injuries, and fatalities 32 Costs of collisions, injuries, and fatalities 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation more attractive 58 Enabling active transportation more attractive 58 Enabling active transportation more accessible 63	Population health benefits of reduced motor vehicle travel	.14
Social, economic, and environmental benefits. 18 Chapter 2. Active transportation in Toronto 21 Walking and cycling in the City of Toronto. 21 Quantifying the health benefits of active transportation in Toronto. 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto. 26 The value of active transportation in Toronto. 26 The value of increasing active transportation in Toronto. 28 Chapter 3. Health risks of walking and cycling in Toronto. 28 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 32 Costs of collisions in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation more attractive 58 Enabling active transportation more accessible 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70	Transportation system benefits	.15
Chapter 2. Active transportation in Toronto 21 Walking and cycling in the City of Toronto. 21 Quantifying the health benefits of active transportation in Toronto. 24 Quantifying the health benefits of active transportation in Toronto. 25 Deaths prevented through active transportation in Toronto. 26 The value of active transportation in Toronto. 26 The value of increasing active transportation in Toronto. 28 Chapter 3. Health risks of walking and cycling in Toronto 32 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 35 Chapter 4. Distribution of health benefits and risks 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation more attractive 48 Making active transportation more accessible 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto<	Social. economic. and environmental benefits	.18
Walking and cycling in the City of Toronto. 21 Quantifying the health benefits of active transportation in Toronto. 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto. 26 The value of active transportation in Toronto. 26 The value of increasing active transportation in Toronto. 28 Chapter 3. Health risks of walking and cycling in Toronto. 32 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 32 Costs of collisions, injuries, and fatalities 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation more attractive 58 Enabling active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Collec	Chapter 2. Active transportation in Toronto	
Quantifying the health benefits of active transportation in Toronto 24 Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto 26 The value of active transportation in Toronto 26 The value of increasing active transportation in Toronto 26 Chapter 3. Health risks of walking and cycling in Toronto 32 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 35 Chapter 4. Distribution of health benefits and risks 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation foronto 48 Making active transportation more attractive 58 Enabling active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Collecting	Walking and cycling in the City of Toronto	21
Estimating current levels of walking and cycling 25 Deaths prevented through active transportation in Toronto 26 The value of active transportation in Toronto 26 The value of increasing active transportation in Toronto 28 Chapter 3. Health risks of walking and cycling in Toronto 32 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 32 Costs of collisions in prives, and fatalities 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto 48 Making active transportation safer 49 Making active transportation more attractive 58 Enabling active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Developing plans, policies and standards 71 Collecting better information 72 Enhancing partnerships	Quantifying the health benefits of active transportation in Toronto	24
Deaths prevented through active transportation in Toronto. 26 The value of active transportation in Toronto. 26 The value of increasing active transportation in Toronto. 28 Chapter 3. Health risks of walking and cycling in Toronto 32 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 32 Costs of collisions, injuries, and fatalities 35 Chapter 4. Distribution of health benefits and risks 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto. 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto. 48 Making active transportation safer. 49 Making active transportation more attractive 58 Enabling active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Developing plans, policies and standards 71 Collecting better information 72 Enhancing partnerships	Estimating current levels of walking and cycling	.25
The value of active transportation in Toronto. 26 The value of increasing active transportation in Toronto 28 Chapter 3. Health risks of walking and cycling in Toronto 32 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 35 Chapter 4. Distribution of health benefits and risks 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto 48 Making active transportation more attractive 58 Enabling active transportation through land-use planning 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Tools to improve decision-making 72 Enhancing partnerships 74 Building community support 75	Deaths prevented through active transportation in Toronto	.26
The value of increasing active transportation in Toronto 28 Chapter 3. Health risks of walking and cycling in Toronto 32 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 35 Chapter 4. Distribution of health benefits and risks 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto 48 Making active transportation safer 49 Making active transportation more attractive 58 Enabling active transportation through land-use planning 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Tools to improve decision-making 72 Enhancing partnerships 74 Building community support 75 Seek	The value of active transportation in Toronto	.26
Chapter 3. Health risks of walking and cycling in Toronto 32 Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 35 Chapter 4. Distribution of health benefits and risks 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto 48 Making active transportation nore attractive 58 Enabling active transportation through land-use planning 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Collecting better information 71 Tools to improve decision-making 72 Enhancing partnerships 74 Building community support 75 Seeking funding from other levels of government 75	The value of increasing active transportation in Toronto	.28
Collisions, injuries and fatalities 32 Costs of collisions, injuries, and fatalities 35 Chapter 4. Distribution of health benefits and risks 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto 48 Making active transportation more attractive 58 Enabling active transportation through land-use planning 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Collecting better information 71 Tools to improve decision-making 72 Enhancing partnerships 74 Building community support 75 Seeking funding from other levels of government 75	Chapter 3. Health risks of walking and cycling in Toronto	.32
Costs of collisions, injuries, and fatalities35Chapter 4. Distribution of health benefits and risks39Walking and cycling in Toronto's core versus suburbs39Active transportation and equity in Toronto42Chapter 5. Strategies for improving active transportation in Toronto47Toronto's actions on active transportation47Improving active transportation in Toronto48Making active transportation safer49Making active transportation more attractive58Enabling active transportation through land-use planning63Making active transportation more accessible65Implementing coordinated packages of actions67Chapter 6. Facilitating effective action in Toronto70Setting goals70Developing plans, policies and standards71Collecting better information71Tools to improve decision-making72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Collisions, injuries and fatalities	.32
Chapter 4. Distribution of health benefits and risks 39 Walking and cycling in Toronto's core versus suburbs 39 Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto 48 Making active transportation safer. 49 Making active transportation more attractive 58 Enabling active transportation through land-use planning 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Collecting better information 71 Tools to improve decision-making 72 Enhancing partnerships 74 Building community support 75 Seeking funding from other levels of government 75	Costs of collisions, injuries, and fatalities	.35
Walking and cycling in Toronto's core versus suburbs.39Active transportation and equity in Toronto42Chapter 5. Strategies for improving active transportation in Toronto47Toronto's actions on active transportation47Improving active transportation in Toronto48Making active transportation safer.49Making active transportation more attractive58Enabling active transportation through land-use planning63Making active transportation more accessible65Implementing coordinated packages of actions67Chapter 6. Facilitating effective action in Toronto70Setting goals70Developing plans, policies and standards.71Tools to improve decision-making72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Chapter 4. Distribution of health benefits and risks	.39
Active transportation and equity in Toronto 42 Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto 48 Making active transportation safer. 49 Making active transportation more attractive 58 Enabling active transportation through land-use planning 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Tools to improve decision-making 72 Enhancing partnerships 74 Building community support 75 Seeking funding from other levels of government 75	Walking and cycling in Toronto's core versus suburbs	.39
Chapter 5. Strategies for improving active transportation in Toronto 47 Toronto's actions on active transportation 47 Improving active transportation in Toronto. 48 Making active transportation safer. 49 Making active transportation more attractive 58 Enabling active transportation more attractive 58 Enabling active transportation more accessible 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Collecting better information 72 Enhancing partnerships 74 Building community support 75 Seeking funding from other levels of government 75	Active transportation and equity in Toronto	.42
Toronto's actions on active transportation47Improving active transportation in Toronto.48Making active transportation safer.49Making active transportation more attractive58Enabling active transportation through land-use planning63Making active transportation more accessible65Implementing coordinated packages of actions67Chapter 6. Facilitating effective action in Toronto70Setting goals70Developing plans, policies and standards.71Collecting better information72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Chapter 5. Strategies for improving active transportation in Toronto	.47
Improving active transportation in Toronto.48Making active transportation safer.49Making active transportation more attractive58Enabling active transportation through land-use planning.63Making active transportation more accessible65Implementing coordinated packages of actions67Chapter 6. Facilitating effective action in Toronto70Setting goals70Developing plans, policies and standards.71Collecting better information71Tools to improve decision-making72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Toronto's actions on active transportation	.47
Making active transportation safer	Improving active transportation in Toronto	.48
Making active transportation more attractive 56 Enabling active transportation through land-use planning 63 Making active transportation more accessible 65 Implementing coordinated packages of actions 67 Chapter 6. Facilitating effective action in Toronto 70 Setting goals 70 Developing plans, policies and standards 71 Collecting better information 71 Tools to improve decision-making 72 Enhancing partnerships 74 Building community support 75 Seeking funding from other levels of government 75	Making active transportation safer.	.49
Making active transportation more accessible .65 Implementing coordinated packages of actions .67 Chapter 6. Facilitating effective action in Toronto .70 Setting goals .70 Developing plans, policies and standards .71 Collecting better information .71 Tools to improve decision-making .72 Enhancing partnerships .74 Building community support .75 Seeking funding from other levels of government .75	Enabling active transportation through land-use planning	.50
Implementing coordinated packages of actions.67Chapter 6. Facilitating effective action in Toronto.70Setting goals	Making active transportation more accessible	65
Chapter 6. Facilitating effective action in Toronto70Setting goals70Developing plans, policies and standards71Collecting better information71Tools to improve decision-making72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Implementing coordinated packages of actions	.67
Setting goals70Developing plans, policies and standards71Collecting better information71Tools to improve decision-making72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Chapter 6. Facilitating effective action in Toronto	.70
Developing plans, policies and standards.71Collecting better information71Tools to improve decision-making72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Setting goals	.70
Collecting better information71Tools to improve decision-making72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Developing plans, policies and standards	.71
Tools to improve decision-making72Enhancing partnerships74Building community support75Seeking funding from other levels of government75	Collecting better information	.71
Enhancing partnerships74 Building community support	Tools to improve decision-making	.72
Seeking funding from other levels of government	Enhancing partnerships	.74
	Building community support	.15 75
Benchmarking other jurisdictions	Benchmarking other jurisdictions	76

Conclusions.		78
References		80
Appendix A.	Methodology for quantifying health benefits	90
Appendix B.	Analysis of types of collisions	96
Appendix C.	Toronto policies and programs	98

Executive Summary

This report focuses on active transportation as a means to improve health and quality of life in Toronto. It reviews the literature on the health impacts of walking and cycling for transportation in urban areas, and also discusses the economic, social, environmental, and transportation system benefits. It presents data on walking and cycling mode shares in the City of Toronto and quantifies the health benefits of active transportation in Toronto. It also analyzes collision data and quantifies the costs of pedestrian and cyclist collisions and injuries in Toronto. Finally, it draws on secondary sources and interviews with municipal staff in other jurisdictions to identify strategies for increasing the safety and use of active transportation across Toronto. The report's conclusions are presented below.

Review of the benefits of active transportation

- Physical activity from active transportation has very important health benefits, including significantly reducing the risk of all-cause mortality, cardiovascular disease, obesity, type II diabetes, and certain types of cancer.
- Increasing the use of active transportation can also generate significant social, environmental, economic and transportation system benefits.
- In North America, users of active transportation generally face greater risks from traffic collisions than users of other modes (such as cars and transit). However, the health benefits experienced by individuals who increase their physical activity through the use of active transportation greatly outweigh the risks.
- Walking and cycling infrastructure investments are extremely cost-effective, even when considering the health benefits alone.
- Better design for active modes, such as walking and cycling, can greatly increase safety for all modes; increasing the proportion of trips made by walking and cycling can also independently lower collision and injury rates (the 'safety in numbers' effect).

The state of active transportation in Toronto

- Toronto's walking and cycling mode shares are increasing, though they continue to lag behind some leading North American cities.
- Surveys suggest that official mode shares understate the number of Torontonians for whom walking and cycling are important modes of transportation. However, it is difficult to determine how many, how often and how far Torontonians walk and cycle based on available data sources.
- Toronto's trip distances suggest that large increases in active transportation mode shares are potentially very feasible. About 55% of all trips in Toronto are less than 7 km, and are therefore very conducive to cycling. Over 20% of all trips are under 2 km and therefore very walkable.
- Of relevance to Toronto, cycling is almost as fast as driving for trips of 7 km in urban areas, and walking is generally as fast as driving for trips of 500 m and less.

Health benefits of active transportation in Toronto

- Higher levels of physical activity through increased cycling and walking can significantly reduce an individual's risk of a number of chronic diseases and prevent deaths. Based on very conservative calculations, 2006 levels of walking and cycling in Toronto are estimated to prevent about 120 deaths each year. Total savings from these prevented deaths range from \$130 million to \$478 million depending on how deaths are valued. Savings in direct medical costs arising from residents staying active by walking and cycling are estimated to provide a further economic benefit of \$110 to \$160 million.
- Achieving walking and cycling commuting mode shares of 12% and 6%, respectively, would prevent about 100 additional deaths each year, yielding additional annual benefits of \$100 million to \$400 million. These increases would bring Toronto to the walking and cycling mode shares in Vancouver and Portland, respectively, and are feasible based on the analysis of trip distances, recognizing that communities in Toronto with higher densities and infrastructure can achieve these increases sooner than others.
- The health benefits of active transportation are not evenly distributed across the city
 – geographically or socioeconomically. Levels of walking and cycling among residents
 of Toronto's core are over three times higher than among residents of the suburbs.
 Toronto's core is also more walkable and bikeable than the suburbs. Since many low income and high-rise neighbourhoods are located in the suburbs, these patterns
 result in transportation and health inequality.

Collisions, injuries and fatalities in Toronto

- While collision rates for pedestrians and cyclists have declined over the last decade, Toronto is still less safe for pedestrians and cyclists than other Canadian cities.
- Collisions involving pedestrians and cyclists in Toronto cost over \$60 million each year. By continuing to improve the safety of active transportation, Toronto could reduce collisions, injuries and fatalities, and generate significant economic benefits.
- Efforts to make walking and cycling safer can start by using collision analysis to identify specific issues and factors related to the frequency of conflicts, near misses and collisions among modes.
- In Toronto, collisions most frequently occur at intersections and on major arterial roads.
- In Toronto 30% of collisions happen mid-block (primarily related to collisions with car doors) and may be linked to inadequate separation of cyclists from motor vehicles.
- Elderly pedestrians are most likely to be killed in collisions with vehicles; children and residents of low-income neighbourhoods may also be particularly at risk of injury when walking and cycling. Speed increases the numbers of collisions, injuries and fatalities of pedestrians and cyclists.

Improving active transportation in Toronto

• Toronto has successfully implemented diverse initiatives that support active transportation, including the Green Standard and the Walking Strategy – both of

which won recognition from the Federation of Canadian Municipalities. In order to further improve active transportation and health, the City of Toronto must continue to make targeted efforts to increase safety for pedestrians and cyclists. The City must also continue to make walking and cycling more attractive and accessible to residents across the city.

- Examples of interventions that have been proven to effectively reduce collision and injury rates for pedestrians and cyclists include: Traffic calming: individual physical interventions to reduce motor vehicle speeds, and areawide speed reduction strategies.
- Traffic signal phasing: leading pedestrian / bicycle phases, and pedestrian / bicycle-only phases.
- Intersection improvements for pedestrians: curb bulbs, medians and improved marking and signage.
- Intersection markings for cyclists: bicycle boxes, solid-line bike lanes approaching intersections, and lanes or markings through intersections.
- Connected sidewalks and paths that buffer pedestrians from traffic.
- Connected facilities that separate cyclists from traffic: separated bike lanes, buffered bike lanes, coloured bike lanes, and off-road bike paths.
- In some parts of Toronto, bicycle facilities are poorly connected and less separated from traffic than in other cities, and Torontonians have identified this as a key factor limiting their use of cycling for transportation. However, developing a viable bikeway network will likely require re-allocation of roadway space from motor vehicles to bicycles. Different strategies may also be needed in Toronto's core and in the suburbs, given the dramatically different land use patterns.
- Many options for improving safety require only road surface coatings or signs, and are relatively inexpensive to implement (e.g. new approaches to bike lanes at intersections). Other improvements may be more expensive, but are needed to keep Toronto on par with other leading cities in North America (e.g. separated bike lanes).
- In other cities, a number of other features in the built environment have also contributed to increases in mode share. These include:
 - o Improvements to pavement quality and snow clearance
 - o Short-cuts and direct walking and cycling routes
 - o Showers and secure bicycle parking at trip destinations
 - Improvements to the convenience of active transportation-transit trips through station design, bicycle parking and signage
 - Streetscape and pathway improvements including lighting and pedestrianfriendly urban design.
- Land use patterns play a crucial role in enabling active transportation, since trip distance is often the limiting factor. In Toronto's re-developing areas, walking and cycling can be supported through mixed use, higher density development with high route connectivity.
- Marketing and education programs also play an important role in overcoming barriers to walking and cycling for transportation.

- Making active transportation accessible to residents of all ages, abilities and incomes may require targeted investments in walking and cycling facilities and programs. It also calls for the development of affordable housing in accessible locations.
- In other cities, interventions in the built environment and in marketing and education have most effectively generated rapid increases in active transportation safety and mode shares.

Facilitating effective action

- In cities including Montreal, Chicago and New York, new goals and targets for active transportation safety and/or mode shares have served as important stimuli for action.
- Toronto and other cities have also adopted policies and/or standards that ensure that walking and cycling are considered in land use, roadway and facility planning.
- Cities including Toronto and Vancouver have recognized the importance of improving the quality of data on walking and cycling.
- Cities including Portland have also adopted transportation planning tools that are specifically designed to assess latent demand for active transportation and to evaluate the need for improvements to walking and cycling environments.
- In a number of cities, public health departments are working closely with the transportation, planning and other municipal departments to achieve integrated, coordinated action on active transportation.
- Involving community stakeholders in planning and decision-making may help to generate widespread support for progressive actions.
- Many provincial, state and federal governments have developed programs to fund improvements in walking, cycling and public health, as the economic benefits of active transportation are shared across levels of government.

The City of Toronto has articulated a commitment to supporting safe active transportation and has made progress towards this objective. However, Toronto's walking and cycling safety, infrastructure and mode shares lag behind other leading North American cities. Toronto should continue to use best practices and benchmark other leading cities that have successfully improved quality of life by enabling safe active transportation.



Introduction: The critical nexus of transportation and health

Transportation and public health have historically been addressed separately by planners and policymakers in Canada. However, it is now recognized that current land use planning and automobile-oriented transportation systems are closely linked to Canadians' low levels of daily physical activity (Canadian Institute for Health Information 2006). Physical inactivity and obesity are in turn generating increased levels of chronic disease, rising healthcare costs and reduced quality of life.

These problems are particularly notable in Toronto, where only 42% of adults are physically active in their leisure-time (Toronto Public Health, 2011). Adults in Toronto aged 20 or older have the second lowest levels of physical activity compared to adults surveyed in 35 other health units throughout Ontario (Ontario Ministry of Health and Long-Term Care 2009). Furthermore, 4 in 10 adults and 22% of adolescents aged 12 to 17 are overweight or obese. Inactivity and obesity are also contributing to increasing levels of chronic disease in Toronto. Overall, 29% of Torontonians have been diagnosed with major chronic conditions, up from 23% in 2001 (Toronto Public Health 2010).

In response to these issues, public health practitioners, urban planners and transportation engineers across Canada are working together to design healthier cities and transportation systems. In particular, the City of Toronto has affirmed its commitment to the goal of healthy, sustainable land use and transportation planning in its Official Plan (City of Toronto 2010a). This report, *Active transportation and health in Toronto*, addresses the goal of creating a healthy city, focusing on active transportation as a means to improve health and quality of life in Toronto.

Active transportation refers to any form of human-powered transportation including walking, cycling, using a wheelchair, or skateboarding (Public Health Agency of Canada 2010). Toronto's Official Plan specifically highlights walking and cycling as a key element of the City's vision of creating an attractive and safe city that evokes pride, passion and a sense of belonging. The Official Plan also emphasizes the importance of walking and cycling in all parts of the city, including employment districts, avenues, centres, and regeneration areas. The objectives of this report, Active transportation and health in Toronto, include:

- To present research evidence on the health impacts of walking and cycling for transportation in urban areas.
- To investigate the health benefits and risks of walking and cycling in the City of Toronto.
- To identify strategies for improving public health by increasing the safety and use of active transportation across Toronto.

The report focuses on walking and cycling, including to and from transit. It also addresses the aspects of land use and transportation planning that enables or hinders safe active transportation. It draws on primary data analysis, the review of secondary sources, and interviews with municipal transportation and public health staff in other jurisdictions. The report is structured as follows:

- Chapter 1 reviews the literature on the benefits of walking and cycling for transportation, focusing on health benefits.
- Chapter 2 explores the current state of walking and cycling in Toronto and estimates the economic value of current and potential future levels of walking and cycling in Toronto.
- Chapter 3 investigates collisions, injuries and fatalities involving users of active transportation in Toronto, and estimates the costs of these collisions.
- Chapter 4 explores the distribution of health benefits and risks across the City of Toronto.
- Chapter 5 highlights actions to increase walking and cycling safety and mode shares, in order to improve public health in Toronto.
- Chapter 6 identifies strategies to facilitate implementation of these actions.

Chapter1. Active transportation: A review of the evidence

This chapter reviews the literature on the benefits of walking and cycling for transportation and their relevance to Toronto, with a focus on health benefits. It presents evidence on the significant health benefits experienced by individuals who increase their physical activity through the use of active transportation. It also discusses the population-level benefits of increases in physical activity and potential reductions in motor vehicle traffic. The chapter compares the health risks and benefits of active transportation and explores the costeffectiveness of investing in active transportation. Finally, it discusses the important social, economic, environmental and transportation system benefits of active transportation.

Active transportation as a source of physical activity

Physical activity is a critical part of staying healthy, and active transportation can help individuals meet the recommended levels of physical activity. The Canadian Physical Activity Guidelines recommend that all adults aged 18 and over obtain 150 minutes of moderate to vigorous physical activity each week, in bouts of at least 10 minutes. This corresponds to 30 minutes of physical activity (a 2 km walking trip or a 7.5 km biking trip), 5 days per week.

The Canadian Physical Activity Guidelines highlight that children aged 5-11 and youth aged 12-17 years need more physical activity than adults. The Guidelines recommend a minimum of 60 minutes of moderate- to vigorous-intensity activity per day. Walking and cycling to and from school can help children reach these levels, and may help to establish active lifestyle habits (Telama *et al.* 2005).

People who commute by active modes (e.g. walking or biking) get more physical activity than those who commute by inactive modes (e.g. driving) (Oja *et al.* 1998). Walking to and from public transit has also been identified as an important source of physical activity. Data from a study in the United States (Edwards 2008) indicates that public transit users walk 8.3 more minutes per day, on average, than non-users.

Research indicates that it is also easier to maintain physical activity levels through activities that are incorporated into daily life – such as walking, cycling or using stairs –than through activities that require a gym or recreation centre. "Lifestyle" physical activity interventions that are not "facility-dependent" are more likely to produce longer-term increases in activity levels (Hillsdon and Thorogood 1996, Dunn *et al.* 1998).

Active transportation may be a particularly important source of physical activity in the City of Toronto, where leisure-time physical activity levels are among the lowest in Ontario (Ontario Ministry of Health and Long-Term Care 2009). In the City of Toronto, only 43% of residents were moderately active or active in their leisure time in 2007¹ – below the Ontario average of 50%. A startling 93% of Toronto youth are not meeting the recommended daily requirements needed to derive health benefits from physical activity (Get Active Toronto 2011). Walking and cycling for transportation may enable Torontonians of all ages to get active and stay active without giving up other aspects of their busy lives.

¹ The Physical Activity Index estimates the age-standardized proportion of the population 12 years and older that is active or moderately active in their leisure time physical activity.

¹⁰ ROAD TO HEALTH: IMPROVING WALKING AND CYCLING IN TORONTO

Individual health benefits of physical activity from active transportation

By increasing their use of walking and cycling for transportation, individuals can significantly reduce their risk of a number of chronic diseases. These improvements in health translate to reductions in the risk of all-cause mortality, and to reduced societal costs of illness and disease. The health benefits of physical activity from active transportation are summarized in **Figure 2** at the end of the section.

Increased fitness and reduced obesity

Walking and cycling for transportation are as effective as traditional structured exercise programs in improving physical activity, cardio-respiratory fitness and blood pressure (Dunn *et al.* 1999). Individuals who begin to use active transportation experience improvements in heart rate, lung capacity and metabolic health (Oja *et al.* 1998, Gordon-Larsen *et al.* 2009).

People who use active transportation are also at a reduced risk of being obese, as walking and cycling can improve energy balance and body composition (Murphy *et al.* 2007, McAuley *et al.* 2000). One US study found that men who walk or cycle to work were half as likely to be obese (Gordon-Larsen *et al.* 2009). Another US study found that every additional kilometre walked per day is associated with a 4.8% reduction in obesity. In contrast, every additional hour spent in a car each day is associated with a 6% increase in the likelihood of obesity (Frank *et al.* 2004).

According to Basset *et al.* (2008), at the population level, countries with higher rates of active transportation and transit use have lower obesity rates, as illustrated in **Figure 1**.





Source: Transport Canada (2010), data from Basset et al. (2008)

In Toronto, over 40% of adults and 22% of adolescents are overweight or obese (Toronto Public Health 2010). Increasing the active transportation mode share could help to improve physical fitness and reduce the proportion of Torontonians who are overweight or obese.

Reduced risk of cardiovascular disease, diabetes, stroke and cancer

Active transportation reduces the risk of cardiovascular disease, including mortality, incident coronary heart disease, stroke, hypertension and diabetes. A meta-analysis concluded that active commuting is associated with an 11% reduction in cardiovascular risk (Hamer and Chida 2008a). High levels of walking for transportation have been associated with a 31% decrease in the risk of cardiovascular disease (Hamer and Chida 2008b).

There is strong evidence for the role of physical activity and active commuting in the prevention of Type II diabetes (Kesaniemi *et al.* 2001, Hu *et al.* 2003). Evidence from case-control and prospective studies also suggest that physical activity can reduce the incidence of stroke, though the evidence for this association is not as strong as for cardiovascular disease (Cavill *et al.* 2007).

Finally, physical activity has been found to reduce the overall risk of cancer, with particularly strong evidence related to colon cancer and breast cancer (Thune and Furberg 2001). Physical activity has been found to reduce the overall relative risk of colon cancer by 24% (Wolin *et al.* 2009). Active commuting in particular has also been associated with significant reductions in the risk of breast cancer (Hou *et al.* 2004).

Reductions in cardiovascular disease, diabetes, stroke and cancer are important for the City of Toronto, given the increasing rates of chronic disease. Rates of self-reported diabetes doubled from 4% to approximately 8% between 2001 and 2008 among Torontonians 12 and older, while high blood pressure increased from 12% to 15%. Circulatory diseases are responsible for more than 1 in 4 deaths in Toronto each year; breast and colorectal cancers alone are responsible for over 5% of deaths in Toronto (Statistics Canada 2011).

Improved mental health

Physical activity has a demonstrated positive effect on a range of mental illnesses. Physical activity has also been found to reduce the symptoms of depression, anxiety and panic disorders, with beneficial effect equal to meditation or relaxation (Paluska and Schwenk 2000). One study demonstrated that walking in particular can reduce anxiety and depressive symptoms in older women, and that walking was as effective as other forms of physical activity in reducing anxiety and depression (Heesch *et al.* 2010).

Research also suggests that physical activity can improve mental health in people without specific disorders. There is evidence that increasing physical activity can improve multidimensional self-esteem (McAuley *et al.* 2000), improve mood, reduce stress (Fox 1999 and Taylor 2000), and enhance perceptions of happiness and satisfaction (Taylor 2000).

In Toronto, 26.5% of the population aged 15 and over reported that most days in their life were quite stressful or extremely stressful (Statistics Canada 2011). Active transportation's contributions to stress reduction may thus prove valuable to many Torontonians.

Reduced risk of all-cause mortality

Physical activity from active transportation is associated with lower mortality. In a metaanalysis of 80 large cohort studies, people who received 150 minutes of moderate to vigorous activity per week have a 14% reduction in mortality, while people who received 300 minutes of physical activity experienced a 26% reduction in mortality. The highest levels of *active commuting* were associated with a 12% decrease in mortality as compared to the lowest levels (Samitz *et al.* 2011). The benefits of physical activity are stronger for people moving from no activity to low activity; the incremental benefit is less when an already active person becomes more active (Woodcock *et al.* 2011).

Population health benefits of physical activity from active transportation

Reduced medical costs

There is strong evidence that reducing physical inactivity yields substantial health care cost savings. A US study found that inactive individuals incur over \$600 in additional health care costs per year as compared to active individuals (Pratt *et al.* 2000). Another US study concluded that taking public transit is associated with walking 8.3 more minutes per day on average, which could save \$5,500 per person in net present value by reducing long-term obesity-related medical costs (Edwards 2008).

A Canadian study calculated that physical inactivity alone is directly associated with \$1.6 billion in annual health care costs in Canada, or 1.5% of all Canadian health care costs (Katzmarzyk and Janssen 2004). Each additional 10% increase in physical activity in Canada would translate to annual direct health care savings of up to \$150 million (Katzmarzyk *et al.* 2000).

Reduced work absenteeism

Research on the relationship between increased physical activity and reduced work absenteeism is limited but rapidly growing. A study of cycling and absenteeism among over one thousand Dutch employees found that, on average, people who cycled to work were absent 1 day less than non-cyclists over a one-year period (Hendriksen *et al.* 2010). A dose-response relationship was observed, with further reductions in absenteeism among people who cycled at least 5 km more than three times a week (Hendriksen *et al.* 2010). The study controlled for other differences between cyclists and non-cyclists. There is also evidence from workplace intervention programs that increased levels of physical activity will reduce absenteeism (Davis and Jones 2007).

Across Canada, physical inactivity is estimated to cost \$3.7 billion in economic productivity loss, due to its role in coronary artery disease, stroke, hypertension, colon cancer, breast cancer, type II diabetes and osteoporosis. Together, inactivity and obesity are estimated to account for \$6.4 billion in lost economic output due to short- and long-term disability and premature death (Katzmarzyk and Janssen 2004).

Increased physical activity from active transportation Reduced risk of cardiovascular disease, diabetes, stroke, cancer, obesity, etc...

Reduced mortality and morbidity

Reduced medical costs and economic output loss

Figure 2: Overview of health benefits of increased physical activity from active transportation

Population health benefits of reduced motor vehicle travel

Increasing the use of active transportation can indirectly improve population-level health by reducing motor vehicle travel and the associated health impacts. These health benefits are summarized in **Figure 3** at the end of the section.

Reduced air and noise pollution

Replacing car trips with walking and cycling trips can reduce air pollution and generate significant health benefits. One study modeled the air quality effects of eliminating automobile round trips of 8 km and shorter and replacing 50% of them with bicycle trips in 11 metropolitan areas in the Upper Midwestern US. They estimated net health benefits from reduced urban particulate matter (PM 2.5) and ozone to be \$3.5 billion per year (Grabow *et al.* 2011).

Exposure to air pollution from vehicles increases the risk of all-cause mortality at the population level (Hoek *et al.* 2002, Pope *et al.* 2002). Traffic-related air pollution is also associated with chronic respiratory conditions (e.g. chronic obstructive pulmonary disease) and acute respiratory effects. Finally, there is evidence that traffic-related pollutants are linked to cardiovascular disease, lung cancer, and increased risk of adverse pregnancy outcomes (Toronto Public Health 2007).

Noise pollution from traffic may also have negative health impacts that can be reduced through a shift to active transportation. These impacts include hearing impairment, sleep disturbance, impaired task performance, and impaired childhood development (Berglund *et al.* 1999, Stansfeld and Matheson 2003).

Air pollution from traffic is a major public health concern in Canada's urban centres, including Toronto. The Ontario Medical Association (2005) estimated that provincial costs associated with air pollution exposure were \$7.8 billion in 2005. In the City of Toronto, traffic-related pollution was estimated to cause 440 premature deaths, 200,000 restricted activity persondays and 1,700 hospitalizations per year in 2007. Mortality costs alone were valued at \$2.2 billion (Toronto Public Health 2007).

Increasing the mode share of active transportation may help Toronto to reduce these health costs of motor vehicle travel. It may also help the City to achieve its goal of reducing locally generated smog causing pollutants by 20% below 2004 levels (City of Toronto 2007).

Reduced traffic accident rates

Reducing the total automobile travel could also reduce motor-vehicle collisions, and the associated health costs. Automobile accidents take a heavy toll on human life and health in Canada. In 2009, they resulted in over 2,000 deaths and 172,000 injuries, 11,000 of them serious injuries (Transport Canada 2009). Motor vehicle collisions are a leading cause of death for young people, accounting for 70% of all accidental deaths in the 15 to 24 age group.

In Toronto, motor vehicle collisions resulted in over 18,000 injuries and over 40 fatalities in 2010. Toronto's injury rates as a result of motor vehicle collisions are significantly higher than in most other major Canadian cities. They are more than double those of Montreal, Calgary and Vancouver (City of Toronto Traffic Safety Unit 2012). Toronto has much to gain from reducing motor vehicle use and the associated burden of collisions, injuries and fatalities.



Figure 3: Overview of health benefits of reduced motor vehicle use from active transportation

Comparing health benefits, risks and costs

Research consistently concludes that the health benefits of active transportation dramatically outweigh the health risks, and that investments in walking and cycling infrastructure are extremely cost-effective.

Health risks of active transportation

In North America, individuals who walk and cycle face a higher risk of traffic injuries. Pedestrians and cyclists are more likely to be killed or injured than car and public transit users, by either per trip or per kilometre measures, in proportion to the number of pedestrians and cyclists on the road (Pucher and Dijikstra 2003; Elvik 2009).

However, these high injury and fatality rates are closely tied to walking and cycling conditions in North America. For example, cyclists in North America are twice as likely to be killed and eight times more likely to be seriously injured than cyclists in Germany, North American cyclists are three times as likely to be killed and 30 times as likely to suffer serious injuries than cyclists in the Netherlands (Pucher and Dijkstra 2003).

Air pollution exposure may also be higher for active commuters, though this remains a subject of debate (Thai *et al.* 2008, de Nazelle *et al.* 2009, Hertel *et al.* 2008). Several studies have found that air pollution exposure levels are lower for active transportation users as compared to inside motor vehicles or buses (e.g. Zuurbier *et al.*, 2010; Kaur *et al.* 2007). Other studies have found that active transportation users have higher air pollution exposure due to longer travel times and higher breathing rates (e.g. Briggs *et al.* 2008).

Health benefits versus risks

Studies conclude that the health benefits of shifting to active transportation outweigh the health risks – by at least a 15:1 ratio. For example, de Hartog (2010) estimated that individuals who shift from cars to bicycles gain 3-14 months of life thanks to the beneficial effects of increased physical activity. They may lose 0.8 to 40 days due to the mortality effect of increased inhaled air pollution doses, and 5 to 9 days due to the increased risk of traffic accidents. This corresponds to more than a 15:1 health benefit to risk ratio.

Rojas-Rueda *et al.* (2011) calculated the net reductions in all-cause mortality that results from participating in Bicing (a bicycle sharing initiative) in Barcelona, Spain. Overall, the 181,982 users resulted in 12.3 avoided deaths per year. Increased physical activity was associated with 12.5 deaths avoided, while increased traffic accidents and air pollution exposure were associated with 0.03 and 0.13 additional deaths, respectively. This leads to more than a 60:1 benefit-risk ratio.

Health benefits versus infrastructure costs

Walking and cycling infrastructure investments are extremely cost-effective – even when only considering the health benefits. In a review of 16 economic analyses of the health-related impact of interventions to increase walking and cycling, the median health benefit to cost ratio was 5:1 (Cavill *et al.* 2009). The five studies deemed highest quality by the authors yielded cost-benefit ratios from ~3:1 to ~14:1, as they included different benefits, costs, timeframes and methodologies to assess benefits and risks.

These ratios increase further when also considering benefits that are not health-related. An analysis by the Sustainable Development Commission (2011) highlights benefit-cost ratios of 18:1 to 38:1 for small-scale cycling schemes using the UK NATA framework for benefit-cost analysis. Major cycling infrastructure projects are pegged at 11:1, while local highway road schemes have benefit-cost ratios of 4:1 or 5:1. The NATA framework monetizes costs and benefits related to: changes in journey time, travel costs, accidents, noise and greenhouse gas emissions (UK Department for Transport 2011).

The high benefits to cost ratios enable the achievement of high mode shares without excessive costs. In Portland where the cycling mode share is between 3% and 7%, bike facilities comprised less than one percent of Portland's capital expenditures for transportation from 2001 to 2007 (Wallasper 2010).

Transportation system benefits

Many people walk and cycle because it makes practical sense from a traditional transportation point of view. For example, in Copenhagen where the mode share for walking, cycling and transit is 74%, 83% of cyclists say it is faster or more convenient than alternative modes (New York City Global Partners 2011). In Europe, walking is faster than driving for urban trips up to 0.5 km in length. **Figure 4** illustrates the time required for trips of different

lengths, using different modes of travel. This data is based on travel in urban areas in Europe. It demonstrates that walking and cycling are very fast modes of transportation for short trips in urban areas. This is in part due to the time required before beginning trips by car, bus or train. For example, **Figure 4** suggests that it takes about 4 minutes to unlock, turn on a car and leave a parking spot, even for a trip of only 0.1 km.

Figure 4 indicates that cycling is the fastest mode of transportation for urban trips of 5 km or less. Even for trips of 7 km in urban areas, cycling is almost as fast as driving, and is faster than other modes of transportation. Walking is generally as fast as driving for trips of about 500 m and less, and is faster than travel by bus for trips of up to 1.5 km.

Trip times using different modes undoubtedly vary by city and by time of day. Given Toronto's increasing levels of congestion, the general trends illustrated in**Figure 4** are highly relevant to Toronto – particularly during peak hours.





*The modes depicted in the figure do not all start at 0 minutes because of the time required at the "start-up" of some of the modes. For example, train trips require ~22 minutes to arrive at the station, purchase a ticket and board the train. Alternatively, there is no "start-up time" for walking, which starts at 0.

Investing in active transportation and public transport can also help to reduce congestion, reduce delays from collisions, reduce the unreliability of travel time, reduce fuel and transport costs and improve residents' ability to access facilities and services (Mohan 2010).

The benefits to individual travelers and to the transportation system as a whole are particularly relevant to Toronto. Seventy percent of Toronto commuters use cars, fuelling consistent traffic congestion in and around the city and metro area (Statistics Canada 2011). An Organization for Economic Co-operation and Development report estimated that traffic jams cost Toronto \$3.3 billion per year (OECD 2010). Recent reports by the Toronto Board of Trade also highlight transportation as a major factor limiting prosperity in Toronto. In the Board's *Scorecard on Prosperity,* Toronto Board of Trade 2011a, 2011b).

The strain on Toronto's transportation systems will only increase, as population and employment in the Greater Toronto and Hamilton Area are expected to grow by 50% between 2001 and 2031 (Ontario Ministry of Public Infrastructure Renewal 2006). As a growth centre, the City of Toronto will be home to more people, more jobs, and more trips of all kinds. Models from the University of Toronto Cities Centre show that travel times for all trips will nearly double by 2031, while speeds on highways and arterial roads will drop by about 20 per cent (Kalinowski 2011a). According to Metrolinx, the regional transportation agency for the Greater Toronto Area, active transportation has an important role to play in the transition to a more effective, sustainable transportation system (Metrolinx 2008).

Social, economic, and environmental benefits

Active transportation also yields benefits not directly related to health or transportation.

Increased neighbourhood safety, social benefits and real estate value

Active transportation as an alternative to car travel is associated with a number of social benefits, including increased social interaction, social networks and social capital (Leyden 2003). Residents of neighbourhoods that are pedestrian-oriented are more likely to feel a stronger sense of community than residents of automobile-oriented neighbourhoods (Lund 2002). Research shows a strong association between the sense of belonging to a community and physical and mental health (Statistics Canada 2011).

Increasing the mode share of walking and cycling may also contribute to reduced crime. Liggett *et al.*'s (2001) analysis of bus stop crime in Los Angeles found that bus stops with more pedestrians and higher visibility had reduced crime rates. Neighbourhoods with strong social networks have also been associated with lower crime rates (Bellair 1997).

Finally, improving the walking environment can increase economic value of local properties. Homes within walking distance of schools, parks and shopping sell for more in the United States (Cortright 2009). Improvements to the walking environment in the UK have been associated with increases in sale and rental prices of both residential and commercial properties (University of the West of England and Cavill Associates 2008).

In Toronto, only 66% of adults report a strong sense of belonging to their local community (Statistics Canada 2011). Developing walkable and bikeable neighbourhoods and communities may help to fulfill Toronto's vision of "creating an attractive and safe city that evokes pride, passion and a sense of belonging", stated in the Official Plan (City of Toronto 2010a).

Local economic activity and job creation

A number of reports demonstrate that cycling generates significant economic activity. In Portland, bicycling now accounts for \$100 million in local economic activity each year, and is directly responsible for almost 1000 jobs in the region (Cortright 2007). A similar study in Wisconsin found that cycling generates \$1.5 billion for the state economy (Wallasper 2010). In the UK, the "gross cycling product" is estimated at \$4.6 billion CAD, with 23,000 people directly employed in the cycling economy (Grous 2011).

There are also ways in which active transportation generates "net" economic benefits – economic activity that would not have otherwise occurred. For example, cycling expenditures are more likely to remain within the local economy rather than going almost directly to companies out of the city or province (as do expenditures on gasoline). Cortright (2007)

concludes that Portland keeps \$800 million that would drain out-of-town if local residents drove cars at the same rate as an average U.S. city. By spending less money on gasoline, a higher proportion of Portlanders' expenditures on goods and services are likely to remain in the local economy (Cortright 2007).

Walking and cycling investments that generate new tourism also generate a net economic benefit for the local economy. North Carolina's Outer Banks draw an estimated 680,000 visiting bicyclists each year and support 1,400 jobs in the area (Lawrie *et al.* 2004). Finally, research indicates that investments in cycling infrastructure generate more jobs per dollar spent than investments in road-only infrastructure. Cycling projects create a total of 11.4 local jobs for each \$1 million spent. Pedestrian-only projects create about 10 jobs per \$1 million and road-only projects create 9.6 jobs per \$1 million (Garrett-Peltier 2011).

Supporting active transportation is clearly well-aligned with Toronto's Official Plan vision of a city with a "a strong and competitive economy with a vital downtown that creates and sustains well-paid, stable, safe and fulfilling employment opportunities for all Torontonians" (City of Toronto 2010a). Promoting active transportation is also aligned with the 2007 "green" economic development strategy that encourages the retention, growth and attraction of companies or organizations whose products and/or services directly or indirectly reduce the impact on the environment (City of Toronto Economic Development, Culture and Tourism 2007).

Reduced greenhouse gas emissions

Investing in active transportation may be a particularly cost-effective way of reducing greenhouse gas emissions. The Intergovernmental Panel on Climate Change (IPCC) suggests that packages of walkways, bikeways and bus/rapid transit could reduce greenhouse gas emissions from light-duty vehicles by 25% at a cost of only US\$33 per tonne of CO_2 equivalent (WHO 2011a). By comparison, the IPCC estimates that reductions from high-efficiency vehicles would cost less than US\$110 per tonne of CO_2 equivalent (WHO 2011a).

One review of measures to promote active transportation found that between 17 and 57 kg of CO_2 equivalent could be saved per person walking to work and school; and 183 kg could be saved through personalized travel plans (Mohan 2010). The Bicing scheme in Barcelona is estimated to have reduced greenhouse gas emissions by 9 million kg of CO_2 equivalent, or just less than 1% of the annual emissions from motor vehicles (Rojas Rueda *et al.* 2011).

Toronto has an ambitious greenhouse gas reduction target within its 2007 *Climate Change, Clean Air and Sustainable Energy Action Plan.* The City is aiming to reduce greenhouse gas emissions to 80% below 1990 levels of 22 million tonnes per year, by 2050. The plan reported that approximately one-third of the locally generated greenhouse gases and a significant portion of smog causing pollutants come from the operation of motor vehicles and recommended creation of a long-term vision of a sustainable transportation system to decrease emissions.

Chapter 1 Findings

- Increasing the use of walking and cycling for transportation generates significant health benefits for individual users, and at the population level.
- The health benefits of walking and cycling significantly outweigh the risks.
- Better design for active modes can greatly increase safety; increasing the proportion of trips made
- by walking and cycling can also independently lower collision and injury rates.
- Increasing the use of active transportation can also generate important social, economic and environmental benefits.
- Investments in walking and cycling tend to be extremely cost-effective.

Chapter 2. Active transportation in Toronto

This chapter discusses recent increases in city-wide levels of walking and cycling in Toronto. It also quantifies the health benefits of current levels of walking and cycling. Despite the assumptions and estimations required, this economic assessment clearly demonstrates that active transportation generates important health benefits – benefits that are rarely considered in transportation planning and decision-making. Finally, it identifies the opportunity to significantly increase walking and cycling mode shares², and quantifies the potential benefits of these increases in active transportation.

Walking and cycling in the City of Toronto

Based on the 2006 census, 7.1% of Torontonians walk to work, and 1.7% cycle to work, leading to a combined active commuting mode share of 8.8%³. The 2006 Transportation Tomorrow Survey (TTS) indicates that 7.1% of trips for all purposes are made by walking, and 1.1% is made by cycling, leading to a combined mode share of 8.2%. The overall TTS mode shares for all trips made by City of Toronto residents are illustrated below (**Figure 5**).



Figure 5: Mode shares for all trips made by Toronto residents **Data**: TTS 2006 from University of Toronto Data Management Group (DMG 2008)

These census and TTS mode shares under-represent the importance of walking and cycling to Torontonians. First, walking and cycling trips for shopping and leisure activities are not included in the TTS, while trips for these purposes using other modes are counted. Furthermore, many Torontonians walk or cycle only in good weather, only to destinations other than to work or school, or only in combination with transit.

• **31% of Torontonians** report walking to work, school, shopping or leisure activities, as their main mode of choice (Harris-Decima 2008).

² Mode share describes the percentage of travelers using a particular type (mode) of transportation.

- **92% of Toronto public transit users** report walking to their transit stop, on average a 5 minute walk (Harris-Decima 2008).
- **29% of Torontonians** report cycling to work, school, shopping or other destinations in good weather an increase from 20% in 1999 (Ipsos-Reid 2009).
- The 2010 downtown bicycle screenline count suggests that approximately 48,000
 Torontonians cycle for transportation each weekday in the summer more than twice the number of cycle commuters in the 2006 census.⁴

City-wide statistics also disguise dramatic differences in levels of walking and cycling among residents of different neighbourhoods. As discussed in Chapter 4, among residents of some neighbourhoods in the Toronto core, over 25% of all trips are made using active transportation (DMG 2008).

Overall, increasing numbers of Torontonians of all ages are walking and cycling for transportation. Between 2001 and 2006, there was an 11% increase in the mode share for walking to work, and a 31% increase in the cycle commuting mode share based on census data (**Figure 6**). Between 1999 and 2009, there was a 20% increase in the number of Torontonians who reported cycling for transportation in the Toronto Bicycle Survey (Ipsos-Reid 2009).

Finally, increasing numbers of older adults are cycling for transportation, demonstrating the viability of this mode for Torontonians of all ages. Between 2001 and 2006, there was a 69% increase in cyclists aged 45-54, a 110% increase in cyclists aged 55-64, and a 55% increase in cyclists aged 65+ (City of Toronto Transportation Services 2011a).



Figure 6: Active transportation mode shares, 2001 & 2006 **Data**: City of Toronto Transportation Services 2011a

Despite these increases, Toronto's active commuting mode shares lag behind those of many other leading North American cities (**Figure 7**). Of these cities, Portland has the highest mode

⁴ In the downtown bicycle screenline count in September 2010, 19,162 cyclists entered the downtown area between 7am and 7pm. The TTS suggests that trips in the downtown area account for just fewer than 40% of all trips across the City of Toronto. One would therefore expect that in September 2010, 48,600 cyclists were cycling across the City. Note this estimation may underrepresent the numbers of trips as the count does not include trips started and ended within the downtown area.

share for cycling (5.8%) while Vancouver has the highest mode share for walking (12.3%) and for combined walking and cycling (16.0%).



Figure 7: Active commuting mode shares of North American cities **Data**: Pucher and Buehler 2011, TransportPolitic 2011

Toronto's trip distances suggest that there is significant potential to increase the active transportation mode shares to the levels observed in other North American cities. 21% of all trips made by City of Toronto residents are under 2 km and therefore very walkable. An additional 34% of all trips are 7 km or under, and therefore very bikeable (**Figure 8**). These data include one-way trips made by Toronto residents using all modes of travel, for all purposes. Therefore, at least 55% of all trips could be made using active transportation, recognizing that communities in Toronto with higher densities and infrastructure can achieve these higher modal shares for active transportation sooner than others.



Figure 8: Distances of all trips made by City of Toronto residents **Data**: 2006 TTS from DMG 2008

Quantifying the health benefits of active transportation in Toronto

In this report, we generate conservative estimates of the health benefits of walking and cycling in Toronto. We quantify only the benefits of reductions in mortality in the Toronto population due to increased physical activity from walking and cycling, using the World Health Organization's (WHO) Health Economic Assessment Tools (HEAT).

Estimating reduced mortality

The WHO tools (*HEAT for Walking* and *HEAT for Cycling*) evaluate the reduction in population-level mortality from a given level of walking and cycling. The HEAT uses data from published studies on the relative risk of death from all causes among regular pedestrians or cyclists, compared to people who do not cycle or walk regularly. Based on these "relative risk estimates", the tool calculates the reduction in expected deaths in the population that cycle or walk at this level. HEAT for cycling is designed for adult populations aged 20-64. HEAT for walking is designed for adult populations aged approximately 20-74 years.

The HEAT requires an estimate of how many people are walking or cycling and an estimate of the average annual duration or distance of walking or cycling. They also require the mortality rate for Canadians aged 20-74 to estimate the number of deaths that would have occurred in the absence of walking or cycling. We calculate this mortality rate at 383 deaths per 100,000, using Statistics Canada (2008) data.

Valuing prevented deaths

We use two different methods to estimate the economic value of the deaths prevented thanks to active transportation in Toronto: the more common "willingness to pay" method, and the more restrictive "discounted future earnings methods".

The value of a statistical life using "willingness to pay" (WTP) reflects how much a representative sample of the population (potential victims) would be willing to pay (in monetary terms) to avoid the risk of sudden death (WHO 2011b). We use the Value of a Statistical Life of \$4.05 million from a report for Transport Canada to monetize the benefits of these prevented deaths using willingness to pay (Sawyer *et al.* 2007).

The economic value of a prevented death can also be quantified based on the avoided loss of economic productivity from premature mortality. This economic productivity loss, called

"Discounted Future Earnings" (DFE), is calculated based on the average number of lost work days and daily earnings. We use the DFE value calculated by Vodden *et al.* (2007) in a report for Transport Canada. Vodden *et al.* (2007) calculate that the average DFE for Ontarians killed in motor vehicle collisions is \$1.1 million.

Estimating current levels of walking and cycling

No single data source provides a clear picture of the number of Torontonians who walk and cycle for transportation, the frequency with which they walk and cycle, or the distances walked and cycled. To estimate current levels of walking and cycling in Toronto, we combine data from multiple sources: the 2006 Transportation Tomorrow Survey (TTS), the 2006 Census, the 2010 Downtown Bicycle Screenline Count report (City of Toronto Transportation Services 2010), the 2008 Walking Survey report (Harris-Decima 2008), and the 2009 Bicycle Survey report (Ipsos-Reid 2009).

We also make conservative assumptions where none of these sources provide appropriate data. For example, none of the data sources provide information on the total number of Torontonians who regularly cycle for transportation in the summer months. They also do not provide adequate information on the use of active transportation for trips to shopping and to leisure destinations. Appendix A further explains the methods, the data sources and their limitations.

Methodologies and assumptions

The 2006 census provides information on the number of people walking and cycling to work, while the 2006 TTS provides data on the number of people walking and cycling to school. The number of people who walk and cycle to shopping and other destinations is estimated based on the TTS and walking survey ratios of people who walk and cycle to shopping versus to work.

Data on trip distances and the number of trips per week are based on the more conservative of two sources: the TTS and the walking and cycling surveys. We assume that walking is used year-round except for on vacations. We estimate the number of weeks of cycling to work per year based on the City of Calgary's survey of cycle commuters (City of Calgary 2007). We estimate seasonal cycling based on the Downtown Bicycle Screenline Count in combination with TTS data on relative cycling downtown and in other areas. We assume a walking speed of 4.8 km/h and a cycling speed of 15 km/h.

Appendix A describes the full methodology used to estimate current levels of walking and cycling. **Table 10** in Appendix A presents the resultant levels of walking and cycling for Torontonians who walk and cycle for different purposes, including: number of people, average trip distance, number of trips per week, and average annual distance. It also presents the associated reductions in mortality from the HEAT tools. The overall results of this analysis are presented below.

Deaths prevented through active transportation in Toronto

Current levels of active transportation are estimated to prevent about 120 deaths per year in the Toronto population. This mortality reduction alone can be associated with over \$475 million in economic benefits.

As indicated in Table 1, we estimate that about 180,000 Torontonians walk regularly for transportation, and about 80,000 cycle for transportation. Torontonians use active transportation for over 136 million kilometres of travel. Current levels of walking are estimated to prevent 69 deaths each year, while cycling prevents 49 deaths per year. Walking prevents more deaths than cycling as there are many more pedestrians than cyclists in Toronto. However, cycling leads to higher levels of physical activity and protection against mortality, resulting in higher benefits per cyclist.

Overall, these are very significant benefits, particularly considering the conservative estimate that under 15% of Torontonians walk or cycle for transportation.

		<u> </u>	
	All walking for transportation	All cycling for transportation	Total
Number of people	180,000	80,000	260,000
Total distance per year	68.3 million km	68.1 million km	136.4 million km
Number of deaths prevented each year	69	49	118
Willingness to pay value ¹	\$278 million	\$200 million	478 million
Discounted future earnings value ²	\$76 million	\$54 million	\$130 million

Table 1: Reductions in all-cause mortality from regular walking and cycling in Toronto

1. Based on Sawyer *et al.*'s (2007) Value of a Statistical Life (\$4.05 million CAD)

2. Based on Vodden *et al.*'s (2007) average Discounted Future Earnings of Ontarians who are killed in motor-vehicle collisions (\$1.1 million CAD)

The value of active transportation in Toronto

The reductions in mortality from current levels of walking and cycling in the Toronto population are worth between \$130 million and \$478 million each year. These benefits – which accrue from about 120 prevented deaths – are rarely considered in transportation planning and decision-making. How do these values compare to Toronto's expenditures on walking and cycling?

Approximately \$14 million of the City's annual capital transportation budget is dedicated to improvements in walking and cycling infrastructure – or, 5% of the total capital expenditures for Transportation Services. Over the next five years, \$9 million of the City's capital transportation budget will be allocated to cycling infrastructure and projects each year. At the same time, cycling is expected to prevent 49 deaths each year, for an annual value of \$54 million to \$200 million.

The City's Pedestrian Projects unit has an annual capital budget of approximately \$3 million for new sidewalks, pedestrian safety and other pedestrian projects. Other units of the City pay for signal improvements, signs and markings, multi-use trails, curb cuts and ramps at TTC stops, traffic calming/safety and new or widening of sidewalks as part of roadway reconstruction. Nonetheless, the total expenditures on pedestrian infrastructure are undoubtedly well below the \$76 to 278 million in benefits from prevented deaths each year.

These calculations consider only the economic benefits of prevented deaths through active transportation. Active transportation yields many other health benefits (**Figure 9**).



Figure 9: The \$130 to \$478 million in health benefits are only a fraction of total health benefits of walking and cycling in Toronto.

Many of the health benefits that are excluded in the above analysis have been quantified in other studies (**Table 2**). For example, increases in physical activity from active transportation improve health and reduce illness (or "morbidity") due to a number of diseases. The value of improved health and reduced morbidity is estimated at \$427 per newly active person using "willingness to pay" methods (Genter *et al.* 2008).

Increases in physical activity from active transportation also reduce health care costs. Pratt *et al.* (2000) estimate that active people in the United States incur \$622 less in annual medical costs than inactive people, on average. Increases in physical activity are associated with decreases in short- and long-term absences from work that have been valued at over \$400 per active person per year (Saelensminde 2004).

Using published estimates of the health benefits of physical activity from avoided chronic diseases, it is possible to estimate the direct savings in medical costs for Toronto residents who walk or cycle to work. From Table 1 it is seen that 260,000 residents use active transportation(based on 2006 data). Applying the per capita benefit estimates from New Zealand and the United States (see Table 2), it is calculated that the savings in direct medical costs range from \$110 to \$160 million.

Finally, active transportation has significant additional environmental, social, economic and transportation system benefits, as discussed in Chapter 1.

Annual value of benefits	Type of benefits	Source
\$427 per active person	Reductions in morbidity	Genter <i>et al.</i> (2008), New Zealand
\$622 per active person	Direct medical cost savings	Pratt et al. (2000), United States
\$449 per active person	Reductions in short-term absence from work	Saelensminde (2004), Norway

Table 2: Value of other types of health benefits associated with active transportation

The value of increasing active transportation in Toronto

As discussed earlier in the chapter, there is significant potential to increase the active transportation mode shares. At least 21% of all trips could be made using walking, and 55% of all trips in Toronto could be made by cycling. Toronto's median commute distance of 7.5 km further indicates that half of all trips to work could be made using active transportation. Increasing the active mode shares would undoubtedly yield important health benefits – benefits that may not be considered in decision-making.

We estimate the economic value of three scenarios of increased walking and cycling to work in Toronto (**Figure 10**):

- Scenario 1: a 25% increase in walking to work and a 50% increase in cycling. This would bring Toronto just above Montreal's mode shares of 8.6% and 2.4%.
- Scenario 2: a 50% increase in walking to work and a 100% increase in cycling. This would bring Toronto to just above New York City's walking mode share (10.3%) and just below Vancouver's cycling mode share (3.7%).
- Scenario 3: a 75% increase in walking to work and a 250% increase in cycling. This would bring Toronto to have the highest levels of active commuting in North America just above Vancouver's 12.3% walking mode share, and just above Portland's 5.8% cycling mode share.

All three scenarios are feasible for Toronto given current land use and trip patterns.



Figure 10: Increased active commuting mode share scenarios

We quantify the health benefits of these scenarios using the same trip distances and frequencies from Table 1. As in the analysis above, we assume that people walk and cycle to work 4-5 times per week, and walk and cycle to shopping or other destinations 1-2 times per week, leading to an average of 12 one-way trips per week. In order to present a conservative estimate of health benefits, we limit our analysis to people who walk and cycle to work. The levels of walking and cycling associated with Scenario 1,2 and 3 are presented in **Table 3**. The associated reductions in morality and values of prevented deaths are also presented in **Table 3**. As in the analysis of Toronto's current levels of walking and cycling, we use both "willingness to pay" (WTP) and "discounted future earnings" (DFE) values to estimate the economic values of the prevented deaths.

	Scenario 1		Scenario 2		Scenario 3	
	Montreal's walking levels	Montreal's cycling levels	New York's walking levels	Vancouver's cycling levels	Vancouver's walking levels	Portland's cycling levels
Number of active commuters	101,956	29,670	122,348	49,450	142,739	69,230
Annual deaths prevented	61	37	73	62	85	87
Economic value – WTP	\$246 million	\$151 million	\$295 million	\$252 million	\$344 million	\$353 million
Economic value –DFE	\$10 million	\$2.9 million	\$12 million	\$4.9 million	\$14 million	\$6.9 million

Table 3: Health benefits of scenarios for increased active transportation

As illustrated in **Table 4**, the three scenarios of increases in the number of active commuters in Toronto would prevent 25, 62, and 99 additional deaths each year, as compared to the 2006 baseline. These additional health benefits of scenarios 1, 2 and 3 can be valued at \$27 million, \$68 million, and \$109 million using Discounted Future Earnings valuations. Using willingness to pay values, these additional reductions in mortality can be valued at \$100 million, \$250 million, and \$400 million. As the City of Toronto prioritizes its investments in transportation infrastructure, the tremendous potential health benefits of walking and cycling should be adequately considered.

Results	Scenario 1	Scenario 2	Scenario 3
Total deaths prevented through active commuting	98	135	172
Additional deaths prevents (compared to 2006)	25	62	99
Economic value - WTP	\$100 million	\$250 million	\$400 million
Economic value - DFE	\$27 million	\$68 million	\$109 million

Table 4: Total and additional deaths prevented (compared to 2006)

Chapter 2 Findings

- Large and increasing numbers of Torontonians are walking and cycling for transportation.
- Toronto's active transportation mode shares lag behind those of other leading North American cities.
- Toronto's trip distances indicate that there is tremendous potential to increase the use of active transportation.
- Existing data sources do not provide a clear picture of the number of Torontonians who walk and cycle, their trip distances, and their trip frequencies.
- Using conservative methods, we estimate that 2006 levels of walking and cycling prevent about 120 deaths in the Toronto population. We estimate that 180,000 Torontonians walk for transportation and 80,000 cycle for transportation, traveling combined annual distances of over 136 million kilometres.
- The economic value of this reduction in mortality is estimated at \$130 million to \$478 million, depending on how deaths are valued.
- In comparison, annual capital expenditures on cycling and pedestrian projects are only about \$14 million, or 5% of total capital expenditures for City of Toronto Transportation Services.
- Increasing Toronto's active commuting mode shares to 12% for walking and 6% for cycling to become the "best in North America" would prevent about 100 additional deaths using conservative estimates, with an associated economic value of \$109 million to \$400 million.
Chapter 3. Health risks of walking and cycling in Toronto

This chapter discusses the health impacts of collisions, injuries and fatalities involving users of active transportation in Toronto⁵. It analyses the relative risk to pedestrians and cyclists, the changes in risk over time, and the types of collisions that are most common. It also estimates the costs of pedestrian and cyclist collisions with motor vehicles – costs that could be avoided through improvements to safety for pedestrians and cyclists.

Collisions, injuries and fatalities

In an average year in Toronto, over 1,000 cyclists and over 2,000 pedestrians are injured in collisions with motor vehicles. In 2009, collisions involving users of active transportation resulted in 31 pedestrian fatalities and one cyclist fatality (City of Toronto Traffic Safety Unit 2010a, 2010b). Due to under-reporting, the actual number of collisions is likely much higher than the number of reported collisions (2,189 for pedestrian-vehicle collisions, and 1,266 for cyclist-vehicle collisions).

Relative risks for pedestrians and cyclists in Toronto

Pedestrians account for 52% of all fatalities from collisions involving motor vehicles in Toronto, despite their mode share of only 7%. They also account for 11% of all injuries from collisions in Toronto. This reinforces that collisions with motor vehicles have very serious consequences for pedestrians-they very frequently lead to injury or death.

Similar trends are observed for collisions involving cyclists, though the discrepancy between mode share and fatalities is not quite as extreme. Cyclists account for 4% of all fatalities and 6% of all injuries from collisions, despite a cycling mode share of less than 2%.

Collisions involving pedestrians are particularly likely to lead to fatality or hospitalization. In 2009, 1.4% of collisions involving pedestrians led to pedestrian fatalities, 8.8% led to hospitalization, and 39% led to hospital trips for pedestrians. In 2010, 39% of collisions involving cyclists led to visits to hospital and 3.3% led to hospitalization for injuries.

⁵ We do not discuss exposure to air pollution since the relative risks or benefits for users of active transportation are highly uncertain (as discussed in Chapter 1).



Figure 11: Outcomes of collisions with vehicles for cyclists and pedestrians **Data**: City of Toronto Traffic Safety Unit 2010a, 2010b

Toronto has higher rates of walking and cycling collisions than other large Canadian cities, including Ottawa, Montreal and Vancouver when comparing pedestrian and cyclist collisions relative to the commuting mode share⁶. The City of Toronto (2011a, 2011b) compares Toronto's pedestrian and cyclist collision rates relative to the total population. In both analyses, Toronto's collision rates are higher than those of comparison cities including Montreal, Ottawa and Vancouver.



Figure 12: Active commuting collision rates in Canadian cities (per 1,000 commuters) **Data**: 2006 Statistics Canada census data, City of Toronto Traffic Safety Unit 2011a, 2011b

Signs of progress

Over the past ten years, collision rates have decreased substantially for both pedestrians and cyclists (**Table 5**). Collision rates have decreased relative to total population, and relative to the number of active commuters. The total number of pedestrian and cyclist collisions and fatalities also decreased between 2000 and 2009. This is particularly impressive given the significant increases in active commuting over this time period.

⁶ We use 2006 census data on population and active commuting mode shares to estimate

However, the relative fatality rates for pedestrians have not improved. In the five years leading up to 2000, pedestrians accounted for 49% of all fatalities from collisions in Toronto. In the five years leading up to 2009, pedestrians accounted for 52% of all fatalities form collisions.

	Pedestrians: % change 2000-2009	Cyclists: % change 2000-2009*
Number of collisions	-8%	-10%
Collisions / 100,000 population	-24%	-12%
Collisions / 1,000 commuters*	-18%	-33%
Number of fatalities (5-yr average)	-29%	-43%
% of fatalities (5-yr average)	+6%	-15%

Table 5: Changes in collisions, collision rates and fatalities for pedestrians and cyclists**Data**: City of Toronto Traffic Safety Unit 2011a, 2011b

*Number of commuters for 2000 and 2009 are based on the 2001 and 2006 census, respectively

Types of collisions

Collisions involving users of active transportation in Toronto most commonly occur on major arterial roads⁷ and at intersections. Pedestrians are often hit when crossing with the right of way, while cyclist-vehicle collisions are often due to inadequate separation of cyclists from vehicles. According to City of Toronto Traffic Safety Unit data (2010a, 2010b), in an average year:

- Over half (58%) of pedestrian-vehicle collisions and 69% of cyclist-vehicle collisions occur on major arterials (City of Toronto 2010a, 2010b).
- 47% of pedestrian-vehicle collisions occur at intersections, and in over 75% of these collisions pedestrians have the right-of-way.
- 37% of cyclist-vehicle collisions occur at intersections, where one party is turning.
- 675 collisions involved pedestrians crossing with the right of way who are hit by turning vehicles. These collisions accounted for 30% of total pedestrian-vehicle collisions, and led to 10 pedestrian fatalities in 2009.
- 30% of cyclist-vehicle collisions involve sideswiping collisions, collisions with car doors, and rear-ending collisions. Table 11 and 14 in Appendix B form the basis of this analysis, and provide a more complete list of the types of collisions and number of each type of collision in Toronto.

⁷ Arterial roads provide the major corridors for surface travel. Major arterial roads are defined by the City of Toronto Transportation Services (2000) as those with counts of greater than 20,000 vehicles per day; greater than 5,000 transit passengers per day, up to 4 lanes for vehicles, sidewalks on both sides, and high priority for winter maintenance. For a list of arterial roads in Toronto, visit: http://www.toronto.ca/transportation/road_class/pdf/city-wide_index.pdf

Costs of collisions, injuries, and fatalities

Pedestrian-vehicle and cyclist-vehicle collisions lead to injuries and fatalities with serious consequences for the people involved and for society as a whole. In this section, we quantify the costs of pedestrian-vehicle and cyclist-vehicle collisions in Toronto based on Vodden *et al.*'s (2007) report for Transport Canada, *The Analysis and Estimation of the Social Cost of Motor Vehicle Collisions in Ontario.* We use Vodden *et al.*'s (2007) analysis of the average cost per collision, distinguishing between those that lead to minimal injury, minor injury, major injury, and fatality. The analysis includes three types of costs:

- Human consequences: costs of fatalities and injuries to those affected
- Direct medical costs: costs of ambulances, emergency room visits, hospital stays, and medical professionals
- Other attributable costs: costs of police, fire services and time in traffic delays.

Average costs of collisions

The human consequences of injuries and fatalities are evaluated using two methods, as in Chapter 2: willingness to pay (WTP) and discounted future earnings (DFE). We use the same mortality values as in Chapter 2, from Sawyer *et al.* (2007) and Vodden *et al.* (2007). We use Vodden *et al.*'s (2007) WTP and DFE values for the human consequences of non-fatal collisions leading to major, minor and minimal injuries. The evaluation of human consequences accounts for the likelihood of a major or minor injury leading to permanent disability. **Table 6** presents Vodden *et al.*'s (2007) average costs for different types of collisions using both types of valuations. **Table 6** also presents the average direct medical costs of collisions, divided according to the severity of injury and fatality, as calculated by Vodden *et al.* (2007).

The "other indirect costs" in **Table 6** are specifically attributable to pedestrian-vehicle collisions, again calculated by Vodden *et al.* (2007). For the costs associated with police response, fire service response, and time costs of traffic delays, Vodden *et al.* (2007) present average costs for only two types of collisions: collisions leading to fatality, and collisions leading to injuries of all kinds. We assume that cyclist-vehicle collisions would have similar costs. To maintain a conservative estimate, we exclude the costs of court services, out of pocket expenses, and the costs of extra fuel and pollution associated with traffic delays that are estimated by Vodden *et al.* (2007).

Outcome of collision	Minimal injury	Minor injury	Major injury	Fatality
All human costs: willingness to pay	\$1,384	\$33,945	\$238,844	\$4,050,000
All human costs: future earnings	\$36	\$3,297	\$52,483	\$1,100,000
Ambulance trip	-	\$313	\$783	\$392
ER visit	-	\$95	\$95	\$48
Hospital stay	-	-	\$2,211	\$583
Health care professionals	\$405	\$920	\$4,600	\$1,887
All direct medical costs	\$405	\$1,328	\$7,689	\$2,909
Police	\$744	\$744	\$744	\$8,491
Fire	\$1,318	\$1,318	\$1,318	\$2,548
Traffic delays (time)	\$383	\$383	\$383	\$22,642
All other indirect costs	\$2,445	\$2,445	\$2,445	\$33,680

Table 6: Average human, direct and indirect costs per collision**Data**: Vodden et al. 2007

Costs of pedestrian-vehicle and cyclist-vehicle collisions in Toronto

To calculate the total costs of collisions involving users of active transportation in Toronto, we apply the average costs in **Table 6** to the annual number of collisions in Toronto. Even using the more restrictive "discounted future earnings" values to quantify the human costs of collisions and injuries, our analysis reveals that pedestrian-vehicle collisions yield costs of over \$53 million. (**Table 7**) Cyclist-vehicle collisions generate costs of over \$9 million (**Table 8**). When using "willingness to pay" values for human consequences, the total costs of pedestrian-vehicle and cyclist-vehicle collisions increase to over \$200 million and \$36 million, respectively.

All City of Toronto residents stand to gain from improvements in safety that reduces the economic burden of traffic collisions. Improving safety for pedestrians and cyclists in Toronto could avoid over \$60 million in direct and indirect economic costs.

Across Ontario, the 2% of collisions that involve pedestrians lead to 11% of the social costs of all collisions in Ontario (Vodden *et al.* 2007). Investing in safety for users of active transportation has the potential to yield disproportionate economic benefits.

	Number of collisions	Discounted future earnings	Direct medical costs	Other indirect costs	Total costs
Minimal injury	1014	\$36,504	\$410,163	\$2,478,954	\$2,925,621
Minor injury	878	\$2,895,089	\$1,166,160	\$2,146,471	\$6,207,719
Major injury	202	\$10,601,557	\$1,553,239	\$493,835	\$12,648,630
Fatality	28	\$30,800,000	\$81,452	\$943,042	\$31,824,494
Total	2,122	\$44,333,149	\$3,211,013	\$6,062,302	\$53,606,465

Table 7: Annual costs of pedestrian-vehicle collisions in Toronto based on 2005-2009 average collisions**Data**: City of Toronto Traffic Safety Unit 2011a, 2011b

Table 8: Annual costs of cyclist-vehicle collisions in Toronto based on 2005-2009 average collisionsData: City of Toronto Traffic Safety Unit 2011a, 2011b

	Number of collisions	Discounted future earnings	Direct medical costs	Other attributable costs	Total costs
Minimal injury	531	\$19,116	\$214,790	\$1,298,151	\$1,532,056
Minor injury	417	\$1,375,002	\$553,859	\$1,019,452	\$2,948,313
Major injury	40	\$2,099,318	\$307,572	\$97,789	\$2,504,679
Fatality	2	\$2,200,000	\$5,818	\$67,360	\$2,273,178
Total	990	\$5,693,436	\$1,082,039	\$2,482,751	\$9,258,227

Chapter 3 Findings

- Collision rates have decreased for both pedestrians and cyclists over the past ten years, even with increases in active commuting over this time period. Further improvements are needed as the relative fatality rates for pedestrians have not improved.
- Collisions involving pedestrians are particularly likely to lead to fatality or hospitalization.
- Toronto is also less safe for pedestrians and cyclists than other Canadian cities.
- Collisions involving users of active transportation in Toronto most commonly occur at intersections and on major arterial roads. Many also appear linked to inadequate separation of cyclists from motor vehicles.
- Collisions involving users of active transportation lead to direct and indirect economic costs of over \$60 million each year, using conservative estimates. Reducing these collisions and injuries would generate significant economic benefits.

Chapter 4. Distribution of health benefits and risks

This chapter discusses the distribution of the health benefits and risks of walking and cycling across the City of Toronto. It identifies dramatically higher levels of walking and cycling in the Toronto's core neighbourhoods (former City of Toronto and East York) – with a corresponding concentration of health benefits among residents of this area. It identifies lower levels of active transportation, walkability, and bikeability in Toronto's suburbs, and discusses how these patterns connect to income and health equity in Toronto. Finally, it discusses how active transportation collisions and injuries appear to be disproportionately borne by vulnerable users in Toronto: children, elderly adults, and residents of low-income neighbourhoods.

Walking and cycling in Toronto's core versus suburbs

The suburbs of the City of Toronto were primarily constructed after the Second World War. Unlike the narrow streets and density of Toronto's core, most of this post-war development is car-oriented, low density and uninviting to pedestrians (OPPI 2011). These differences in urban form are closely connected to current differences in transportation patterns – and the associated health benefits and risks.

The health benefits of active transportation in Toronto appear concentrated in the city's core, where levels of walking and cycling are dramatically higher, and where land use patterns enable safe walking and cycling. In the former City of Toronto, the combined active transportation mode share is three times higher than in North York, Scarborough and Etobicoke (**Figure 13**). In the wards closest to downtown (Wards 20 and 27), walking and cycling account for at least 26% of all trips over a 24-hour period (DMG 2006).



Figure 13: Walking and cycling trips as percentage of all trips over a 24-hour period **Data**: TTS 2006 from University of Toronto Data Management Group (DMG) 2008

In Toronto's core neighbourhoods with the highest levels of active transportation, residents make over five times more walking trips and over fifteen times more cycling trips, on average, than residents of many neighbourhoods in the suburbs (Figure 14 and Figure 15).



Figure 14: Daily cycling trips per person **Source**: Glazier et al. 2007



Figure 15: Daily walking trips per person **Source**: Glazier et al. 2007

Walking and cycling in Toronto's core is facilitated by the proximity to destinations. The Toronto –East York district has a high active transportation mode share of 15%, which is comparable to other leading North American cities (City of Toronto). In Toronto-East York, all types of destinations are significantly closer than in other parts of the city (**Figure 16**). For example, in the former City of Toronto-East York, the mean walking time to reach a convenience store is 3.2 minutes, and the mean walking time to a library is 10 minutes. In Scarborough, the mean walking times to reach a convenience store and library are 7.4 minutes and 18.2 minutes, respectively. As discussed in Chapter 5, the proximity to destinations in the former City of Toronto reflects land use patterns: land use mix, residential and commercial density, and route connectivity.



Figure 16: Mean walk time to destinations in minutes Data: Harris-Decima 2008

Cycling in Toronto's core is also facilitated by the presence of bicycle facilities. Bicycle lanes and paths are concentrated in south-central Toronto (Toronto –East York District), while bicycle facilities in the northern and eastern parts of the city are notably lacking (**Figure 17**).



Figure 17: Bikeway network – downtown versus suburbs Source: Created from City of Toronto Transportation Services (2010) cycling map

The health risks of active transportation may also be higher in Toronto's suburbs where levels of walking and cycling are lower. Intersections in Toronto that have the highest volumes of pedestrians also appear to have the lowest pedestrian collision rates, as illustrated in Reid's 2011 maps (**Figure 18**). These intersections are predominantly located in the downtown core and along subway lines. In contrast, intersections in Toronto's suburbs have lower pedestrian volumes and higher collision rates (red points, bottom right map). The finding that walking is less safe in Toronto's suburbs is aligned with the general "safety in numbers" trend, described in Chapter 1.



Figure 18: Left: volume of pedestrians (red = more pedestrians). Right: collision rates (red=higher rates)

Source: Reid 2011, adapted from Shaw Media / Global News 2011

Active transportation and equity in Toronto

Safe walking and cycling opportunities can reduce inequality by enabling individuals without motor vehicles to more easily access goods and services. In Canada, older adults, children, and low-income families are less likely to own cars, and are therefore most likely to benefit from improvements to alternative modes of transportation (Federation of Canadian Municipalities 2010). Active transportation can also reduce health inequality by improving access to health services and increasing physical activity.

Unfortunately, active transportation does not appear to effectively reduce inequalities in Toronto. This is in large part due to the dramatic differences between walking and cycling in the former City of Toronto and the suburbs, discussed above. In Toronto, low housing affordability in the core, land use patterns in high-rise neighbourhoods, and collision patterns may create health and transportation inequalities.

First, walkable and bikeable neighbourhoods in the former City of Toronto tend to be the least affordable for those with low incomes. Neighbourhoods in Toronto's core with the highest levels of walking and cycling have high land values, and are increasingly home to higher income households. As illustrated in **Figure 19**, average incomes in many parts of Toronto's core have increased over the past 35 years (blue, "City 1"), while incomes in Scarborough and in northwest Toronto have consistently decreased by 20% or more (dark red, "City 3").



Figure 19: Changes in income in Toronto, 1970-2005 Source: Hulchanski 2007

Second, low-income families often live in high-rise neighbourhoods in Toronto's suburbs. These neighbourhoods are characterized by wide roads, long city blocks, segregated land uses, and clusters of high rise apartment buildings (City of Toronto 2009a). They frequently feature limited pedestrian and bicycle infrastructure and few public gathering places (City of Toronto 2009b). Many residents feel uncomfortable walking in their neighbourhood at night (Harris-Decima 2008, Hess and Farrow 2010).

In a recent survey of residents of Toronto high-rise neighbourhoods, 28% of respondents disagreed or strongly disagreed that they feel safe from traffic when walking, and 29% disagreed that there are enough places to safely cross the large streets. Only 24% of parents reported feeling comfortable letting their children walk unaccompanied in their neighbourhood (Hess and Farrow 2010).

Additional physical barriers may also limit the use of active transportation in high-rise neighbourhoods. For example, two highways cross the Lawrence-Allen neighbourhood: the Allen Road Corridor and Provincial Highway 401. There are few connections across these highways, and those that exist are characterized by high traffic volumes and speeds, wide roadways, narrow sidewalks, nonexistent bicycle lanes, and highway on- and off-ramps. Only 38% of residents of the Lawrence-Allen neighbourhood described their neighbourhood as "very walkable", as compared to 76% for the city as a whole (City of Toronto, 2009a). In addition, other 400 series highways, rail corridors, and natural features such as ravines can similarly create barriers to persons wishing to walk or cycle.

Figure 20 identifies low-income areas in Toronto that also feature low or medium-low walkability, based on an index that considers land use mix, street connectivity, and density. These areas are primarily located in Scarborough and in the northwest corner of the City. This walkability index is further discussed in Chapter 5.



Figure 20: Low-income areas with low and medium-low walkability based on the UD4H walkability index

Source: Urban Design 4 Health (UD4H) 2012

Finally, children, older adults and residents of low-income neighbourhoods in Toronto may be most vulnerable to the risks of collision while walking and cycling.

As discussed above, pedestrian collision rates appear to be higher in the suburbs of Toronto, where pedestrian volumes are lower. Though studies have not quantified the risks to pedestrians in Toronto's low-income versus high-income neighbourhoods, the positive association between income and pedestrian safety has been recorded in many American cities (Active Living Research 2011). In Montreal, twice as many young cyclists and almost four times as many young pedestrians require medical interventions in the poorest areas than in the wealthiest areas (Agence de la Santé et des Services Sociaux de Montréal 2006).

Young and elderly people using active transportation also appear more vulnerable to collisions and fatalities in Toronto.

- Older people (over 65) account for 50% of pedestrian fatalities, though they account for only 13% of the population and 13% of pedestrians involved in collisions (City of Toronto Traffic Safety Unit 2007). The high fatality rates may be linked to the physical vulnerability of older adults, and to pedestrian infrastructure that ignores their needs (Ernst 2011).
- Boys aged 5-14 and Torontonians aged 15-24 are over-represented in pedestrianvehicle collisions (City of Toronto Traffic Safety Unit 2007). Cyclists aged 15-24 are also more likely to be involved in collisions than older cyclists; they represent only 12% of cyclists based on TTS data but are involved in 22% of collisions based on City of Toronto data (DMG 2008, City of Toronto Transportation Services 2011b).

However, there is also evidence that high levels of walking and cycling in low-income neighbourhoods may help to address health disparities. In Toronto, lower incomes are associated with higher premature mortality, higher cardiovascular disease, higher rates of diabetes, lower self-rated health and lower levels of physical activity (Toronto Public Health 2011, Creatore *et al.* 2007).

Figure 21 illustrates where low rates of diabetes coincide with low incomes in south-central Toronto (light blue), defying the dominant trend of high diabetes rates in low-income neighbourhoods (red). These neighbourhoods have higher levels of walking and cycling and are "activity-friendly" (Glazier and Booth 2007). Though dietary, genetic and other lifestyle factors also influence diabetes rates, these findings illustrate that increasing access to safe active transportation may help to improve health equity.



Figure 21: Significant relationships between diabetes rates and household income Source: Creatore et al. 2007

Overall, the links between health, equity and active transportation in Toronto have not been well studied. Future research should continue to explore how collision rates, access to safe walking and cycling, physical activity, and transportation choices vary for different groups across Toronto, with a particular focus on vulnerable users.

Chapter 4 Findings

- Levels of walking and cycling among residents of the Toronto's central areas are over three times higher than among residents of the suburbs. The city core (Toronto- East York District) is also more walkable and bikeable than the suburbs. The health benefits of walking and cycling are, therefore, concentrated in the central areas of Toronto.
- Pedestrian-vehicle collision rates appear higher in the suburbs. The health risks of walking are thus concentrated among residents of the suburbs.
- The suburbs of Toronto where walkability and bikeability are lower, and collision rates are higher
 – are predominantly low-income areas. Active transportation thus appears to further aggravate
 health and transportation inequality.
- However, there is evidence that high levels of walking and cycling in low-income and high-risk neighbourhoods may help to address health disparities.

Chapter 5. Strategies for improving active transportation in Toronto

This chapter highlights actions to increase walking and cycling safety and mode shares, in order to improve public health in Toronto. It presents infrastructure-based solutions that have been effectively implemented in other cities, and that appear likely to dramatically improve safety in Toronto. It also presents actions to improve the attractiveness of walking and cycling, and to ensure that active modes are safe and accessible for all Torontonians.

In this chapter we focus only on actions. However, we recognize that effective action is underpinned by enabling measures, goals and supporting capacity. In Chapter 6, we briefly review the enabling measures – plans, policies, and standards – that support these actions. We discuss the development of visions, goals and targets to provide the impetus for actions and measures. We also discuss the capacity that is needed to support the development and implementation of goals, policies and actions.



Toronto's actions on active transportation

The City of Toronto's efforts to promote walking and cycling demonstrate the City's commitment to active transportation. The Official Plan highlights the importance of walking and cycling as a core component of an effective transportation system. The 2001 Bike Plan's groundbreaking wheel and spoke model emphasized that hard infrastructure must be supplemented by education and softer measures; this model was used by many jurisdictions that developed bicycle plans subsequently. The City of Toronto's 2009 Walking Strategy received national recognition from the Federation of Canadian Municipalities in 2011 for "excellence, leadership and innovation" in transportation.

Many of Toronto's recent actions to promote active transportation have demonstrated leadership and innovation, and have been extremely successful. For example, the City's Green Standard is an exciting approach to sustainable new development that includes both mandatory requirements and incentives for going beyond the minimum requirements. The PATH underground walkway system in Toronto's downtown facilitates pedestrian linkages to public transit, and accommodates more than 100,000 daily commuters, providing shelter in the winter cold and in the summer heat. Other capital-related initiatives to improve walking in Toronto include pedestrian countdown signals, longer walk times for signalized crossings, zebra crossing pavement markings, and an Essential Links program to address missing sidewalks. Initiatives to improve cycling in Toronto include the' installation of over 17,000 post and ring bicycle parking stands, the introduction of BIXI bike sharing program, bike racks on all TTC buses, and the bike parking zoning bylaw The Active and Safe Routes to School program, the Bicycle-Friendly Business Awards and the guidelines for bicycle parking are examples of other exciting City of Toronto initiatives related to marketing and education, and policy tools.

The City is also experimenting with bicycle boxes and pedestrian scramble intersections. The Metrolinx regional transportation plan that emphasizes walking, cycling and integration will complement municipal efforts with transit at mobility hubs. Finally, Toronto features an active network of profit and non-profit organizations that work to improve walking and cycling for city residents. There are over 100 bicycle specialty shops, clubs and teams in Toronto. These efforts are further described in Appendix C, and in the relevant sections of this chapter.

Nonetheless, Toronto lags behind other leading North American jurisdictions. Toronto's walking and cycling mode shares of 7.1% and 1.7% respectively, are well behind those of other North American jurisdictions including Vancouver and Montreal (Chapter 2). Toronto's pedestrian and cycling collision and injury rates are above those of Vancouver and Montreal (Chapter 3).

Improving active transportation in Toronto

Improving active transportation in Toronto would bring significant public health benefits. The City of Toronto can reduce the health burden of transportation by reducing the number of pedestrian-vehicle and cyclist-vehicle collisions and fatalities. The City of Toronto can also generate millions of dollars in health benefits by increasing the number of pedestrians and cyclists – both due to reductions in mortality, reduced health care costs and reduced costs of lost economic output from physical inactivity (Chapter 2).

There are a number of dimensions to improving active transportation in Toronto. First and foremost, Toronto should continue to target safety improvements for pedestrians and cyclists. Walking and cycling is currently less safe than other modes, as pedestrians and cyclists are more likely to be injured in collisions. Pedestrians are particularly at risk, accounting for 50% of all fatalities from collisions in Toronto (Chapter 3).

Addressing concerns about safety is also essential to increasing active transportation mode



shares – particularly cycling mode shares. According to the Ontario Bike Plan, the number one reason that people do not cycle is because they are afraid of cycling on the road amidst traffic (Stantec and Vandermark Consulting 2008). Surveys in North American cities suggest that up to 70% of residents would cycle if they felt safe doing so. In Calgary, only 2% are "fearless" riders who will cycle in all conditions, and only 28% are not interested in cycling (City of Calgary 2011). This is supported by Toronto survey data, where general safety is the number one concern about cycling in Toronto (Ipsos-Reid 2009). According to the "safety in numbers" theory, increasing the active transportation mode shares would also help to increase safety. The incidence of collisions and injuries decreases as the mode share of cycling increases (Jacobsen 2003, Robinson 2005, Elvik 2009, Aertsens *et al.* 2010). This is likely due to drivers' increased awareness of cyclists and ability to anticipate their actions. There is a positive feedback between cycling safety and mode share. As cities invest in safe cycling infrastructure, cycling safety and mode shares increase, leading to further increases in safety for all modes.

This chapter presents actions that have been demonstrated to increase active transportation safety and mode shares in other cities, and discusses their relevance to Toronto. The following types of actions are presented:

- Making active transportation safer
 - o Reducing vehicle speed limits
 - o Traffic calming
 - o Separating pedestrians and cyclists from traffic
 - o Increasing safety at intersections
- Making active transportation more attractive
 - o Route quality and speed
 - Transit integration, especially in suburbs where active transportation distances are longer
 - o End-of-trip facilities
 - o Marketing and education programs
- Enabling active transportation through land use planning
 - o Destination accessibility, street connectivity and density
- Making active transportation more accessible
 - Walking and cycling facilities for vulnerable users
 - o Affordable housing in vibrant communities
 - o Culturally competent community-based programs

The chapter also discusses how cities can maximize results by implementing coordinated packages of actions.

Making active transportation safer

The analysis of Toronto collision data, Toronto survey data, and research from other cities specifically suggests the importance of traffic calming, separating pedestrians and cyclists from traffic, and increasing safety at intersections. These efforts are particularly important on major arterials where 69% of cyclist-vehicle collisions and 58% of pedestrian-vehicle collisions occur. Major arterials provide direct, connected walking and cycling routes, feature a wide variety of destinations, and are designated as priority areas for re-urbanization in the Official Plan (City of Toronto 2010).

Reducing Vehicle Speed Limits

The speed of a vehicle is an important factor in both the occurrence of a collision and the severity of injuries that result (WHO 2004). A number of studies have looked at the relationship between vehicle speed and pedestrian or cyclist injuries when a collision occurs. A pedestrian is 8 times more likely to die as a result of a collision when a vehicle is going 50 km/hr than when it is going 30 km/hr (WHO 2004). Based on work by Anderson et al, 1997, Figure 22 shows that less than 5% of pedestrians are likely to die as a result of a collision when it occurs at speeds below 30 km/hr; this rate rises to about 25 percent at 40 km/hr and about 85 percent at speeds of 50 km/hr. The greatly increased probability of death or serious injury when hit by a vehicle travelling 50 km/hr compared with 40 km/hr provides compelling evidence to reduce the typical city-wide speed limit in Toronto from 50 to 40. A pedestrian hit while a vehicle is going at 60 km/hr or more is unlikely to survive.

In another study, the UK Department of Transport found that the risk of pedestrian death in crashes rises from 5% at about 30 km/h, to 45% at about 50 km/h, and 85% at about 65 km/h (Pucher and Dijkstra 2003), findings that again demonstrate the greatly increased risk with traffic speeds of 50 km/hr or higher.



Figure 22: Probability of Pedestrian Fatality by Impact Speed **Source**: Derived from Anderson et al. 1997

Some pedestrians are at greater risks than others. For example, pedestrians over 65 are about five times more likely to die as a result of a collision than children under 14 (WHO 2004).

A review of 19 traffic-calming initiatives in four European countries found that injuries caused by collisions for all road users fell by 41-83%, while fatalities dropped by 14-85% (Preston 1995). After 30 km/h zones were introduced in London, these zones experienced a 42% reduction in fatalities (Grundy et al. 2009). In 1988 the Town of Baden, Austria restricted speeds to 30 km/hr to about 75 percent of its road network. This and other measures reduced the rate of casualties by 60 percent (WHO 2004). New York City is now piloting reduced neighbourhood speed zones, with speed limits of about 30 km/hr.

Traffic calming

Traffic calming strategies generally feature physical treatments that reduce vehicle speeds, as noted in the City of Toronto's 2010 Traffic Calming policy. Physical treatments such as road narrowing, curb bulges, traffic islands and medians have been demonstrated to effectively reduce vehicle speeds (Transport Association of Canada and Canadian Institute of Transportation Engineers 1998). Innovative uses of landscaping, signage, and pavement markings can also narrow roads and reduce vehicle speeds at low cost. Traffic calming strategies may also include reduction of speed limits and increased enforcement of existing speed limits.



Figure 22: Traffic calming using curb extensions, pavement markings, speed bumps and signage **Source**: Wikipedia 2012 (left), stock.xchng 2007 (right)

Traffic calming can significantly reduce the total number of collisions, injuries, and fatalities. Traffic calming can also help to promote increased levels of walking and cycling. Traffic calming increases the speed of walking and bicycling relative to the speed of driving, and also increases the safety of walking and bicycling (Pucher *et al.* 2010). In Hilden, Germany, widespread automobile speed restrictions led to a significant increase in bicycling (Pucher *et al.* 2010). Area-wide traffic calming has been extensively used in France, the United Kingdom, and Switzerland as part of overall strategies to increase the use of walking and cycling for transportation (Gagnon and Bellefleur 2011).

In Toronto, speed humps are frequently used to reduce vehicle speeds on residential streets. However, there are a number of ways in which traffic calming can be further used to improve safety for cyclists and pedestrians in Toronto. Options can be evaluated and implemented based on the specific needs of roadway users. For example:

- Toronto can reduce speed limits on streets with high pedestrian collision and injury rates (e.g. major arterial roads), and increase enforcement of existing speed limits.
- Toronto can reduce speed limits on streets that form part of the bicycle network (either on major arterials with bike lanes, or on local streets without bike lanes intended for bicycle traffic).
- Toronto can develop additional bicycle routes on local streets where traffic calming is already implemented (see *Bicycle boulevards* section below).

• Toronto can implement physical measures such as curb extensions, traffic islands and landscaping at intersections (see *Safer intersections* section below).

Separation from traffic

Cycling and walking are safer and more enjoyable if cyclists and pedestrians are well separated from traffic for the entire length of their journey. The extent and continuity of sidewalks are associated with increased walking and reduced pedestrian-vehicle collisions (Hess *et al.* 1999). In residential and mixed residential neighbourhoods in the US, pedestrian crashes were more than twice as likely to occur at locations without sidewalks as would be expected on the basis of exposure (Knoblauch *et al.* 1988).

Well-connected and high quality bicycle facilities are also essential to improving safety and attracting new people to bicycling (Dill 2009, Titze *et al.* 2008). Bicycle facilities that separate cyclists from motor vehicle traffic are strongly associated with increased levels of cycling (Pucher *et al.* 2010), and reduced crashes and injuries (Reynolds *et al.* 2010). In the City of Portland, the 210% increase in cycling from 1991 to 2004 is linked to two key factors: the quality of cycling facilities; and the completeness of the bikeway network, including clear connections and near-continuous service. Where the bikeway network was not well connected and not of the highest quality, bicycle use essentially remained at the levels that existed prior to network expansion (Birk and Geller 2005).

For pedestrians, Toronto has a fairly complete sidewalk network and a large number of offroad paths. Toronto has also developed the PATH network, the largest underground shopping complex in the world. The PATH network provides a climate-controlled environment for yearround use by 100,000 daily commuters, and thousands of additional tourists and residents (City of Toronto 2012).

While the City maintains an extensive network of approximately 7,900 kilometres of sidewalks, there is still a lack of sidewalk continuity in many areas. Over 1,600 km of sidewalks are missing from Toronto's local roads, and over 450 km of local roads are missing sidewalks on both sides of the road. These roads are almost exclusively located in North York, Scarborough and Etobicoke-York (City of Toronto 2011). In locations where sidewalks are missing, pedestrians often have no alternative but to walk on unimproved shoulders that may be inaccessible in the winter, or on the roadway (City of Toronto 2011). The City's Missing Sidewalk Program is working to gradually fill in these gaps.

Toronto is also making progress towards a full, connected set of bike paths and lanes. However, Toronto lags behind other cities in providing cyclists with connected facilities that separate them from traffic. Toronto has fewer kilometres of bike lanes per resident than other Canadian cities. As of 2010, Toronto had 17 km of bicycle lanes per 100,000 residents. Montreal and Vancouver had 28 and 38, respectively.

Toronto's bike lanes were described as "poor" by 54% of Torontonians and 64% of Toronto cyclists. Over 70% of Torontonians – those who already cycle for transportation and those who do not – reported that separating bike lanes from traffic would most improve cycling in Toronto (Ipsos-Reid 2009). Toronto's on-street bicycle lanes and routes are also commonly critiqued for being neither continuous nor connected (City of Toronto 2011). Improvements to cycling facilities are also needed to reduce the 30% of Toronto collisions that involve sideswiping, car doors, and rear-ending.

As demonstrated in **Table 9**, Montreal and Vancouver have used different strategies to create connected bicycle facilities that separate cyclists from traffic. Montreal has invested heavily

in separated bike lanes on major arterials, while Vancouver has developed a connected network of traffic-calmed, bicycle-oriented local street bikeways (bicycle boulevards). Toronto has 14 km of separated facilities in development but does not yet have any separated bike lanes built at this time (City of Toronto 2012).

Table 9: Kilometers of roadway/pathways with cycling infrastructure in Toronto, Montreal and Vancouver as of 2010*

Bicycle Pro	ogram Coordi	nator (2012))				
City	Separated bike lanes	On-street bike lanes	Off-road paths	Marked shared lanes (sharrows)	Local street bikeways	Total	Kilometers per 100,000
Montreal	249 km	94 km	84 km	0 ¹ km	0 km	528 km	28

9 km

10 km

147 km

134 km

460km

221 km

17

38

Data: City of Toronto Transportation staff (2012), Ville de Montreal (2010), City of Vancouver Bicycle Program Coordinator (2012)

191 km

41 km

*The values in Table 9 represent approximate kilometers of roadways and pathways with bicycle lanes. A one kilometer path or road with two-way cycling infra-structure is counted as 1 km.

1. Montreal primarily uses sharrow markings in intersections

113 km

30 km

2. Estimates based on data on one-way lane-kilometers

0

6 km

Toronto

Vancouver²

Toronto's 2011 bikeway strategy includes the construction of an additional 100 km of offstreet trails along rail, hydro corridors, ravines and valleys to serve as the backbone for bicycle transportation across Toronto, particularly in the suburbs. It also includes the completion of "critical on-street bike lane connections where the community supports them and where they do not impede traffic flow, including a separated bike lane network downtown" (City of Toronto Transportation Services 2011b).

A complete bikeway network that combines off-street trails, separated bike lanes and onstreet connections would have the potential to dramatically improve cycling in Toronto. However, this strategy disregards the importance of major arterials to cyclists, and the large number of collisions on these streets. In addition, the "critical on-street bike lane connections" that link trip origins, off-street trails, and destinations are essential to the network's connectivity – and to the viability of cycling in Toronto (Birk and Geller 2005). This is particularly important in the suburbs, where cycling is rare and high-speed vehicle traffic dominates.

To develop these links, Toronto can use a variety of innovative options that consider cyclists' needs and local roadway conditions. Cyclists' needs are primarily affected by vehicle speeds and volumes; they may also be affected by roadway and lane widths, pavement quality, onstreet parking, construction, and other roadway obstructions. The different roadway conditions in Toronto's core versus in the suburbs will create different opportunities and constraints.

Several on-road options to link trip origins, trails and destinations are presented below.

Separated bike lanes (cycle tracks). "Separated bike lanes" are lanes on major roads that are exclusively for bicycle use and are separated from motor vehicles by a curb or other physical barrier. They provide riders with greater confidence and significantly reduce collisions. Based on surveys, separated bike lanes are viewed as being equally desirable as off-street paths, a testament to their perceived safety (Winters *et al.* 2011).

In New York City, the installation of cycle tracks decreased injuries to cyclists and pedestrians by 57% and 29%, respectively (District Department of Transportation 2011). Studies in London found that new cycle tracks decreased the rate of bicycling crashes and increased the number of cyclists on the roadway by 58% over 3.5 years (Pucher *et al.* 2010).

Bicycle lanes that are separated from cars by concrete curbs are estimated to cost \$470,000 per kilometre on existing roads, or \$320,000 when as part of a road reconstruction project (Toronto Coalition for Active Transportation 2011). However, less expensive options for separating cars from bicycles have been used in other cities. In Washington D.C., bike lanes are separated from parked cars and moving traffic by yellow posts. **Figure 23** (right) illustrates the use of posts to separate bike lanes from traffic.

Buffered bicycle lanes. Buffered bike lanes can also improve on-street separation of cyclists from traffic at low cost. With buffered bike lanes, conventional on-street bike lanes are paired with painted, designated buffer spaces separating the bike lane from the adjacent motor vehicle travel lane and/or parking lane (National Association of City Transportation Officials - NACTO 2011) (**Figure 23**, left).

Coloured on-street bike lanes. In cities including New York and Vancouver, on-street bike lanes are brightly painted to visually reinforce the separation of areas for motorists and cyclists. Though few studies have evaluated the impact of coloured lanes, it is consistent with the research by Retting et al. (2003) that increasing the visibility of bike lanes would also increase safety for cyclists. Anecdotally, most cyclists believe4that the green paint is more effective at keeping cars from parking in bike lanes than regular striping (NYC Department of Transportation 2011). Coloured bike lanes can also be used in combination with buffered or separated bike lanes (**Figure 23**).



Figure 23: Buffered bike lane (left) and separated, coloured bike lane (right) **Source**: NACTO 2011

Bicycle boulevards. Bicycle boulevards (also known as "greenways" or local street bikeways) are low-volume and low-speed streets that have been optimized for bicycle travel through treatments such as traffic calming, signage and pavement markings, and intersection crossing treatments (IBPI 2009). Bicycle boulevards may also include contra-flow bicycle lanes that allow cyclists to ride in the opposite direction of motor vehicle traffic. They convert a one-way traffic street into a two-way street: one direction for motor vehicles and bikes, and the other for bikes only (NACTO 2011). Studies have found that some cyclists go out of their way to use bicycle boulevards, particularly women and less-experienced cyclists. In a survey, 42%

of residents reported that living on a bicycle boulevard makes them more likely to bicycle (VanZerr 2010).

Re-allocating space from motor vehicles to active transportation

Separating pedestrians and cyclists from traffic often requires reallocating space from motorized vehicles to pedestrians and cyclists. This is particularly relevant to the City of Toronto. According to City of Toronto Transportation Services (2011b), there are few remaining opportunities to accommodate bicycle lanes on Toronto roadways without reducing traffic capacity or on-street parking. Consequently, in both the downtown and the suburbs, safe cycling facilities will likely require the reallocation of roadway space from on-street parking or motor-vehicle traffic to cyclists.

Re-allocating space from motor vehicles to pedestrians and cyclists need not mean increased traffic (or congestion) for motorists. In a study of 62 roadway reallocation projects in eleven countries, 51 of the 62 cases showed a traffic decrease as the overall result, with 11% reductions on average (Cairns *et al.* 2002). There may be reports of short-term problems and problems on particular streets. However, most transportation staff reported, "a lot of the traffic seems to have disappeared, and we don't know where it has gone" (Cairns *et al.* 2002, p.14). These results indicate that people have a wide range of behavioural responses: some may simply change their route or their time of travel, while others may change their means of travel, the frequency of travel, or their destination.

In Vancouver, the installation of separated bicycle lanes on Hornby Street had little effect on vehicle travel times along the street. Only the northernmost block of Hornby Street had increased delays, averaging 30 seconds in peak periods (City of Vancouver 2011).

In New York City, a bold program to reallocate space has transformed Times Square and Herald Square into pedestrian plazas. The change has increased the number of pedestrians by 11% in Times Square and has decreased injuries to motorists and passengers by 63% in the project area. It has also reduced traffic congestion in the area and increased travel speed along most routes. Finally, the scheme has broad public support; 74% of New York residents agreed that Times Square has improved dramatically as a result of the change (New York City DOT 2010).

The Toronto Centre for Active Transportation reports that many Toronto residents are in favour of roadway re-allocations that favour pedestrians and cyclists. In Bloor West Village, nearly half (48%) of those who usually drive to the neighbourhood reported that they would prefer a more pedestrian- or cyclist-oriented street, even if it meant removing on-street parking (TCAT, Forkes and Smith Lea 2010). In a survey of visitors to Toronto's Annex neighbourhood, 62% of people were most in favour replacing on-street parking with bike lanes on Bloor Street compared to other alternatives including the status quo, and an additional 16% were most in favour of widening sidewalks at the expense of on-street parking. Almost 75% of business owners and managers thought that their business would improve or stay the same with these changes (Sztabinski 2009).

Safer intersections

Improvements to intersections can reduce collisions and injuries for pedestrians and cyclists alike (Retting *et al.* 2003). Interventions to improve safety at intersections generally fall into three categories:

- Traffic signal phases
- Physical interventions
- Roadway and intersection markings.

As presented in Appendix B, 47% of collisions between pedestrians and motor-vehicles could be prevented through improved safety at intersections in Toronto. 37% of collisions between cyclists and motor vehicles could be prevented through improved safety at intersections. The City of Toronto is already making efforts to improve intersection safety:

- Since 2006, the City has installed pedestrian countdown signals at over 2,100 intersections (TCAT 2010).
- Beginning in 2007, signalized intersections were being converted to permit longer pedestrian crossing time based on a reduction in presumed pedestrian walking speed from 1.2 m/second to 1.0m/second.
- Red light cameras have proven to reduce injury collisions attributed to red light running by more than 60 percent. There are currently approximately 87 red light cameras operating at Toronto signalized intersections.
- Three busy downtown intersections in use at Yonge/Dundas, Bloor/Yonge and Bloor/Bay intersections currently have a Pedestrian Priority Phase, or "pedestrian scramble".
- Pedestrians also now receive a head start at the intersection of University Avenue and Adelaide Street West (thanks to a "leading pedestrian interval" signal phase).
- "Bicycle boxes" were implemented at five intersections in 2010 and 2011 in Toronto's downtown-west end area.

To improve safety at intersections, Toronto can continue to experiment with different treatments of intersections for pedestrians and cyclists, monitor outcomes, and roll-out effective options in a systematic way. The City could consider a wide range of promising options that have been demonstrated to improve safety. These options are discussed below. Some of these options require only surface coatings, and are therefore available at relatively little cost.

Traffic signal phases

Pedestrian scrambles. Pedestrian scrambles–already in use at the Yonge-Dundas, Bloor-Yonge, and Bloor-Bay intersections-provide an all way stop for motor vehicle traffic to allow pedestrians to cross in all directions. Pedestrian scrambles were found to cut the risk of pedestrian-vehicle collisions in half (Zegeer *et al.* 1982).

Leading pedestrian signal intervals. With a leading pedestrian signal interval (LPI), the WALK sign is turned on 3-5 seconds before the green light, so that the first vehicle intending to turn right or left has improved visibility and yielding response time for pedestrians that have already started to cross. In a US study, LPIs reduced the odds of conflict with turning vehicles by 95% at the beginning of the walk phase (Van Houten *et al.* 2000).

In Cambridge, Massachusetts, pedestrians receive a head start at almost all intersections where vehicles and pedestrians move at the same time. The WALK signal comes on a few seconds in advance of the green light.

Bicycle phases. Separate traffic signal phases can provide time for cyclists to cross an intersection without motor vehicle traffic. A study in Davis, California, found that bicycle phases dramatically reduced automobile bicycle collisions. There were 10 collisions near a focus intersection in the 35 months before the installations of a bicycle phase; in the 35 months following installation there were no collisions at this same intersection (Korve and Niemeier 2002).

Leading bicycle signal intervals can also provide a "head start" strategy for cyclists and can be expected to yield a similar reduction in collisions at signalled intersections.

Physical interventions

Curb extensions. Curb extensions (curb bulbs) reduce crossing distances, provide more pedestrian storage space, slow vehicles and allow pedestrians and approaching vehicles to see each other (Retting *et al.* 2003).

Medians and refuge islands. Refuge islands and raised medians can lower pedestrian-car collision rates on multilane roads (e.g. Zegeer *et al.* 2001).

Crosswalk lighting and signage. Clear lighting of crosswalks can improve safety for pedestrians. In two studies, in-pavement flashing lights that were automatically activated by the presence of pedestrians reduced vehicle speeds and conflicts at crosswalks (Retting *et al.* 2003). Simple advance warning signs for drivers to "Yield to Pedestrians" have been shown to reduce pedestrian crashes by 10% (Lynott *et al.* 2009).

Roadway and intersection markings

Crosswalk markings. Modifications to pavement markings around crosswalks may improve safety for pedestrians. For example, simply repositioning stop lines further from crosswalks has also been show to increase the distance between pedestrians and halted cars in the US (Retting and Van Houten 2000). Textured and coloured markings can also improve safety at crosswalks, as recognized by Toronto's 2006 zebra crossing policy. As of 2010, 300 intersections had zebra crosswalk markings installed in the City of Toronto.

Intersection markings for cyclists. Pavement markings through intersections indicate the intended path of bicyclists through an intersection and raise awareness of potential conflict zones. The appropriate markings vary based on the intersection, but may include dashed lines, coloured boxes, chevrons and/or shared roadway markings. Blue cycle crossings in Portland, Oregon were estimated to reduce accidents by 10% and injuries by 19% (Jensen 2008). They also increased cyclists' confidence and increased the number of motorists that yielded to cyclists from 72% to 92% (Hunter et al, 2000).

Through bike lanes or combined bicycle lane/turn lanes. Different types of lane markings can be used to delineate the space for bicyclists and motorists as they approach an intersection. Combined bicycle/turn lines use dotted lines within the shared lane to indicate the intended path for through bicyclists (**Figure**, left). "Through bike lanes" provide cyclists with a dedicated lane to enter intersections to the left of right turning vehicles, and are generally the preferred option (**Figure**, right). In Toronto, many bicycle lanes simply disappear before intersections. Terminating bike lanes in advance of intersections is discouraged (NACTO 2011). City of Toronto recognizes the benefits and has installed some markings of this type. The concept has been incorporated into Toronto's the bikeway design and could be expanded and improved so that designs are more consistent City-wide.



Figure 25: Combined bicycle / turn lanes (left) and "through" bicycle lanes (right) Source: NACTO 2011

Bicycle boxes (a.k.a. advanced stop lines) accompanied by educational marketing. Bicycle boxes are painted pavement markings that give cyclists a dedicated space to wait at a signalized intersection, in front of cars. Cyclists in bike boxes get a head start at signals, are more visible and are less likely to be cut off or hit by vehicles. In New York City, the 204 bicycle boxes are also painted bright green to increase visibility.

In Portland, observations revealed an improvement in motorists yielding to cyclists at bike box locations (Dill *et al.* 2012). Most cyclists report feeling safer with bike boxes (Pucher *et al.* 2010, Dill *et al.* 2012). The majority of motorists in Portland also reported that bike boxes made driving safer (Dill *et al.* 2012).

However, without education and marketing, cyclists will not understand the purpose of the bike box and will not feel confident using them (Hunter 2000). NACTO (2011) also recommends that bike boxes be accompanied by signs that reinforce that bicyclists have the right-of-way going through the intersection.

Making active transportation more attractive

In order to increase active transportation mode shares, the City of Toronto could take additional action to make walking and cycling more attractive. Toronto could seek to improve the speed and comfort of walking and cycling routes, and improve access to transit and to end-of trip facilities. The City could also use media, marketing and education programs to overcome social barriers to the use of active transportation (Titze *et al.* 2008, Edwards and Tsouros 2006).

Route quality and speed

Walking and cycling are generally the fastest modes of travel for short trips in urban areas (up to 5 km in length). Toronto can further increase the relative speed of active transportation by developing shortcuts and routes with minimal stops. Pedestrian and cyclist-only shortcuts can shorten trips significantly, particularly in areas where streets have low connectivity, long blocks and/or dead ends (Walker 2009). Bikeway routes may also seek to follow roads with fewer traffic signals. A decrease in the number of stops along a route increases bicycling along the route (Rietveld and Daniel 2004).

However, people's decisions about mode of travel are not just about speed – they are also about comfort and enjoyment of travel. People will not walk unless the experience is

positive. In Toronto, additional effort is needed to ensure that walking and cycling routes are safe, comfortable and enjoyable year-round. The walking and cycling surveys indicate the importance of pavement quality, snow clearance and lighting:

- Repairing potholes and bad pavement ranked second on Torontonians' wish list for improving cycling in Toronto, right behind separating bike lanes from car traffic.
- 39% of Torontonians who do not ride in the winter reported that they would ride more often with safer, cleared bike lanes and paths.
- 38% of Torontonians reported being very unsatisfied or somewhat unsatisfied with the Toronto's sidewalk snow clearance. Satisfaction with Toronto's sidewalk snow removal is significantly lower among older adults (Harris-Decima 2008)
- 17% of Torontonians report being very unsatisfied or somewhat unsatisfied with lighting on Toronto's walkways and paths, and 16% report dissatisfaction with sidewalk maintenance and repair.

In other cities, pavement quality and snow clearance have been identified as important predictors of the use of active transportation. Street lighting can increase pedestrian street use after dark, reduce crime and reduce fear of crime (Painter 1996).

Quality walking and cycling surfaces. Pavement quality is very important to cyclists' rating of road segments (Landis *et al.* 1997). The number of cyclists on a path in London doubled after the path was resurfaced Pucher *et al.* 2010). Paved routes are preferred over unpaved routes, especially by people who cycle frequently (Winters and Teschke 2010).

Sidewalk quality is also important to pedestrians, particularly to those with limited mobility. Paths that are sufficiently smooth, sufficiently wide and have turning radii adequate for a wheelchair or walker are needed to support walking among elderly and disabled people (City of New York 2010).

Snow clearance. Clearance of snow and ice from sidewalks and roads is a key enabler of safe, enjoyable walking in winter. Elderly people and people with impaired mobility are particularly vulnerable in snowy or icy conditions; elderly pedestrians are overrepresented in winter collisions and in winter injuries from falls (Zegeer *et al.* 1993, Merrild and Bak 1983).

Snow clearance is also an important enabler of cycling in winter. Winter reductions in cycling may be countered by snow clearing and sanding or salting of ice along cycling routes, dedicated bike lanes, bike-friendly transit, and education about how to ride safely in inclement weather (Winters *et al.* 2007). For example, Minneapolis has a policy of clearing snow from off-street trails soon after the end of a snowfall – generally, in less than 24 hours. The off-road bicycle paths that form the backbone of their bikeway network can thus be used all year long. Minneapolis has very high rates of winter cycling.

Transit integration

All aspects of the Greater Toronto Area's transportation system will face increasing stress as 100,000 people move to the area each year. Avoiding doomsday predictions of total gridlock in the City of Toronto will require significant investments in regional transit, active transportation, and modal integration (Kalinowski 2011a).

The integration of transit and active transportation is particularly important given the increasing number of trips that start or end outside the City of Toronto. Between 1985 and 2006, the number of inbound vehicle trips between 6:30 and 9:30am increased by 75%, and outbound vehicle trips increased by 79% (City of Toronto Planning 2007). Many of these

vehicle trips are too long to be made by walking or cycling alone, but could effectively be replaced by active transportation in combination with transit.

Replacing segments of transit trips with walking and cycling trips can also help to ease the increasing strain on Toronto's struggling public transit system. According to Metrolinx predictions, transit trips through Union Station — already the country's busiest transit hub — could nearly triple by 2031 (Kalinowski 2011b).

Research supports the possibility of increasing walking-transit trips through targeted investment in the comfort and convenience of these integrated trips. Creating safe, visible and comfortable transit facilities can support increases in transit use - and therefore in walking to transit (Sutherland and Carlisle 2007, Wener and Evans 2007). Almost all transit users in Toronto walk to and from transit (92%), so it is critical that Toronto ensure safe and convenient pedestrian access to transit.

There is also substantial evidence that better integration of bicycling with public transport leads to more "bike and ride" trips, and probably to more bicycling overall (Pucher *et al.* 2010). Toronto surveys suggest that improving the convenience of bike-and-ride trips could dramatically increase bike-and-ride trips in Toronto. For example, almost three quarters of Toronto cyclists (74%) and more than half of recreational cyclists (66%) report that they would combine cycling and transit more often if stations featured secure bicycle parking (Ipsos-Reid 2009).

The City of Toronto is already taking action to support active transportation in and around transit, particularly in the downtown area:

- The PATH underground walkway system integrates walking, local transit and regional transit.
- The new bicycle station at Union Station provides secure bicycle parking and other cycling-related services to Toronto residents and commuters.
- Streetcars and subways allow bikes during off-peak hours, and the TTC Rack It Rocket program is aiming to install bike racks on all buses by the end of 2012.

Toronto can build on these exciting efforts with additional investments in transit station areas, in bus stops and in secure bicycle parking. Of course, these efforts must be accompanied by investments in the public transit system itself.

Secure bicycle parking. A study of pilot projects in the Netherlands in the 1990s found that bicycle parking significantly increased bicycle trips between homes and rail stations (Martens 2007). In the Washington D.C. area, bike-and-ride trips increased by 60% between 2002 and 2007 after the Washington Metropolitan Area Transit Authority invested in bicycle parking at Metrorail stations (DDOT 2011). All stations in the Area now have bicycle racks and almost 50 stations have bicycle lockers available for rent.

Transit station areas. The Ministry of Transportation of Ontario's (2012)Transit-Supportive Guidelines provides detailed recommendations for enhancing pedestrian and cyclist access in station areas, including:

- Treating the sidewalks in and immediately adjacent to a station as a pedestrian priority area, with appropriate snow clearance, lighting, seating, waste receptacles and landscaping.
- Avoiding cycling barriers such as curbs or stairs where possible.
- Keeping pedestrian and cyclist access routes separate to minimize conflicts.

Bus stops. The City of New York (2010) recommends bus bulbs and wider sidewalks at bus stops to better accommodate pedestrian traffic and transit use. Bus shelters with seating or places to lean are important for accommodating the elderly and other users who have trouble standing for extended periods.

End-of trip facilities

Secure bicycle parking and showers are important enablers of cycling, particularly to work and to school. An analysis of UK National Travel Survey concluded that universal outdoor bicycle parking would raise the bicycle commuting mode share from 5.8% to 6.3%, secure indoor parking would raise the mode share to 6.6%, and indoor parking plus showers would increase it to 7.1% (Wardman *et al.* 2007). In Calgary, 88% of commuter cyclists use some type of change room or shower facility (City of Calgary 2007).

In Toronto, 68% of work/school cyclists report having access to convenient and secure bicycle parking, as compared to only 46% of recreational cyclists. 57% of cyclists report that secure indoor bicycle parking would improve cycling in Toronto a great deal, and 39% said the same about shower / change facilities. 45% of non-cyclists and 43% of recreational cyclists stressed the importance of secure indoor bicycle parking (Ipsos-Reid 2009).

The Toronto Green Standard will dramatically improve the availability of end-of-trip facilities in new developments. In place since 2010, it requires that new developments meet requirements including:

- Cycling infrastructure—provision of short-term bicycle parking, long-term bicycle storage, and shower and change facilities
- Pedestrian infrastructure—provision of sun and weather protection for pedestrians, building integration with walking routes, grading and surface treatment, signage, lighting, and separation from vehicular routes and air intake or exhaust systems
- Transit accessibility—provision of building integration with transit facilities, proximity to transit service, and direct pedestrian linkage to transit

The City can continue to require, encourage and promote the provision of secure bicycle parking and access to shower facilities in public spaces and in existing private developments.

Marketing and education programs

While interventions in the built environment generally receive greatest attention, it is also critical to understand barriers to physical activities that are rooted in people's attitudes and habits (Lavizzo *et al.* 2003). Marketing and education programs to promote walking and cycling play an important role in overcoming barriers to cycling and walking that go beyond the physical environment (Pooley *et al.* 2010, Pucher *et al.* 2010).

Toronto has developed a number of marketing programs for cycling including: Bike Month, Cycling Ambassadors, Bicycle-Friendly Business Awards, and the Kids CAN-BIKE camp. Toronto also launched a public bike system (BIXI) in May, 2011, that has resulted in over 420,000 cycling trips to date. Toronto's programs to promote walking include numerous promotions during bike month, pedestrian newsletters, pedestrian pilot projects, walking tours and street festivals, promotion of Explore Toronto's parks and Trails and Discovery Walks maps, Toronto Public Health's Walk into Health Program and the pedometer lending program at Toronto Public Library. However, Toronto can do more to develop marketing programs that cost-effectively increase walking and cycling, improve safety and reduce car use. Toronto could also consider the role of the primary and secondary school system in promoting safe, healthy and sustainable transportation.

Marketing programs

Media campaigns, special events, trip reduction programs, individualized marketing programs, incentive programs, bicycle access programs and travel awareness programs can all be used to shift car trips to transit, walking, and bicycling. For example:

- Bike-to-Work days have been associated with long-term increases in cycling. In San Francisco, bicycle counts remained 25.4% higher for one month after a Bike-to-Work event; in Victoria, Australia, over one quarter of first-time cyclists were still bicycling five months later (Pucher *et al.* 2010).
- Marketing programs in South Perth, Australia, increased public transportation by 17%, cycling by 61% and walking by 35% among area residents changes that were sustained two years after the program ended (Thogersen 2007).
- Community-level marketing programs in Portland and in Washington State have contributed to long-term increases in the share of trips made by walking, bicycle and transit (Cooper 2007, Pucher *et al.* 2010).
- Signage for pedestrians and cyclists can provide directions, distances and times to various destinations. They can also encourage people to walk an extra transit stop or two (City of New York 2010).
- In the BikeBusters bicycle access pilot program in Arhus, Denmark, participants were given a new bicycle and bus tickets free for a year, in exchange for signing a contract promising to reduce driving. Bicycling for everyday trips increased from 8% to 40%, while bicycling to work increased from 15% to 60% (Pucher *et al.* 2010).

Because of their relatively minimal cost, marketing programs are extremely cost-effective, particularly at the city- or community-level (UWE and Cavill and Associates 2007). One study found benefit-cost ratios exceeding 20:1 (Buchan 2009). The UK reported that walking information packs and the Sustainable Travel Towns marketing programs reduced car use at a cost of \$63 to \$69 CAD per 1,000 car kilometres (Ashcroft *et al.* 2011).

Mass-media marketing is being effectively and creatively used in a number of Canadian cities. For example, in Ottawa, the iWalk iBike social marketing campaign challenges residents to leave their cars at home and try active forms of transportation. The campaign includes widespread advertisements that underscore the advantages of walking and cycling instead of driving. The campaign also features an interactive website where residents can register and track their distance travelled, calories burned, reduction in greenhouse gas emissions, and money saved by using less fuel. Residents can earn badges for their active trips, and can monitor citywide reductions in car trips. The total cost of Ottawa's iWalk iBike campaign is only \$90,000, \$40,000 of which is covered by a grant from Ontario's Transportation Demand Management Municipal Grant Program.

Education programs

Safe, healthy transportation habits can start at a young age. In both Germany and the Netherlands, curriculum-based education on safe walking and cycling encourages habitual and safe active transportation in the next generation of citizens. All schoolchildren have received extensive instruction on safe walking and cycling by age 10. In addition to learning traffic regulations, they learn how to walk and bike defensively, and how to anticipate and react to dangerous situations (Pucher and Dijkstra 2003).

Other Canadian cities are looking to Europe's educational programs for inspiration. In Montreal, curriculum-based cycling education is now reaching a small (but growing) percentage of elementary school children. In Vancouver, the Cycling Training for the School Community (CTSC) program promotes active transportation by offering bicycle commuting training to Vancouver teachers, support staff and school administrators.

In the Netherlands and Germany, rigorous driver education also contributes to safe walking and cycling conditions. Car drivers must assume that pedestrians and cyclists will make unsafe (and illegal) moves in traffic, and are required to anticipate such unsafe moves. The ability to anticipate potentially unsafe moves by pedestrians and cyclists is included in the driver's license exam (Pucher and Dijkstra 2000).

Enabling active transportation through land-use planning

The City of Toronto can increase the feasibility of walking and cycling for all Torontonians by promoting higher density, mixed-use neighbourhoods with high street connectivity and pedestrian-friendly urban design in Toronto's high-growth areas – as envisioned in the city's Official Plan and supported by tools such as the Streetscape Manual.

Destination accessibility, street connectivity, density, and urban design

Land use patterns play an important role in enabling active transportation, since trip distance is often the most significant limiting factor. This is particularly relevant for walking trips. Trip origins and destinations are more likely to be sufficiently close together in neighbourhoods with more destinations and a greater mix of land uses, in neighbourhoods with connected, grid-like street layouts, and in higher density neighbourhoods.

There is substantial evidence to support these trends. For example, in Montreal, women aged 45 and older living in neighbourhoods with higher "destination density" were 53% more likely to walk at least 30 minutes per day, 5 days per week (Gauvin *et al.* 2008). In the United States, increasing the number of business types in a neighbourhood from three to four increases the number of walking trips by 24% (Boer *et al.* 2007). One study found that urban sites with small blocks and extensive sidewalk systems had, on average, three times the pedestrian volumes of suburban sites with long blocks and short, incomplete sidewalk systems (Hess *et al.* 1999). The Transportation Research Board (2009) concluded that doubling residential density reduces household vehicle miles traveled (VMT) by 5-12% in the US. When combined with mixed-use developments and improvements to public transit and employment concentration, VMT reductions reach 25%⁸.

There is also evidence that street-level urban design significantly increases levels of walking and cycling (Forsyth *et al.* 2008, Larsen *et al.* 2009, Ewing and Handy 2009). Pedestrian-

⁸ Though reductions in VMT are not directly associated with increases in walking and cycling, they are linked.

friendly streetscapes often include public spaces, benches, fountains, restrooms, art installations, vegetation and effective exterior lighting. The City of Toronto's Streetscape Manual and Vibrant Streets Guidelines (for street furniture) help guide the quality of design and experience of the city's streets and built environment for pedestrians. Trees and vegetation can provide shade and separate pedestrians from traffic. Benches, drinking fountains and restrooms are particularly important for seniors (Lockett *et al.* 2005).

In Toronto, the different land use patterns in the core and in the suburbs lead to dramatic differences in walkability **Figure 26** illustrates the "walkability" of different areas in Toronto based on a land use index that considers: land use mix (destination accessibility), intersection density (route connectivity), residential density, and floor-to-area ratios for retail land (residential and retail density). The former City of Toronto features more mixed use neighbourhoods with high connectivity, high residential density, and high retail density – factors that increase "walkability".



Figure 26 Walkability in Toronto based on the UD4H walkability index **Source**: Urban Design 4 Health (UD4H) and Toronto Public Health (2012)

Neighbourhoods in the former City of Toronto also tend to feature highly engaging streetscapes and pedestrian-friendly urban design. In contrast, many neighbourhoods in the suburbs are characterized by wide streets, long city blocks with little visual richness, and few engaging public spaces (Chapter 3).

Efforts to enable long-term increases in active transportation will be most effective if they also focus on land use. In Toronto, the greatest opportunities exist in the 25% of the City where the most growth will occur over the next 30 years: Toronto's five "Centres" (Downtown/Waterfront, Scarborough Centre, North York Centre, Etobicoke Centre and Yonge-Eglinton Centre), Avenues, Employment Districts and Port Lands. The Official Plan's (2010) goal of focusing growth into a pattern of compact mixed use communities and corridors is well aligned with increases in active transportation mode shares.

Toronto can build off of the City's Green Standard and ensure that walking and cycling are enabled by land use and roadway design in all of the areas targeted for growth. Toronto can

also invest in gradual revitalization and redevelopment in the low-walkability neighbourhoods that are home to large numbers of low-income households (Chapter 4). "Mainstreet" programs, business improvement areas, grants, and incentives can increase jobs and destinations in Toronto's least walkable neighbourhoods. For example, in Washington D.C., the Great Streets Initiative is a multi-agency effort to transform nine underinvested corridors into thriving neighbourhood centers through investment in mixed use development projects, storefront improvements, transportation, streetscape, and transit improvements (District of Columbia 2012).

These types of efforts are supported – and required – under the Government of Ontario's Growth Plan for the Greater Golden Horseshoe Area. The 2006 Growth Plan is a regulatory regional planning framework that requires the development of vibrant, compact, mixed use, transit-supportive communities with pedestrian-friendly urban environments and high quality open spaces. Planners must offer multi-modal access to destinations, and provide connectivity among modes (Government of Ontario 2006).

Making active transportation more accessible

The City could make special efforts to ensure that safe walking and cycling is accessible to all Torontonians by developing affordable housing in central areas, improving walking and cycling infrastructure in areas with vulnerable users, and developing programs to promote walking and cycling at the community level.

Walking and cycling facilities for vulnerable users

Interventions that increase the year-round safety and convenience of walking, cycling and transit use tend to support non-drivers and economically disadvantaged people (Martens 2006). Unfortunately, as discussed in Chapter 4, Toronto's current active transportation patterns and facilities appear to create transportation and health inequalities. Young pedestrians in Toronto appear particularly likely to be hit by motor vehicles, while elderly pedestrians are particularly at risk of being killed. Toronto's low-income neighbourhoods may also have lower levels of walkability and bikeability.

To improve health and transportation equity, Toronto could prioritize the development of safe walking and cycling facilities for vulnerable users: young people, older adults, and residents of low-income neighbourhoods. Traffic calming strategies are particularly valuable for the elderly, for children and for other vulnerable users (City of New York 2010, Preston 1995). Traffic calming in the UK was associated with an increase in the proportion of children walking to school (UK Department for Transport 2010).

In areas around schools and with large numbers of senior citizens, it is also important to provide continuous sidewalks cleared of snow, generous crossing times, and appropriate pedestrian-scale lighting and signage (City of New York 2010, OPPI 2009a, 2009b). In Toronto, older adults are significantly less likely to be satisfied with intersection crossing times and snow clearance (Harris-Decima 2008).

In New York City, the Age-Friendly City initiative will lead to a wide variety of physical transformations to improve city life for older adults. Initiatives will include: more seating in bus shelters, more public restrooms, more elevators and escalators, "age-friendly" parks, and improved safety at intersections. Three neighbourhoods have been named Aging Improvement Districts as part of a pilot program to implement changes requested by seniors.

The initiative has also won eager cooperation from local businesses, institutions and elected officials (Herndon 2012).

Toronto may also seek to address equity concerns by improving walking and cycling environments in and around low-income neighbourhoods. Safe walking and cycling routes could be developed to connect residents to destinations in their neighbourhood and in adjacent areas of the city. Additional crosswalks may also be needed in the outer areas of Toronto where long city blocks lead to higher rates of jaywalking, and pedestrian-vehicle collisions at mid-block locations (Hess et al. 1999, Hess and Farrow 2010). Unsafe intersections and crossings could be identified and enhanced, and streetscape improvements could be prioritized in communities where benches, trees, and human-scale design are particularly lacking. Finally, the City can address safety concerns at night by improving lighting on paths and around transit.

In Montreal, the "Quartier 21" programme provides funding to community groups to improve safe and active transportation in low-income neighbourhoods. Each year, four projects are awarded grants totalling \$130,000 over a three-year period (Ville de Montreal 2012).

Affordable housing in vibrant communities

Residents of multi-modal communities with high destination accessibility tend to spend less time and money on transportation. However, high demand may reduce housing affordability in these same communities unless special efforts are made to ensure a supply of lower-priced housing (Malefkalazi 2010).

These trends are apparent in Toronto, where many low-income families often live in the outer areas where active transportation is least feasible. Toronto's housing market is among the least affordable in Canada, second only to Vancouver. High prices and low rental vacancy rates are particularly problematic in the (walkable and bikeable) former City of Toronto (Toronto Community Foundation 2011). Toronto could make a special effort to increase the supply of affordable housing in neighbourhoods with high destination accessibility. Increasing the supply of affordable housing in walkable and bikeable locations may help to address health disparities (Chapter 4).

Different types of strategies could be used. For example, inclusionary zoning requires or encourages developers of market residential projects to construct some proportion – generally 10 to 25% – for affordable housing. Inclusionary zoning is used by 200 or so municipalities in the United States.

In Minneapolis, the Corridor Housing Strategy aims to foster affordable housing growth along the city's transit corridors, using three innovations:

- Planning giving neighbourhood organizations an early and comprehensive role in planning affordable housing development.
- Site acquisition acquiring critical sites on transit corridors.
- Funding awarding priority housing funding to development that is tied to jobs and transit (City of Minneapolis 2011).

Culturally competent community-based programs

Well-designed programs to increase physical activity in low-income communities have successfully increased walking, and had associated social benefits. For example, a walking

promotion program in a low-income housing complex in Seattle sponsored walking groups, improved walking routes, advocated for pedestrian safety, and provided information about walking options. Walking group participants increased their walking levels from 65 to 109 minutes per day and also reported an increase in the number of neighbours they knew (University of West of England and Cavill and Associates 2011).

However, planners must focus on "culturally competent" planning – planning that begins by understanding active transportation from the perspectives of these communities (Day 2006). For example, do people travel to shopping and other destinations in groups? Are safety concerns a limiting factor? For efforts that focus on specific minority populations, there is a need for a delicate balance between embracing group customs and values, and recognizing the non-monolithic nature of any group (Yancey 2006). Communities themselves should be centrally engaged in developing strategies to support active living, and planners should provide a menu of program choices to fit different needs (Day 2006).

In Toronto, CultureLink, a settlement organization, is among the organizations seeking to promote healthy lifestyles in immigrant neighbourhoods. The City could do more to promote and enable active transportation in low-income and immigrant communities. In a survey conducted by the Community Cycling Centre in Portland, 60% of respondents expressed concern about the cost of a bicycle, indicating that such an expense would be outside their means (Ersoz 2011). In Toronto, programs that connect low-income families to free or low-cost bicycles or Bixi memberships may significantly reduce this first and most formidable barrier to cycling for transportation.

Implementing coordinated packages of actions

The City can maximize results by ensuring that its actions are integrated and coordinated. Multi-level interventions that target individuals, social environments, physical environments and policies can best achieve population-level changes (Sallis *et al.* 2004). A coordinated package of complementary infrastructure measures, programs, and policies may enhance the impact of any intervention that is a component of that package (Pucher *et al.* 2010).

For example, in the City of Paris, there was a 150% increase in the bicycle share of trips in only six years – the bicycle share of trips increased from 1% in 2001 to 2.5% in 2007. This dramatic increase can be linked to a number of coordinated actions:

- The bike lane network more than tripled from 122 km to 399 km from 1998 to 2007. Bicycling parking also tripled between 2000 to 2007.
- The City implemented extensive traffic-calmed areas, car-free zones, narrowed roadways and widened sidewalks, and "civilized travel corridors" of restricted motor vehicle access.
- The Velib bicycle sharing program started in 2007, and is now the world's largest bicycle sharing program (Pucher *et al.* 2010).

In the UK, the Sustainable Travel Towns Initiative led to a 9% decrease in the number of car driver trips, a 10-13% increase in walking trips, and a 26-20% increase in cycling trips by residents of the Sustainable Travel Towns. Walking and cycling to school increased by up to 7%, and average distances walked by residents increased by 18-27% (Sloman *et al.* 2010). The Initiative included a suite of programs including improvements to pedestrian infrastructure and the public realm; personalized, school and workplace travel planning programs; and cycling and walking media promotion and brand identity.
The City of Whitehorse's 2006-2007 Wheel 2 Work program included a four-month marketing campaign and a \$2 million municipal infrastructure investment. Interventions in the built environment included: improved trail linkages between downtown and residential areas, a new pedestrian and bicycle bridge, improved lighting and staircases, on-street bike lanes, and new artisan-designed bicycle racks in the downtown area. Marketing and education included: incentive prizes for cycling to work, online bicycle route maps and safety information, bicycle maintenance and repair workshops, and improved safety signage for bicycles along major arterials. In 2006, 40,000 km of bicycle travel were logged, offsetting 4.5 tonnes of GHG-eq. (Ministry of Municipal Affairs and Housing and Ontario Professional Planners Institute 2009).

Prioritizing Transportation Infrastructure Funding in Toronto

Finally, the City can maximize the tremendous potential health benefits of walking and cycling by prioritizing its investments in transportation infrastructure. As discussed in Chapter 2, only 5% of the total capital expenditures for Transportation Services are dedicated to improvements in walking and cycling infrastructure.

Other cities who have invested in infrastructure are outpacing Toronto. For instance, the City of Montreal has made significant investments to improve conditions for cycling, and this has yielded impressive results. The combined mode split in Montreal is 11%, about 3.2% higher than in the City of Toronto.

Chapter 5 Findings

- In order to improve public health, Toronto should make targeted efforts to increase safety for
 pedestrians and cyclists. Toronto could also work to make walking and cycling more attractive,
 more feasible, and more accessible to residents across the city; increases in safety and in mode
 share tend to go hand-in-hand.
- Safety for Toronto users of active transportation can be improved through targeted action to reduce vehicle speeds, separate cyclists from vehicles, improve intersections for pedestrians and cyclists, and increase safety on major arterials.
- Many of these options require only pavement markings or signs, and are relatively inexpensive to implement (e.g. new approaches to bike lanes at intersections, new speed limit signs). Others are more expensive but important for Toronto to keep pace with other cities (e.g. separated bike lanes).
- The City can also increase the attractiveness of active transportation by improving snow clearance, pavement quality, short-cuts, and lighting; by improving access to transit and end-of-trip facilities; and by developing marketing and education programs.
- In order to make active transportation feasible for all Torontonians, the City can invest in mixed use, compact communities in high-growth areas and gradual redevelopment of neighbourhoods with low walkability and bikeability.
- Making active transportation accessible to residents of all ages, abilities and incomes will require targeted investments in walking and cycling facilities, in affordable housing in accessible locations, and in community-based programs.
- Coordinated packages of interventions in the built environment and in the social environment can produce the greatest results.
- The City can also maximize the tremendous potential health benefits of walking and cycling by prioritizing its investments in transportation infrastructure.

6. Facilitating effective action in Toronto

This chapter draws on interviews with municipal staff in other jurisdictions to identify strategies and capacities that underpin effective action on active transportation. It discusses the role of visions, goals and targets, and enabling policies, plans and standards. It also discusses the role of data, evaluation tools, partnerships, community support and funding in enabling effective implementation.



Setting goals

Toronto's 2009 Walking Strategy articulates a well-developed vision of a city with a culture of walking, where streets, parks, public spaces and neighbourhoods are accessible, secure, vibrant and enjoyable so that people choose to walk more often, including in combination with other sustainable modes of travel. The 2001 Bike Plan identifies the vision of a "safe, comfortable and bicycle friendly environment in Toronto, which encourages people of all ages to use bicycles for everyday transportation and enjoyment". Toronto needs to continue to set goals to guide and expand its actions on walking and cycling from 2012 onwards.

Developing quantitative goals that are supported by Council, by City staff, and by community stakeholders have helped other cities increase their effectiveness in promoting safe walking and cycling. For example:

- Chicago's goals are to increase bicycle use so that 5% of all trips less than 8 km are by bicycle; to reduce the number of bicycle injuries by 20% from current levels, and to increase the number of bike-transit trips by 19% each year (City of Chicago 2006).
- New York City's goal is to reduce traffic fatalities by 50% by 2030 (New York City DOT 2012).
- Calgary's goal is to increase the downtown walking mode share to 11% and the downtown cycling mode share to 4% by 2020 (City of Calgary 2009).

Developing plans, policies and standards

In many other cities, policies and standards ensure that the needs of pedestrians and cyclists are consistently considered in planning and decision-making at all scales – from the city- and community-scale, down to the street- and site-level.

For example, the City of Toronto could proactively identify opportunities for pedestrian and cyclist facilities as part of every road reconstruction or resurfacing project. Every "secondary land use plan" is an opportunity to specifically address the needs of pedestrians, cyclists, and users of all ages and abilities. As of 2006, few Toronto secondary plans addressed active transportation (Hess and Milroy 2006). Zoning bylaws, site plan controls, and development permit systems are tools that can be used to ensure that active transportation is routinely considered in decision-making (MMAH and OPPI 2009).

Policies and standards can also be designed for adaptability to different areas of a city. For example, New York City is developing different "complete streets" design templates for different streets, ranging from walking-only streets to major bus and truck routes (NYC DOT 2008). This approach may be particularly relevant to Toronto. As described in Chapter 4, land use patterns, roadway design, and current levels of walking and cycling are dramatically different in the former City of Toronto and in the suburbs. In downtown Toronto as in New York City, not all roads can feature a bike lane, dedicated transit lane and lanes for motor vehicles; considering the needs of all roadway users may thus involve eliminating car travel from some streets and improving conditions for car travel on others.

Priorities for improving pedestrian and cyclist safety may also be dramatically different in the central and outer areas. For example, in outer areas, safe, visible crosswalks may be needed to reduce collisions along long city blocks. In inner areas, leading pedestrian phases may be needed to reduce conflicts with returning vehicles. The City can consider these differences in developing policies, plans and standards.

Collecting better information

Existing Toronto data sources do not provide a clear picture of the number of people who walk and cycle, the frequency with which they walk and cycle, or the distances walked and cycled. This is not a unique situation. In many North American cities, professionals doubt that available data can produce reliable estimates of current levels of walking and cycling (Fenton and Junot 2010). According to City of Portland staff, "one of the greatest challenges facing the bicycle and pedestrian field is the lack of documentation on usage and demand" (Birk and Geller 2005).

Without accurate and consistent data, it is very difficult to measure the benefits of investments in walking and cycling, including public health benefits. Better information on walking and cycling travel patterns can also improve infrastructure and program planning, help to prioritize walking and cycling routes, and enhance a city's ability to track changes over time. Finally, data on walking and cycling demand and use can help in evaluating the adequacy of funding of these modes.

Bicycle facility planning and evaluation in Toronto could be significantly enhanced through regular data collection on cycling across the city. The downtown screenline bicycle count conducted in September 2010 provided a valuable baseline of cycling in the downtown core. However, it was not repeated in 2011, nor was it accompanied by counts of cyclists outside the downtown core. Toronto's annual downtown bicycle count could be supplemented with additional data:

- Manual or automated bicycle counts in locations across the city
- Regular counts for all bike lanes and paths, including "before" and "after" counts for new lanes and paths.

For example, in Vancouver, manual bicycle counts are regularly conducted at more than 100 key intersections throughout the city, and eight permanent bike counters supplement this manual data. The City is further seeking to develop a comprehensive cycling monitoring program to provide baseline data to define the next 10-year Cycling Program Master Plan, to enable timely reporting of cycling statistics, and to allow successful evaluation of the new Cycling Plan (City of Vancouver 2010).

Data on walking and cycling in Toronto is severely limited by the Transportation Tomorrow Survey's exclusion of walking trips other than to work or to school. New types of data and uses of data could also enhance pedestrian project planning including:

- Discretionary walking trips in the Transportation Tomorrow Survey could help to address the deficit of data on walking in Toronto.
- Mid-block counts on streets across Toronto could provide a useful supplement to current counts of pedestrians at intersections.

Finally, Toronto has very few data on walking, cycling, collisions and injuries among lowincome, immigrant and minority groups. Toronto also has limited qualitative information on barriers and perceptions of active transportation in different parts of the city. Both qualitative and quantitative data can help planners develop infrastructure and programmatic measures to support transportation and health equity. The improved understanding of travel patterns and barriers can also help planners engage with communities on priority issues and partner with appropriate community organizations.

Tools to improve decision-making

The use of walking and cycling-specific modeling tools can increase the likelihood that infrastructure investments deliver the maximum benefits (Moudon and Lee 2002). The general scarcity of walking and cycling evaluation tools is particularly notable when compared to the decades of experience modeling demand, usage and benefits of other modes (Birk and Geller 2005).

Different types of new and modified tools are needed. For example:

- Cost-benefit and cost-effectiveness evaluation tools that consider a broader range of costs and benefits, including those related to health.
- Tools that evaluate the quality of walking and cycling routes and identify intersections and road segments for safety audits.
- Transportation models that assess latent demand for walking and cycling, and forecast increases in demand.
- Models that consider a wide range of behavioural responses in assessing the impacts of interventions (e.g. roadway re-allocation projects), based on evidence from other cities.
- Tools that evaluate the impacts of land use and development on opportunities for active living.

Like other types of decision support tools, these tools alone are a starting point. They can identify problem areas, and point towards the type of infrastructure that is appropriate in a given location. They can more accurately assess the importance of investment in active transportation. Finally, they can provide a shared starting point for conversations about problems, priorities and solutions.

For example, Copenhagen developed a simple tool to evaluate the need for separated bike lanes (cycle tracks) and on-street bike lanes based on daily motor vehicle traffic and posted speed limits (Figure 27). The model states that below speed limits of 40 km/h and vehicle traffic of 2,000 vehicles per day, cyclists ride in mixed traffic. Cycle tracks are developed on high speed roads with relatively low traffic (e.g. 55 km/h and 6,000 vehicles per day), and on lower speed roads with very high traffic flows (e.g. 35 km/h and 12,000 vehicles per day).



Figure 27 Tool to plan cycling facilities in Copenhagen based on vehicle speed and volume **Source**: Jensen 2000

In many cities including Portland and Washington D.C., evaluation tools also link collision analyses to project prioritization for improvements to intersections, crosswalks, and sidewalks:

- The City of Portland developed the "Pedestrian Potential Index" and the 'Pedestrian Deficiency Index" to highlight priority areas for improvements in the walking environment. The Pedestrian Potential Index on a given roadway segment is determined using proximity to pedestrian "attractors/generators", anticipated growth in population and employment density near the location, and scheduled transportation improvements. The Pedestrian Deficiency Index identifies problematic roadway segments based on sidewalk characteristics, traffic, crossing conditions and historical collisions.
- Washington D.C.'s 2009 Pedestrian Plan identified eight high priority corridors (one in each ward) using modified versions of the *Pedestrian Potential Index* and the *Pedestrian Deficiency Index*. Recommendations for improvements included sidewalk repair and construction, relocation of bus stops, signalization changes, and reduced crossing distances at intersections. The use of the two indices helped the transportation department identify and prioritize projects with the maximum potential benefit (Washington transportation staff 2011).

Finally, several municipalities have adopted tools that specifically consider how planning and development decisions affect opportunities for active living and active transportation. For example, San Francisco's Healthy Development Measurement Tool assesses how urban planning affects community health. The Tool includes a checklist of over 100 community health indicators, and a menu of policies and design strategies.

Enhancing partnerships

Collaboration across municipal government increases the likelihood of developing a common vision, securing solutions that address this vision and implementing actions needed to implement that vision (MMAH and OPPI 2009). Efforts to enable active transportation are thus enhanced through cooperation between municipal planning, transportation, public health, housing, social services, and parks departments (Edwards and Tsouros 2006).

The health sector is well positioned to lead collaborative initiatives to promote active transportation. In New York City, the public health department played a key role in coordinating and advancing the Active Design Guidelines, developed in partnership with Design + Construction, City Planning, and the Office of Management and Budget. All departments were able to rally around the goals of reducing congestion, improving health and/or improving environmental sustainability, and provided diverse technical knowledge that was important for success (New York City public health staff 2011).

Public health departments can also develop and implement their own initiatives that are complementary to the work of other departments. In Minneapolis, the public health department's community active living programs is coordinated with planning and transportation programs but are run separately (Minneapolis transportation staff 2011).

Finally, public health departments can recommend that health benefits and risks be considered in transportation and land use decisions. Public health practitioners can work with transportation engineers and city planners to develop tools that evaluate health benefits and risks, and can facilitate conversations about the factors that promote safe walking and cycling.

Building community support

The private and voluntary sectors are important potential partners in promoting active transportation (Edwards and Tsouros 2006). Cities including Calgary and Seattle have effectively used community engagement, outreach to local businesses and the energy of the non-profit sector to generate support and resources for active transportation projects (Calgary transportation staff 2012, Seattle Department of Transportation 2011).

In Toronto, residents and business owners are generally supportive of investments in walking, cycling and the public realm. For example, two studies conducted in Bloor Street West neighbourhoods indicate that visitors, residents, business owners and managers recognize the value of walking and cycling for the local economy and quality of life (Sztabinski 2009, Forkes and Smith Lea 2010). Toronto's non-profit organizations dedicated to walking and cycling are also valuable potential partners.

Finally, municipal experience indicates that engaging community stakeholders can reduce time wasted on conflict (MMAH and OPPI 2009). For example, the success of traffic calming projects often relies on the involvement of community members in decision-making (City of Calgary 2006). Engaging stakeholders in the earliest stages of active transportation project planning reduces the likelihood of opposition from local groups (NIMBY-ism) and increases the likelihood that the project parameters will meet the needs of all stakeholder groups.

Seeking funding from other levels of government

The Ontario Ministry of Transportation is increasingly recognizing the value of active transportation. Last year, the Government of Ontario's Transportation Demand Management (TDM) Municipal Grant Program awarded grants of up to \$50,000 to 21 municipalities for TDM, including active transportation (for a total investment up to \$1 million). Nonetheless, the City of Toronto remains primarily responsible for planning and funding improvements to active transportation infrastructure.

The City of Toronto's investments undoubtedly pay off as walking and cycling projects improve quality of life, enhance the city's competiveness, and reduce the strain on the transportation system. However, the health benefits and avoided costs of increased physical activity from walking and cycling are primarily accrued at the provincial and federal levels through reduced health care costs.

The City of Toronto could seek funding from the provincial government for investments in walking and cycling that reduce mortality, chronic disease, and health care costs. The City could also use its influence to push for changes to provincial health and transportation legislation, funding, and programs to support cost-effective investments in active transportation. Finally, the City could work with provincial and federal government agencies to develop new programs that enable municipal innovation and leadership in healthy community planning, transportation equity and active transportation infrastructure.

A number of federal, provincial and state programs in Canada and the United States aim to promote active transportation by providing funding for municipal action. Some focus on land use and street-level planning while others support investment in infrastructure and programs. Examples are highlighted below.

Quebec Route Verte. The Government of Quebec collaborated with Velo Quebec to invest \$88.5 million over ten years to create a 4,000 km bikeway all throughout the province, making it the most extensive bicycle route in all of North America. The Ministry of Transportation administers the Route Verte construction assistance program that provides funding for up to 25% of the eligible costs of the project.

British Columbia BikeBC. Bike BC is a \$31 million program with the objective of improving cycling infrastructure across British Columbia. It includes the Cycling Infrastructure Partnerships Program (CIPP), a cost-shared program for the construction of new transportation cycling infrastructure. All British Columbia municipalities and regional districts are eligible to apply for up to \$100,000 in CIPP funding.

U.S. Partnership for Sustainable Communities. The US Partnership for Sustainable Communities is collaboration between the U.S. Departments of Housing and Urban Development, Transportation and the Environmental Protection Agency. The Partnership offers grants for integrated sustainable community planning, emphasizing the importance of investing in healthy, safe and walkable neighbourhoods; allowing people to live closer to jobs and school; and increasing regional sustainability.

Minnesota Statewide Health Improvement Program. Minnesota's Statewide Health Improvement Program (SHIP) awards grants to fund projects that promote physical activity and health at the community- or population-level. Examples include: interventions to offer more time in physical education, community design to promote biking and walking, and policy changes to provide time off during working hours for physical activity.

U.S. Safe Routes to School. In 2005, the US federal government established a grant program to support projects that improve the safety of walking and biking routes and encourage children and parents to walk and bike between home and school. As of 2011, 11,371 schools in 50 states have participated. Though few studies have measured the program's effects, in a California study, a parental estimate suggested 10% increases in walking and bicycling for participating schools, and also decreases in injury rates (Orenstein *et al.* 2007).

Benchmarking other jurisdictions

The City of Toronto's Official Plan (2010) describes a successful city as "one with a competitive advantage over others locally, nationally and internationally. It has a quality of life that will attract and retain people who have capital, skills, knowledge, ingenuity and creativity." Benchmarking the active transportation plans, policies and standards in other jurisdictions in Canada, the United States and Europe could help to enhance Toronto's competitive advantage.

Toronto may be inspired and challenged by the aggressive plans of other cities – for example, New York City plans to redesign almost 100 km of streets for greater pedestrian safety as part of their goal of reducing fatalities by 50% (NYC DOT 2012). Through benchmarking and review of best practices in other jurisdictions, Toronto may also be able to implement the actions and measures that have proven most effective in other cities – both those that are relevant to Toronto's downtown and those that relevant to the suburbs. Toronto may improve its ability to monitor the health and transportation impacts of actions using methods pioneered in other cities. Finally, Toronto may be able to build the case for investment in active transportation based on the outcomes of action in other cities.

Chapter 6 Findings

- In cities including Montreal, Chicago and New York, goals and targets for active transportation safety and/or mode shares have served as important stimuli for action. Toronto could set new targets for active transportation safety and mode share.
- New policies and standards could enable action by requiring that walking and cycling are considered in planning and decision-making at a variety of scales.
- Cities including Vancouver have recognized the importance of high-quality data for effective investment in walking and cycling. Toronto could enhance planning, monitoring and evaluation by improving the quality of data on walking and cycling.
- Cities including Portland have also adopted transportation planning tools that are specifically designed to assess latent demand for active transportation and to evaluate the need for improvements to walking and cycling environments.
- A number of public health departments are now working closely with the transportation, planning and other municipal departments to achieve integrated, coordinated action on active transportation.
- Involving community stakeholders in planning and decision-making also has the potential to generate widespread support for progressive actions.
- Many provincial, state and federal governments have developed programs to fund improvements in walking, cycling and public health, as the economic benefits of active transportation are shared across levels of government.
- Reviewing active transportation policies, plans, and actions in other cities may help Toronto to enhance its competitive advantage.



Conclusions

This report concludes that increased City of Toronto investment in the safety, attractiveness and feasibility of walking and cycling can be expected to improve public health in Toronto. The major research findings are as follows:

There is clear evidence that physical activity from active transportation generates very important health benefits. Physical activity from active transportation reduces the risk of cardiovascular disease, diabetes, cancer, and numerous other diseases, leading to significant reductions in the risk of all-cause mortality. Increasing walking and cycling mode shares and safety can also improve quality of life, reduce travel times, and improve the sustainability of Toronto's struggling transportation system.

Walking and cycling prevent an estimated 120 premature deaths each year, yielding annual benefits of over \$100 million. Large increases in mode share – and in the associated health benefits – are very feasible given Toronto trip distances. Toronto lags behind other leading North American cities in active transportation mode share, despite the fact that 55% of all trips are conducive to active transportation. By achieving Vancouver and Portland's walking and cycling commuting mode shares (12% and 6%, respectively), Toronto could prevent an estimated 100 additional deaths each year.

The health benefits of active transportation are not evenly distributed across Toronto. In the former City of Toronto, destinations are closer together, street blocks are shorter, and levels of walking and cycling are over three times higher than in the suburbs, where many of Toronto's high-rise and low-income neighbourhoods are located.

Walking and cycling in Toronto is more dangerous than in other Canadian cities. Though Toronto has made progress in reducing collision rates, pedestrians and cyclists are still disproportionately at risk of injury and fatality due to collisions; pedestrians are particularly vulnerable. We estimate that Toronto could save dozens of lives and avoid over \$60 million in costs by improving the active transportation safety.

Many interventions have the proven potential to increase walking and cycling safety and mode shares, and yield associated public health benefits. Actions related to traffic calming, intersection improvements, and separated walking and cycling facilities have the demonstrated ability to reduce collision and injury rates. They also contribute to mode share increases. Some of these actions can be implemented at minimal cost; for example, intersection markings for pedestrians and cyclists require only surface coatings. Marketing and education programs are also important, cost-effective enablers of safe active transportation, and are particularly successful in combination with action in the built environment. Targeted actions to address the needs of vulnerable users can improve transportation and health inequalities.

Effective goals, standards, data, evaluation tools, and collaboration can facilitate action to improve walking and cycling. Many cities have adopted ambitious safety and mode share goals, and have developed supporting policies and standards that require consideration of the needs of walkers and cyclists in decision-making. Many cities have also enhanced their capacity to promote active transportation by improving the quality of walking and cycling data and evaluation tools. Many cities have received funding from provincial and federal governments, who share in the economic benefits of improved public health. Finally, many cities have enhanced collaboration across municipal departments, and have generated widespread support for progressive actions by involving community stakeholders in planning.

Going forward, Toronto planners and policymakers should consider the health impacts of transportation decisions. This report provides a starting point for valuing the health benefits of walking and cycling, and for investing in interventions that cost-effectively improve public health. Toronto planners can also learn from other North American cities that have successfully improved health and quality of life by enabling safe walking and cycling for transportation.

References

- Active Living Research. Do All Children Have Places to Be Active? Disparities in Access to Physical Activity Environments in Racial and Ethnic Minority and Lower-Income Communities [Internet]. 2011 Nov [cited 2011 Nov 16];Available from: http://www.activelivingresearch.org/resourcesearch/summaries
- Aertsens J, de Geus B, Vandenbulcke G, Degraeuwe B, Broekx S, De Nocker L, et al. Commuting by bike in Belgium, the costs of minor accidents. Accident; analysis and prevention. 2010;42(6):2149-57. Epub 2010/08/24.
- Agence de la sante et des services sociaux de Montreal (ASSSM). Urban transportation: A question of health. Direction de sante publique, Ville de Montreal; 2006.
- Anderson RWG, McLean AJ, Farmer MJB, Lee, BH, Brooks, CG. Vehicle travel speeds and the incidence of fatal pedestrian crashes. Accident analysis and prevention. 1997;29(5):667–674.Ashcroft P, Davis A, Ginger M, Heat H, Miller I, Roper A, et al. Soft measures hard facts: The value for money of transport measures which change travel behaviour: A review of the evidence [Internet]. 2011. Available from: http://www.swpho.nhs.uk/resource/item.aspx?RID=81967
- Bassett DR, Pucher J, Buehler R, Thompson DL, Crouter SE. Walking, Cycling, and Obesity Rates in Europe, North America, and Australia. J Phys Act Health. 2008;5(6):795-814.
- Bellair PE. Social Interaction and Community Crime: Examining the Importance of Neighbor Networks. Criminology. 1997;35:677.
- Berglund B, Lindvall T, Schwela D, editors. Guidelines for Community noise [Internet]. World Health Organization; 1999. Available from: http://www.who.int/docstore/peh/noise/guidelines2.html
- Birk M, Geller R. Bridging the Gaps: How the Quality and Quantity of a Connected Bikeway Network Correlates with Increasing Bicycle Use [Internet]. Transportation Research Board 2006 Annual Meeting; 2005. Available from: www.onegreencity.com/images/crucial/builditandtheywillcome.pdf
- Boer R, Zheng Y, Overton A, Ridgeway GK, Cohen DA. Neighborhood Design and Walking Trips in Ten U.S. Metropolitan Areas. American Journal of Preventive Medicine. 2007 Apr;32:298–304.
- Briggs DJ, de Hoogh K, Morris C, Gulliver J. Effects of travel mode on exposures to particulate air pollution. Environment international. 2008;34(1):12-22. Epub 2007/08/11.
- Buchan K. Investing in transport: making the change [Internet]. Green Alliance; 2009. Available from: http://www.mtru.com/mtru%20publications/NATA_making%20the%20change.pdf
- Cairns S, Atkins S, Goodwin P. Disappearing traffic? The story so far. Proceedings of the ICE Municipal Engineer. 2002;151:13–22.
- Calgary transportation staff. Promoting active transportation and health in Calgary. 2012.
- Canadian Institute for Health Information (CIHI). Improving the health of Canadians: An introduction to health in urban places [Internet]. Ottawa, Canada; 2006. Available from: http://secure.cihi.ca/cihiweb/products/PH_Full_Report_English.pdf
- Canadian Society for Exercise Physiology (CSEP). Canadian Physical Activity Guidelines [Internet]. [cited 2011 Nov 4];Available from: http://www.csep.ca/english/view.asp?x=804
- Cavill N, Kahlmeier S, Rutter H, Racioppi F, Oja P. Economic analyses of transport infrastructure and policies including health effects related to cycling and walking: A systematic review (vol 15, pg 291, 2008). Transport Policy. 2009;16(1):46-.
- Cavill N, Kahlmeier S, Rutter H, Racioppi F, Oja P. Methodological guidance on the economic appraisal of health effects related to walking and cycling. Copenhagen, Denmark: World Health Organization Regional Office for Europe; 2007.
- City of Calgary. Calgary downtown commuter cyclist survey report [Internet]. 2007. Available from: http://www.calgary.ca/Transportation/TP/Documents/transportation_solutions/downtown_cyclist_survey_2007_p1of4. pdf
- City of Calgary. Calgary Transportation Plan [Internet]. 2009. Available from: http://www.calgary.ca/Transportation/TP/Pages/transportation-planning.aspx
- City of Calgary. Cycling Strategy [Internet]. 2011. Available from: http://www.calgary.ca/Transportation/TP/Documents/cycling/Cycling-Strategy/2011-cycling-strategy.pdf
- City of Calgary. The City of Calgary Traffic Calming Policy [Internet]. 2006. Available from: http://www.calgary.ca/CA/cityclerks/Documents/Council-policy-library/tp002.pdf
- City of Chicago. Bike 2015 Plan [Internet]. Available from: http://www.bike2015plan.org/pdf/bike2015plan.pdf
- City of New York. Active Design Guidelines: Promoting physical activity and health in design. 2010.

- City of Toronto Economic Development, Culture and Tourism. Green Economic Sector Development Strategy [Internet]. 2007. Available from: http://www.toronto.ca/planning/official_plan/introduction.htm
- City of Toronto Planning. 2006 City of Toronto Cordon Count Program Information Bulletin [Internet]. 2007. Available from: http://www.toronto.ca/planning/pdf/cordon_count_2006.pdf
- City of Toronto Traffic Safety Unit. 2009 Cyclist Collision Summary Leaflet [Internet]. Available from: http://www.toronto.ca/transportation/publications/brochures/2009_bike.pdf
- City of Toronto Traffic Safety Unit. 2009 Pedestrian Collision Summary Leaflet [Internet]. Available from: http://www.toronto.ca/transportation/publications/brochures/2009_ped.pdf
- City of Toronto Traffic Safety Unit. 2010 Cyclist Collision Summary Leaflet [Internet]. 2011b; Available from: http://www.toronto.ca/transportation/publications/brochures/2010_bike.pdf
- City of Toronto Traffic Safety Unit. 2010 Pedestrian Collision Summary Leaflet [Internet]. 2011a ;Available from: http://www.toronto.ca/transportation/publications/brochures/2010_ped.pdf
- City of Toronto Traffic Safety Unit. 2011 Personal Injury and Fatal Collision Summary Leaflet [Internet]. 2012; Available from: www.toronto.ca/transportation/publications/brochures/2011_fatal.pdf
- City of Toronto Traffic Safety Unit. Pedestrian Collision Study [Internet]. 2007. Available from: http://www.toronto.ca/transportation/walking/ped_safety.htm
- City of Toronto Transportation Services. 2010 Bicycle count report [Internet]. 2010 [cited 2011 Oct 3]. Available from: http://www.toronto.ca/cycling/data/index.htm
- City of Toronto Transportation Services. Appendix 1: Road Classification Criteria [Internet]. 2000. Available from: http://www.toronto.ca/transportation/road_class/update.htm
- City of Toronto Transportation Services. Bikeway Network Staff Report 2011 Update [Internet]. 2011b. Available from: www.toronto.ca/legdocs/mmis/2011/pw/.../backgroundfile-38906.pdf
- City of Toronto Transportation Services. Cycling in Toronto Cycling statistics [Internet]. 2011a [cited 2011 Nov 3];Available from: http://www.toronto.ca/cycling/reports/statistics/statistics-tables.htm#mode
- City of Toronto Transportation Services. Pedestrian and cycling capital budget. 2011.
- City of Toronto Transportation Services. Toronto Bicycle / Motor-Vehicle Collision Study [Internet]. 2003 [cited 2011 Oct 2];Available from: http://www.toronto.ca/transportation/publications/bicycle_motor-vehicle/index.htm
- City of Toronto Transportation staff. Bikeway network status. [Email] February 2012.
- City of Toronto Transportation Services. Toronto Cycling map [Internet]. 2010 [cited 2012 Feb 23]. Available from: http://www.toronto.ca/cycling/map/index.htm
- City of Toronto. Bikeway Network project status [Internet]. 2010 [cited 2012 Jan 31]; Available from: http://www.toronto.ca/cycling/network/network-project-status.htm
- City of Toronto. City of Toronto Bike Plan Shifting Gears [Internet]. 2001 [cited 2011 Oct 3]. Available from: http://www.toronto.ca/cycling/bikeplan/
- City of Toronto. Climate change, clean air and sustainable energy action plan. 2007. Available from: www.toronto.ca/changeisintheair/pdf/clean_air_action_plan.pdf
- City of Toronto. Lawrence-Allen Cycling and Walking Study. 2009a.
- City of Toronto. PATH Toronto's downtown walkway [Internet]. 2012 [cited 2012 Feb 22]. Available from: http://www.toronto.ca/path/
- City of Toronto. Pedometer E-Newsletter [Internet]. 2011 [cited 2012 Feb 22]. Available from: http://www.toronto.ca/transportation/walking/newsletter/
- City of Toronto. Toronto Official Plan [Internet]. 2010. Available from: http://www.toronto.ca/planning/official_plan/introduction.htm
- City of Toronto. Toronto Walking Strategy. 2009b; Available from: http://www.toronto.ca/transportation/walking/walking_strategy.htm
- City of Vancouver Bicycle Program Coordinator. Cycling facilities in Vancouver [Email Feb. 2012]
- City of Vancouver. Cycling in Vancouver Looking Forward to 2010/2011: Administrative Report to Standing Committee on Transportation and Traffic [Internet]. 2010. Available from: vancouver.ca/ctyclerk/cclerk/20100506/documents/csbu5.pdf
- City of Vancouver. Statistics Separated bike lanes [Internet]. 2011 [cited 2012 Feb 8];Available from: http://vancouver.ca/engsvcs/transport/cycling/separated/dunsmuir_results.htm
- Cooper C. Successfully Changing Individual Travel Behavior: Applying Community-Based Social Marketing to Travel Choice. Transportation Research Record: Journal of the Transportation Research Board. 2007;2021:89–99.

Cortright J. Portland's Green Dividend. CEOs for Cities; 2007 [cited 2011 Oct 2]. Available from: http://www.ceosforcities.org/work/portlands_green_dividend

- Cortright J. Walking the Walk: How Walkability Increases Home Values in US cities. [Internet]. CEOs for Cities; 2009 [cited 2011 Oct 2]. Available from: http://www.ceosforcities.org/work/walkingthewalk
- Creatore MI, Gozdyra P, Booth GL, Ross K, Glazier RH. Socioeconomic status and diabetes. In: Glazier RH, Booth GL, Gozdyra P, Creatore MI, Tynan, M, editors. Neighbourhood Environments and Resources for Healthy Living — A Focus on Diabetes in Toronto: ICESAtlas. Toronto:Institute for Clinical Evaluative Sciences; 2007.
- Data Management Group, University of Toronto Civil Engineering (DMG). Transportation Tomorrow Survey Internet Data Retrieval System [Internet]. 2008 [cited 2011 Nov 3];Available from: https://www.jpint.utoronto.ca/cgi-bin/xtab-query
- Data Management Group, University of Toronto Civil Engineering. 2006 Transportation Tomorrow Survey. City of Toronto: 2006, 2001 and 1996 travel survey summaries [Internet]. 2009 [cited 2011 Oct 3]. Available from: http://www.dmg.utoronto.ca/reports/ttsreports.html
- Davis A, Jones M. Physical activity, absenteeism and productivity: an Evidence Review. Travel Demand Management team Transport for London; 2007.
- Day K. Active Living and Social Justice: Planning for Physical Activity in Low-income, Black, and Latino Communities. Journal of the American Planning Association. 2006 Mar 31;72:88–99.
- de Hartog JJ, Boogaard H, Nijland H, Hoek G. Do the Health Benefits of Cycling Outweigh the Risks? Environ Health Perspect. 2010;118(8):1109-16.
- de Nazelle A, Rodriguez DA, Crawford-Brown D. The built environment and health: impacts of pedestrian-friendly designs on air pollution exposure. The Science of the total environment. 2009;407(8):2525-35. Epub 2009/02/10.
- Dekoster J, Schollaert U. Cycling: the way ahead for towns and cities [Internet]. Belgium: European Communities; 1999. Available from: http://ec.europa.eu/environment/archives/cycling/cycling_en.pdf
- Dill J, Gliebe J. Understanding and measuring bicycling behavior: A focus on travel time and route choice. Oregon Transportation Research and Education Consortium; 2008.
- Dill J, Monsere CM, McNeil N. Evaluation of bike boxes at signalized intersections. Accident Analysis & Prevention. 2012 Jan;44(1):126–34.
- Dill J. Bicycling for Transportation and Health: The Role of Infrastructure. Journal of Public Health Policy. 2009;30:S95–110.
- District Department of Transportation (DDOT). Bicycle Infrastructure in DC [Internet]. 2011;Available from: http://ddot.dc.gov/DC/DDOT/On+Your+Street/Bicycles+and+Pedestrians/Bicycles
- District of Columbia. Great Streets [Internet]. Office of the Deputy Mayor for Planning and Economic Development. 2012. Available from: http://dmped.dc.gov/DC/DMPED/Programs+and+Initiatives/Great+Streets
- Dunn AL, Andersen RE, Jakicic JM. Lifestyle physical activity interventions: History, short- and long-term effects, and recommendations. American Journal of Preventive Medicine. 1998; 15(4):398–412.
- Dunn AL, Marcus BH, Kampert JB, Garcia ME, Kohl HW, Blair SN. Comparison of Lifestyle and Structured Interventions to Increase Physical Activity and Cardiorespiratory Fitness. JAMA: The Journal of the American Medical Association. 1999 Jan 27;281(4):327–34.
- Edwards P, Tsouros AD. Promoting physical activity and active living in urban environments. The role of local governments. The solid facts. [Internet]. World Health Organization; 2006 [cited 2011 Oct 3]. Available from: http://www.euro.who.int/en/what-we-publish/abstracts/promoting-physical-activity-and-active-living-in-urbanenvironments.-the-role-of-local-governments.-the-solid-facts.
- Edwards RD. Public transit, obesity, and medical costs: Assessing the magnitudes. Prev Med. 2008;46(1):14-21.
- Elvik R. The non-linearity of risk and the promotion of environmentally sustainable transport. Accident Analysis and Prevention. 2009;41(4):849-55.
- Ernst M. Dangerous by design: Solving the epidemic of preventable pedestrian deaths [Internet]. Transportation for America; 2011 [cited 2011 Oct 2]. Available from: http://t4america.org/resources/dangerousbydesign2011/
- Ersoz C. Barriers for the Low Income Cyclist. Bicycle Paper.com [Internet]. 2011; Available from: http://www.bicyclepaper.com/articles/273-Barriers-for-the-Low-Income-Cyclist-
- Ewing R, Handy S. Measuring the Unmeasurable: Urban Design Qualities Related to Walkability. Journal of Urban Design. 2009 Feb;14:65–84.
- Federation of Canadian Municipalities. Mending Canada's Social Safety Net: The role of municipal governments [Internet]. 2010. Available from: http://www.fcm.ca/home/resources/reports.htm
- Fenton M, Junot B. Health Economic Analysis Tool (H.E.A.T.) for Walking and Bicycling in the United States: Summary of prospective U.S. users' interest and recommendations [Internet]. 2010. Available from: http://www.activelivingresearch.org/files/HEATReport Final.pdf

- Forkes J, Smith Lea N. Bike lanes, on-street parking and business Year 2 report: A study of Bloor Street in Toronto's Bloor West Village [Internet]. Clean Air Partnership; 2010. Available from: http://www.cleanairpartnership.org/files/BikeLanes Parking Business BloorWestVillage.pdf
- Forsyth A, Hearst M, Oakes JM, Schmitz KH. Design and Destinations: Factors Influencing Walking and Total Physical Activity. Urban Studies. 2008;45(9):1973–96.
- Fox KR. The Influence of Physical Activity on Mental Well-Being. Public Health Nutrition. 1999;2(Supplement 3a):411-8.
- Frank LD, Andresen MA, Schmid TL. Obesity relationships with community design, physical activity, and time spent in cars. American Journal of Preventive Medicine. 2004 Aug;27(2):87–96.
- Gagnon F, Bellefleur O. Traffic calming: Political dimensions [Internet]. National Collaborating Centre for Healthy Public Policy; 2011. Available from: www.ncchpp.ca/docs/PolDimensions_TrafficCalming_En.pdf
- Garrett-Peltier H. Pedestrian and Bicycle Infrastructure: A National Study of Employment Impacts [Internet]. Political Economy Research Institute; 2011 [cited 2011 Oct 2]. Available from: http://www.peri.umass.edu/193/
- Gauvin L, Riva M, Barnett T, Richard L, Craig CL, Spivock M, et al. Association between Neighborhood Active Living Potential and Walking. American Journal of Epidemiology. 2008 Apr 15;167(8):944–53.
- Genter JA, Donovan S, Petrenas B, Badland H. Valuing the benefits of active transport modes. 2008. Available from: www.nzta.govt.nz/resources/research/reports/359/docs/359.pdf
- Get Active Toronto. 2011 Get Active Toronto Report on Physical Activity [Internet]. 2011. Available from: www.tcf.ca/vitalinitiatives/Get_Active_Toronto_Report_2011.pdf
- Glazier RH, Ross K, Gozdyra P, Creatore MI, Booth GL. Neighbourhood Infrastructure. In: Glazier RH, Booth GL, Gozdyra P, Creatore MI, Tynan, M, editors. Neighbourhood Environments and Resources for Healthy Living — A Focus on Diabetes in Toronto: ICESAtlas. Toronto:Institute for Clinical Evaluative Sciences; 2007.
- Gordon-Larsen P, Boone-Heinonen J, Sidney S, Sternfeld B, Jacobs DR, Jr., Lewis CE. Active commuting and cardiovascular disease risk: the CARDIA study. Archives of internal medicine. 2009;169(13):1216-23. Epub 2009/07/15.
- Government of Ontario. Growth Plan for the Greater Golden Horseshoe. 2006. Available from: https://www.placestogrow.ca/index.php?option=com_content&task=view&id=9&Itemid=14
- Grabow ML, Spak SN, Holloway T, Stone Jr. B, Mednick AC, Patz JA. Air Quality and Exercise-Related Health Benefits from Reduced Car Travel in the Midwestern United States. Environmental Health Perspectives [Internet]. 2011 Nov 2 [cited 2011 Nov 3];Available from:

http://ehp03.niehs.nih.gov/article/fetchArticle.action?articleURI=info%3Adoi%2F10.1289%2Fehp.1103440

- Grous A. The British cycling economy: "gross cycling product" report. London School of Economics, Sky; 2011 [cited 2011 Oct 2]. Available from: http://corporate.sky.com/documents/pdf/press_releases/2011/the_british_cycling_economy
- Grundy C, Steinbach R, Edwards P, Green J, Armstrong B, Wilkinson P. Effect of 20 mph traffic speed zones on road injuries in London, 1986-2006: controlled interrupted time series analysis. BMJ. 2009 Dec 10;339(dec10 3):b4469–b4469.
- Halcrow Consulting Inc. The Future of the Transportation Tomorrow Survey. 2008.
- Hamer M, Chida Y. Active commuting and cardiovascular risk: a meta-analytic review. Prev Med. 46(1):9–13.
- Hamer M, Chida Y. Walking and primary prevention: a meta-analysis of prospective cohort studies. Brit J Sport Med. 2008;42(4):238-43.
- Harris / Decima. City of Toronto: Walking Habits and Attitudes Report [Internet]. 2008 [cited 2011 Oct 3]. Available from: http://www.toronto.ca/transportation/walking/index.htm
- Heesch KC, Burton NW, Brown WJ. Concurrent and prospective associations between physical activity, walking and mental health in older women. Journal of Epidemiology and Community Health [Internet]. 2010 Jun 1 [cited 2011 Nov 7];Available from: http://jech.bmj.com/content/early/2010/05/31/jech.2009.103077.abstract
- Hendriksen IJM, Simons M, Garre FG, Hildebrandt VH. The association between commuter cycling and sickness absence. Preventive Medicine. 2010 Aug;51(2):132–5.
- Herndon LK. Ruth Finkelstein. Metropolis. 2012 Jan;31(6):62.
- Hertel O, Hvidberg M, Ketzel M, Storm L, Stausgaard L. A proper choice of route significantly reduces air pollution exposure A study on bicycle and bus trips in urban streets. Science of the Total Environment. 2008;389(1):58-70.
- Hess P, Farrow J. Walkability in Toronto's high-rise neighbourhoods [Internet]. 2010. Available from: http://janeswalk.net/assets/uploads_docs/Walkability_Full_Report.pdf
- Hess P, Moudon A, Snyder M, Stanilov K. Site Design and Pedestrian Travel. Transportation Research Record: Journal of the Transportation Research Board. 1999 Jan 1;1674(-1):9–19.
- Hess PM, Milroy B. Making Toronto's streets [Internet]. Centre for Urban Health Initiatives, University of Toronto; 2006. Available from:

http://www.utoronto.ca/cuhi/research/supportingdocs/Toronto%20Streets%20Report%20Sept%202006.pdf

Hillsdon M, Thorogood M. A systematic review of physical activity promotion strategies. British Journal of Sports Medicine. 1996 Jun 1;30(2):84–9.

- Hoek G, Brunekreef B, Goldbohm S, Fischer P, van den Brandt PA. Association between mortality and indicators of trafficrelated air pollution in the Netherlands: a cohort study. The Lancet. 2002;360(9341):1203–9.
- Hou L, Ji B-T, Blair A, Dai Q, Gao Y-T, Chow W-H. Commuting Physical Activity and Risk of Colon Cancer in Shanghai, China. American Journal of Epidemiology. 2004 Nov 1;160(9):860–7.
- Hu G, Qiao Q, Silventoinen K, Eriksson JG, Jousilahti P, Lindström J, et al. Occupational, commuting, and leisure-time physical activity in relation to risk for Type 2 diabetes in middle-aged Finnish men and women. Diabetologia. 2003;46(3):322–9.
- Hulchanski D. The 3 cities within Toronto: Income polarization among Toronto's neighbourhoods, 1970-2005 [Internet]. Cities Centre, University of Toronto; 2007. Available from: http://www.urbancenter.utoronto.ca/hulchanski.html
- Hunter W. Evaluation of Innovative Bike-Box Application in Eugene, Oregon. Transportation Research Record: Journal of the Transportation Research Board. 2000;1705:99–106.
- IBPI | Bike Boulevard Guidebook [Internet]. [cited 2011 Nov 18];Available from: http://www.ibpi.usp.pdx.edu/guidebook.php
- Ipsos Reid. 2009 City of Toronto Cycling Study [Internet]. 2010. Available from: http://www.toronto.ca/cycling/reports/index.htm
- Jacobsen PL. Safety in numbers: more walkers and bicyclists, safer walking and bicycling. Injury Prev. 2003;9(3):205-9.
- Jensen SU, editor. Collection of cycle concepts. Copenhagen, Denmark: Danish Road Directorate; 2000.
- Jensen SU. Safety effects of blue cycle crossings: a before-after study. Accident analysis and prevention. 2008;40(2):742-50.
- Kalinowski T. Metrolinx confirms downtown relief line is still on the map [Internet]. Toronto Star. 2011b [cited 2012 Feb 2];Available from: http://www.thestar.com/mobile/NEWS/article/1091627
- Kalinowski T. The day Toronto stops: True gridlock is around the corner [Internet]. The Toronto Star. 2011a Nov 5 [cited 2012 Feb 5];Available from: http://www.thestar.com/news/article/1081673--the-day-toronto-stops-true-gridlock-is-aroundthe-corner
- Katzmarzyk PT, Gledhill N, Shephard RJ. The economic burden of physical inactivity in Canada. CMAJ. 2000 Nov 28;163(11):1435–40.
- Katzmarzyk PT, Janssen I. The economic costs associated with physical inactivity and obesity in Canada: an update. Can J Appl Physiol. 2004 Feb;29(1):90–115.
- Kaur, Nieuwenhuijsen MJ, Colvile RN. Fine particulate matter and carbon monoxide exposure concentrations in urban street transport microenvironments. Atmospheric Environment. 2007;41(23):4781–810.
- Kesaniemi YK, Danforth E Jr, Jensen MD, Kopelman PG, Lefèbvre P, Reeder BA. Dose-response issues concerning physical activity and health: an evidence-based symposium. Med Sci Sports Exerc. 2001 Jun;33(6 Suppl):S351–8.
- Knoblauch RL. INVESTIGATION OF EXPOSURE BASED PEDESTRIAN ACCIDENT AREAS: CROSSWALKS, SIDEWALKS, LOCAL STREETS AND MAJOR ARTERIALS. FINAL REPORT - Transport Research International Documentation - TRID [Internet]. [cited 2011 Nov 18];Available from: http://trid.trb.org/view.aspx?id=287673
- Korve MJ, Niemeier DA. Benefit-Cost Analysis of Added Bicycle Phase at Existing Signalized Intersection. Journal of Transportation Engineering. 2002 Jan 1;128(1):40.
- Landis B, Vattikuti V, Brannick M. Real-Time Human Perceptions: Toward a Bicycle Level of Service. Transportation Research Record: Journal of the Transportation Research Board. 1997 Jan 1;1578(-1):119–26.
- Larsen K, Gilliland J, Hess P, Tucker P, Irwin J, He M. The Influence of the Physical Environment and Sociodemographic Characteristics on Children's Mode of Travel to and From School. Am J Public Health. 2009 Mar 1;99(3):520–6.
- Lavizzo-Mourey R, McGinnis JM. Making the Case for Active Living Communities. Am J Public Health. 2003 Sep;93(9):1386-8.
- Lawrie J, Guenther J, Cook T, Meletiou MP, O'Brien SW. The economic impact of investments in bicycle facilities: A case study of the Northern Outer Banks [Internet]. North Carolina Department of Transportation; 2004. Available from: http://www.ncdot.gov/bikeped/researchreports/
- Leyden KM. Social Capital and the Built Environment: The Importance of Walkable Neighborhoods. Am J Public Health. 2003 Sep 1;93(9):1546–51.
- Lockett D, Willis A, Edwards N. Through Seniors' Eyes: An Exploratory Qualitative Study to Identify Environmental Barriers to and Facilitators of Walking. CJNR (Canadian Journal of Nursing Research). 2005;37(3):48–65.
- Lund H. Pedestrian Environments and Sense of Community. Journal of Planning Education and Research. 2002 Mar 1;21(3):301–12.
- Lynott J, Taylor A, Twaddell H, Haase J. Planning Complete Streets for an Aging America [Internet]. Washington D.C.: AARP Public Policy Institute; 2009. Available from: http://www.aarp.org/home-garden/livable-communities/info-08-2009/Planning_Complete_Streets_for_an_Aging_America.html

- Malekafzali S, editor. Healthy, equitable, transportation policy: Recommendations and research. PolicyLink, Prevention Institute, Convergence Partnership; 2010. Available from: http://www.convergencepartnership.org/site/c.fhLOK6PELmF/b.5327643/k.BF0B/Transportation_RX.htm
- Martens K. Basing Transport Planning on Principles of Social Justice. Berkeley Planning Journal. 2006;19:1–17.
- McAuley E, Blissmer B, Katula J, Duncan TE, Mihalko SL. Physical activity, self-esteem, and self-efficacy relationships in older adults: A randomized controlled trial. Annals of Behavioral Medicine. 2000 Jun;22:131–9.
- Merrild U, Bak S. An excess of pedestrian injuries in icy conditions: A high-risk fracture group—elderly women. Accident Analysis & Prevention. 1983 Feb;15(1):41–8.
- Metrolinx. The Big Move: Transforming transportation in the Greater Toronto and Hamilton Area [Internet]. Government of Ontario; 2008 [cited 2011 Oct 3]. Available from:

http://www.metrolinx.com/en/regionalplanning/bigmove/big_move.aspx

Ministry of Municipal Affairs and Housing (MMAH), Ontario Professional Planners Institute (OPPI). Planning by design: a healthy communities handbook [Internet]. Queen's Printer for Ontario; 2009. Available from: http://www.mah.gov.on.ca/Page6737.aspx

Ministry of Transportation of Ontario (MTO). Draft - Transit-supportive Guidelines. 2011.

- Minneapolis transportation staff. Promoting active transportation and health in Minneapolis. 2011.
- Mohan D. Urban transport and climate change: issues and concerns in the Indian context. In: 3i-Network, editor. India infrastructure report 2010: infrastructure development in a low carbon economy. Oxford University Press;
- Moudon AV, Lee C. Walking and bicycling: an evaluation of environmental audit instruments. Am J Health Promot. 2003 Oct;18(1):21–37.
- Murphy MH, Nevill AM, Murtagh EM, Holder RL. The effect of walking on fitness, fatness and resting blood pressure: a metaanalysis of randomised, controlled trials. Prev Med. 2007 May;44(5):377–85.
- National Association of City Transportation Officials. Urban Bikeway Design Guide [Internet]. 2011 [cited 2011 Oct 3]; Available from: http://nacto.org/cities-for-cycling/design-guide/
- New York City Department of Transportation (DOT). Pedestrian Safety Study and Action Plan [Internet]. 2012 [cited 2012 Feb 7];Available from: http://www.nyc.gov/html/dot/html/about/pedsafetyreport.shtml
- New York City Department of Transportation (DOT). Sustainable Streets [Internet]. 2008. Available from: http://www.nyc.gov/html/dot/downloads/pdf/stratplan_compplan.pdf
- New York City Department of Transportation (NYC DOT). Evaluation of Solid Green Bicycle Lanes, to Increase Compliance and Bicycle Safety. 2011.
- New York City Department of Transportation. Greenlight for Midtown Evaluation Report [Internet]. 2010. Available from: http://www.nyc.gov/html/dot/html/about/broadway.shtml
- New York City Global Partners. Best Practice: City-wide Bicycle Commuting Program [Internet]. 2011. Available from: http://www.nyc.gov/html/unccp/gprb/downloads/pdf/Copenhagen_CityofCyclists.pdf
- NYC Public Health staff. Promoting active transportation and health in New York City. 2011.
- OECD. OECD Territorial Reviews: Toronto, Canada [Internet]. 2010. Available from: www.oecd.org/gov/regional/toronto
- Oja P, Vuori I, Paronen O. Daily walking and cycling to work: their utility as health-enhancing physical activity. Patient Education and Counseling. 1998;33, Supplement 1(0):S87-S94.
- Ontario Medical Association (OMA). The Illness Costs of Air Pollution: 2005-2026 health and economic damage estimates [Internet]. Available from: http://www.oma.org/phealth/smogmain.htm.
- Ontario Ministry of Health and Long-Term Care (OMHLTC). Initial Report on Public Health: Physical Activity Index [Internet]. 2009 [cited 2011 Oct 14]. Available from:

http://www.health.gov.on.ca/english/public/pub/pubhealth/init_report/pai.html

- Ontario Ministry of Public Infrastructure Renewal (OMPIR). Growth Plan for the Greater Golden Horseshoe [Internet]. 2006. Available from: https://www.placestogrow.ca/index.php?option=com_content&task=view&id=9&Itemid=14
- Ontario Professional Planners Institute (OPPI). Healthy Communities and Planning for a Sustainable Greater Toronto Area: A Call to Action from the Ontario Professional Planners Institute [Internet]. 2011. Available from: http://www.ontarioplanners.on.ca/pdf/A_Call_to_Action_from_OPPI.pdf
- Ontario Professional Planners Institute (OPPI). Healthy Communities and Planning for Age-Friendly Communities: A Call to Action [Internet]. 2009. Available from: http://www.ontarioplanners.on.ca/pdf/Call%20to%20Action%20Age-Friendly%20Communities%20June%2018,%202009.pdf
- Ontario Professional Planners Institute (OPPI). Healthy Communities and Planning for the Needs of Children and Youth: A Call to Action [Internet]. 2009. Available from:

http://www.ontarioplanners.on.ca/pdf/Plan%20for%20the%20Needs%20of%20Children%20and%20Youth%20OPPI%20C all%20to%20Action%20Final%20Version.pdf

- Painter K. The influence of street lighting improvements on crime, fear and pedestrian street use, after dark. Landscape and Urban Planning. 1996 Aug;35(2-3):193–201.
- Paluska S.A., Schwenk T.L. Physical Activity and Mental Health: Current Concepts. Sports Medicine. 2000;29(3):167-80.
- Pooley C, Horton D, Scheldeman G, Harrison R. Shaping the City for Walking and Cycling: A Case Study of Lancaster. Built Environment. 2010;36(4):447–60.
- Pope CA, Burnett RT, Thun MJ, Calle EE, Krewski D, Ito K, et al. Lung Cancer, Cardiopulmonary Mortality, and Long-term Exposure to Fine Particulate Air Pollution. JAMA: The Journal of the American Medical Association. 2002;287(9):1132–41.
- Pratt M, Macera CA, Wang G. Higher direct medical costs associated with physical inactivity. Phys Sportsmed. 2000 Oct;28(10):63–70.
- Preston B. Cost effective ways to make walking safer for children and adolescents. Inj. Prev. 1995 Sep;1(3):187-90.
- Public Health Agency of Canada (PHAC). Physical Activity in Canada 2007 [cited 2011 Oct 2]; Available from: http://www.phac-aspc.gc.ca/alw-vat/intro/canada-eng.ph
- Pucher J, Buehler R. Analysis of bicycling trends and policies in large North American Cities: Lessons for New York [Internet]. U.S. Department of Transportation; 2011. Available from: www.utrc2.org/research/assets/176/Analysis-Bike-Final1.pdf
- Pucher J, Dijkstra L. Making walking and cycling safer: Lessons from Europe. Transportation Quarterly. 2000;54(3):25–50.
- Pucher J, Dijkstra L. Promoting Safe Walking and Cycling to Improve Public Health: Lessons From The Netherlands and Germany. Am J Public Health. 2003 Sep 1;93(9):1509–16.
- Pucher J, Dill J, Handy S. Infrastructure, programs, and policies to increase bicycling: An international review. Preventive Medicine. 2010 Jan;50, Supplement(0):S106–25.
- Reid D. The most dangerous intersections for pedestrians in Toronto [Internet]. Spacing Toronto. 2011 Jun 8 [cited 2011 Oct 2]; Available from: http://spacingtoronto.ca/2011/06/08/the-most-dangerous-intersections-for-pedestrians-in-toronto/
- Retting R, Van Houten R. Safety benefits of advance stop lines at signalized intersections: Results of a field evaluation. Institute of Transportation Engineers. ITE Journal. 2000;70(9):47.
- Retting RA, Ferguson SA, McCartt AT. A review of evidence-based traffic engineering measures designed to reduce pedestrianmotor vehicle crashes. Am J Public Health. 2003;93(9):1456-63. Epub 2003/09/02.
- Retting RA, Ferguson SA, McCartt AT. A Review of Evidence-Based Traffic Engineering Measures Designed to Reduce Pedestrian–Motor Vehicle Crashes. Am J Public Health. 2003;93(9):1456–63.
- Reynolds C, Harris MA, Teschke K, Winters M. The impact of transportation infrastructure on bicycling injuries and crashes: a review of the literature. Environmental Health. 2009;8:47.
- Rietveld P, Daniel V. Determinants of bicycle use: do municipal policies matter? Transportation Research Part A: Policy and Practice. 2004 Aug;38(7):531–50.
- Robinson DL. Safety in numbers in Australia: more walkers and bicyclists, safer walking and bicycling. Health promotion journal of Australia : official journal of Australian Association of Health Promotion Professionals. 2005;16(1):47-51.
- Rojas-Rueda D, de Nazelle A, Tainio M, Nieuwenhuijsen MJ. The health risks and benefits of cycling in urban environments compared with car use: health impact assessment study. BMJ. 2011 Aug 4;343:d4521–d4521.
- Saelensminde K. Cost-benefit analyses of walking and cycling track networks taking into account insecurity, health effects and external costs of motorized traffic. Transport Res a-Pol. 2004;38(8):593-606.
- Sallis JF, Frank LD, Saelens BE, Kraft MK. Active transportation and physical activity: opportunities for collaboration on transportation and public health research. Transportation Research Part A: Policy and Practice. 2004 May;38(4):249–68.
- Samitz G, Egger M, Zwahlen M. Domains of physical activity and all-cause mortality: systematic review and dose–response meta-analysis of cohort studies. International journal of epidemiology. 2011.
- Sawyer D, Stiebert S, Welburn C. Evaluation of Total Cost of Air Pollution Due to Transportation in Canada [Internet]. Transport Canada; 2007. Available from: http://publications.gc.ca/site/eng/342373/publication.html
- Seattle Department of Transportation. Seattle Pedestrian Master Plan: Development [Internet]. 2011. Available from: http://www.seattle.gov/transportation/pedestrian_masterplan/pmp_dev_people.htm
- Shaw Media / Global News Toronto. Pedestrian traffic at Toronto intersections. June 7, 2011. Available from: http://www.globaltoronto.com/Pedestrian+traffic+Toronto+intersections/4907973/story.html
- Sloman L, Cairns S, Newson C, Anable J, Pridmore A, Goodwin P. The effects of Smarter Choice Programmes in the Sustainable Travel Towns: Summary Report. Department for Transport; 2010.
- Stansfeld SA, Matheson MP. Noise pollution: non-auditory effects on health. British Medical Bulletin. 2003 Dec 1;68(1):243– 57.

Stantec, Vandermark Consulting. Ontario Bike Plan [Internet]. Cycle Ontario Alliance; 2008. Available from: www.tbn.ca/forms/OntarioBikePlan-2008-web.pdf

Statistics Canada. 2006 Census. http://www12.statcan.gc.ca/census-recensement/2006/index-eng.cfm

Statistics Canada. Health Profile - City of Toronto Health Unit [Internet]. 2011 [cited 2012 Feb 6]; Available from: http://www12.statcan.ca/health-sante/82-

228/details/page.cfm?Lang=E&Tab=1&Geo1=HR&Code1=3595&Geo2=PR&Code2=35&Data=Rate&SearchText=City%20of %20Toronto%20Health%20Unit&SearchType=Contains&SearchPR=01&B1=All&Custom=

- Statistics Canada. Mortality, Summary: List of Causes [Internet]. 2008. Available from: http://www.statcan.gc.ca/pub/84f0209x/84f0209x2008000-eng.pdf
- Sustainable Development Commission. Fairness in a car dependent society [Internet]. 2011. Available from: http://www.sd-commission.org.uk/publications.php?id=1184
- Sutherland E, Carlisle R. Healthy by Design: an innovative planning tool for the development of safe, accessible and attractive environments. N S W Public Health Bull. 2007;18(12):228–31.
- Sztabinski F. Bike lanes, on-street parking and business: A Study of Bloor Street in Toronto's Annex Neighbourhood [Internet]. Clean Air Partnership; 2009. Available from: www.cleanairpartnership.org/pdf/bike-lanes-parking.pdf
- Taylor AH. Physical activity, anxiety and stress. In: Biddle S, Fox KR, Boutcher SH, editors. Physical activity and psychological well-being. Routledge; 2000. p. 10–45.
- Telama R, Yang X, Viikari J, Välimäki I, Wanne O, Raitakari O. Physical activity from childhood to adulthood: A 21-year tracking study. American Journal of Preventive Medicine. 2005 Apr;28(3):267–73.
- Thai A, McKendry I, Brauer M. Particulate matter exposure along designated bicycle routes in Vancouver, British Columbia. Sci. Total Environ. 2008 Nov 1;405(1-3):26–35.
- Thogersen J, Garling T, Steg L. Social Marketing of Alternative Transportation Modes. In: Threats to the Quality of Urban Life from Car Traffic: Problems, Causes, and Solutions. Elsevier; 2007. p. Chapter 23.
- Thune I, Furberg AS. Physical activity and cancer risk: dose-response and cancer, all sites and site-specific. Med Sci Sports Exerc. 2001 Jun;33(6 Suppl):S530–50; discussion S609–10.
- Titze S, Stronegger WJ, Janschitz S, Oja P. Association of built-environment, social-environment and personal factors with bicycling as a mode of transportation among Austrian city dwellers. Preventive Medicine. 2008 Sep;47(3):252–9.
- Toronto Board of Trade. Reaching Top Speed Infrastructure: Unleashing Ontario's Ability To Grow [Internet]. 2011b. Available from: http://www.bot.com/Content/NavigationMenu/Policy/VoteOntario2011/ReachingTopSpeed/default.htm
- Toronto Board of Trade. Toronto as a global city: Scorecard on Prosperity 2011a. 2011. Available from: http://www.bot.com/Content/NavigationMenu/Policy/Scorecard/default.htm
- Toronto Coalition for Active Transportation (TCAT). Action 2014 Taking the Next Steps: 2010 Active Transportation Platform [Internet]. 2010. Available from: http://tcat.ca/platform
- Toronto Coalition for Active Transportation (TCAT). Backgrounder: Bikeway options [Internet]. 2011 Sep [cited 2011 Sep 30];Available from: http://www.torontocat.ca/node/1689
- Toronto Community Foundation. Toronto's Vital Sign 2011. 2011. Available from: http://www.tcf.ca/vitalinitiatives/vitalsigns.html
- Toronto Public Health (TPH). Air pollution burden of illness from traffic in Toronto Problems and solutions [Internet]. 2007 [cited 2011 Oct 3]. Available from: http://www.toronto.ca/health/hphe/air_pollution_burden.htm
- Toronto Public Health (TPH). The Unequal City: Health Inequalities in Toronto. 2008 [cited 2011 Oct 3]. Available from: http://www.toronto.ca/health/map/inequalities.htm
- Toronto Public Health (TPH). Toronto's Health Status 2010. 2010 [cited 2011 Oct 3]. Available from: http://www.toronto.ca/health/map/index.htm
- Toronto Public Health (TPH). Toronto's Health Status Indicator Series Physical Activity. 2011. Available from: http://www.toronto.ca/health/map/indicators/pdf/physicalactivity.pdf
- Transport Association of Canada and Canadian Institute Transportation Engineers. Canadian Guide to Neighbourhood Traffic Calming. 1998.
- Transport Canada. Active transportation in Canada: a resource and planning guide. Government of Canada; 2010.
- Transport Canada. Canadian Motor Vehicle Traffic Collision Statistics: 2009 [Internet]. 2011 [cited 2011 Nov 4]. Available from: http://www.tc.gc.ca/eng/roadsafety/tp-tp3322-2009-1173.htm#t1
- Transportation Research Board. Driving and the Built Environment: The Effects of Compact Development on Motorized Travel, Energy Use, and CO2 Emissions [Internet]. 2009. Available from: onlinepubs.trb.org/Onlinepubs/sr/sr298.pdf

- TransportPolitic. Mode share US Cities [Internet]. 2011 [cited 2011 Oct 28]. Available from: http://www.thetransportpolitic.com/2010/10/13/transit-mode-share-trends-looking-steady-rail-appears-to-encouragenon-automobile-commutes/
- UK Department for Transport. Guidance documents 3.5.4. Cost Benefit Analysis [Internet]. 2011 [cited 2012 Feb 21]. Available from: http://www.dft.gov.uk/webtag/documents/expert/unit3.5.5.php
- UK Department for Transport. Interim evaluation of the implementation of 20 mph speed limits in Portsmouth [Internet]. 2010 [cited 2011 Nov 18]; Available from: http://www.dft.gov.uk/publications/speed-limits-portsmouth/
- University of the West of England, Bristol, Cavill Associates. Making the case for investment in the walking environment [Internet]. Living Streets; 2011 [cited 2011 Oct 2]. Available from: http://www.livingstreets.org.uk/index.php?cID=651
- Urban Design 4 Health (UD4H). Toronto Walkability Map. City of Toronto; 2012.
- Van Houten R, Retting R, Farmer C, Houten J. Field Evaluation of a Leading Pedestrian Interval Signal Phase at Three Urban Intersections. Transportation Research Record: Journal of the Transportation Research Board. 2000 Jan 1;1734(-1):86–92.
- VanZerr M. Resident Perceptions of Bicycle Boulevards. IBPI; 2010.
- Ville de Montreal, Direction des transports. Bilan 2010: Mise en oeuvre du Plan de Transport [Internet]. 2010. Available from: http://servicesenligne2.ville.montreal.qc.ca/sel/publications/htdocs/porteaccespublication_Fr/porteaccespublication.jsp ?systemName=94985883
- Ville de Montreal. Sustainable development Quartiers 21 [Internet]. 2012 [cited 2012 Feb 8];Available from: http://ville.montreal.qc.ca/portal/page?_pageid=7137,88561575&_dad=portal&_schema=PORTAL
- Vodden K, Smith D, Eaton F, Mayhew D. Analysis and Estimation of the Social Cost of Motor Vehicle Collisions in Ontario: Final Report [Internet]. Transport Canada; 2007. Available from: www.tc.gc.ca/media/documents/roadsafety/tp14800e.pdf
- Walker L, Tressider M, Birk M. Fundamentals of bicycle boulevard planning & design [Internet]. IBPI Initiative for Bicycle and Pedestrian Innovation; 2009. Available from: www.ibpi.usp.pdx.edu/media/BicycleBoulevardGuidebook.pdf
- Wallasper J. Portland's Not Perfect, But Offers Bright Ideas For Making Biking Mainstream [Internet]. NRDC Smarter Cities. 2010 Dec 3 [cited 2011 Oct 2];Available from: http://smartercities.nrdc.org/articles/portland%E2%80%99s-not-perfectoffers-bright-ideas-making-biking-mainstream
- Wardman M, Tight M, Page M. Factors influencing the propensity to cycle to work. Transportation Research Part A: Policy and Practice. 2007 May;41(4):339–50.
- Washington transportation staff. Promoting active transportation and health in Washington. 2011.
- Wener RE, Evans GW. A Morning Stroll. Environment and Behavior. 2007 Jan 1;39(1):62-74.
- Wikipedia. Traffic calming [Internet]. 2012 [cited 2012 Feb 23]. Available from: http://en.wikipedia.org/wiki/Traffic_calming
- Winters M, Davidson G, Kao D, Teschke K. Motivators and deterrents of bicycling: comparing influences on decisions to ride. Transportation. 2011;38:153–68.
- Winters M, Friesen MC, Koehoorn M, Teschke K. Utilitarian Bicycling: A Multilevel Analysis of Climate and Personal Influences. American Journal of Preventive Medicine. 2007 Jan;32(1):52–8.
- Winters M, Teschke K, Grant M, Setton EM, Brauer M. How Far Out of the Way Will We Travel? Built Environment Influences on Route Selection for Bicycle and Car Travel. Transp Res Record. 2010(2190):1-10.
- Winters M, Teschke K. Route Preferences Among Adults in the Near Market for Bicycling: Findings of the Cycling in Cities Study. American Journal of Health Promotion. 2010b; 25:40–7.
- Wolin KY, Yan Y, Colditz GA, Lee I-M. Physical activity and colon cancer prevention: a meta-analysis. Br J Cancer. 2009 Feb 10;100(4):611–6.
- Woodcock J, Franco OH, Orsini N, Roberts I. Non-vigorous physical activity and all-cause mortality: systematic review and meta-analysis of cohort studies. International journal of epidemiology. 2011;40(1):121-38.
- World Health Organization (WHO). Health Co-Benefits of Climate Change Mitigation: Transport sector [Internet]. 2011a. Available from: http://www.who.int/hia/green_economy/en/index.html
- World Health Organization. HEAT: Health economic assessment tool [Internet]. 2011b [cited 2011 Oct 2];Available from: http://www.heatwalkingcycling.org/
- World Health Organization (WHO). World report on road traffic injury prevention. 2004
- (Available from: http://www.who.int/violence_injuryprevention)
- Yancey AK, Ory MG, Davis SM. Dissemination of Physical Activity Promotion Interventions in Underserved Populations. American Journal of Preventive Medicine. 2006 Oct;31(4, Supplement):82–91.
- Zegeer C, Stewart J, Huang H, Lagerwey P. Safety Effects of Marked Versus Unmarked Crosswalks at Uncontrolled Locations: Analysis of Pedestrian Crashes in 30 Cities. Transportation Research Record: Journal of the Transportation Research Board. 2001 Jan 1;1773(-1):56–68.

- Zegeer CV, Opiela KS, Cynecki MJ. Effect of pedestrian signals and signal timing on pedestrian accidents. Transportation Research Record [Internet]. 1982 [cited 2011 Nov 22];(847). Available from: http://trid.trb.org/view.aspx?id=182187
- Zegeer CV, Stutts JC, Huang H, Zhou M, Rodgman E. Analysis of elderly pedestrian accidents and recommended coutnermeasures. Transportation Research Record [Internet]. 1993 [cited 2011 Nov 15];(1405). Available from: http://trid.trb.org/view.aspx?id=384454
- Zuurbier M, Hoek G, Oldenwening M, Lenters V, Meliefste K, van den Hazel P, et al. Commuters' exposure to particulate matter air pollution is affected by mode of transport, fuel type, and route. Environ Health Perspect. 2010;118(6):783-9. Epub 2010/02/27.

Appendix A. Methodology for quantifying health benefits

A.1. Overview of health benefits and quantification tools

Calculating mortality reduction: WHO Health Economic Assessment Tools

HEAT for cycling is designed for adult populations age 20-64. The relative risk estimate was derived from three combined cohort studies that controlled for leisure time physical activity as well as for socioeconomic variables (age, sex, smoking, etc.) The studies yielded a relative risk of all-cause mortality of 0.72 for cycling 3 hours per week, 36 weeks per year, at 14 km/h. This means that, in any given year, regular cyclists were 28% less likely to die from any cause than non-cyclists with similar characteristics and levels of leisure time activity.

HEAT for walking is designed for adult populations aged approximately 20-74 years. This is the age range for which the relative risk estimate used by the tool is applicable. The tool applies to walking at a pace that achieves health benefits (of at least 4.8 km/hour) and is designed for habitual behaviour rather than for one-time walking events. The relative risk estimate was derived from fifteen studies that reported associations between walking and reduction in all-cause mortality. For a walking exposure of 29 minutes seven days per week, aggregated relative risk for all cause mortality was 0.78 (0.64-0.98 with a 95% confidence interval).

Both tools use a linear dose-response relationship between walking/cycling and mortality across the range of likely behaviours. For example, people who cycle 1.5 rather than 3 hours per week are 14% less likely to die from any cause, rather than 28%. To avoid inflated estimates at high levels of walking and cycling, risk reduction reaches a maximum of 50% in HEAT. This is based on currently available studies that conclude that regular physical activity usually does not reduce all-cause mortality by more than 50%.

The HEAT tools for walking and cycling use the following input data on walking and cycling:

- An estimate of how many people are walking or cycling. We include only walking and cycling by residents 20-74 years of age for walking, and 20-64 years of age for cycling.
- An estimate of the average annual duration or distance spent walking or cycling. Distances are converted to durations based on assumed paces of 4.8 km / hour (walking) and 14 km / hour (cycling). Again, we include only walking and cycling by residents 20-74 years of age for walking, and 20-64 years of age for cycling.

The HEAT tools also require the mortality rate, value of a statistical life, and time period of benefits. We use the following inputs:

- Mortality rate (age 20-74). We were unable to find Canada- or Toronto-specific mortality rates for age 20-74. However, we did find World Health Organization data for all countries on mortality for adults age 15-60. For this age range, Norway's mortality rate is very close to Canada's (WHOSIS 2011). We thus use Norway's mortality rate of 224.98 per 100,000 adults age 20-74 (provided within the HEAT tools).
- Time period of benefits and discount rate. We calculate benefits over a single year. The discount rate is therefore not relevant.

A.2. Toronto data sources

2006 Transportation Tomorrow Survey. The Transportation Tomorrow Survey uses telephone interviews that cover 5% of the population to gather data regarding weekday travel patterns of Torontonians. The Survey uses interviews with one members of a household to record trips made by all members of a household over the previous 24 hours. The Transportation Tomorrow Survey uses telephone interviews that cover 5% of the population to gather data regarding weekday travel patterns of Torontonians. The Survey uses interviews with one members of a household to record trips made by all members of a household over the previous 24 hours. The survey covered 129,273 people age 12 and over, and these results were expanded to represent all 2,445,937 Toronto residents age 12+. The 2006 survey was conducted for weekdays in the months of September, October, November, December and January, excluding the holiday season, plus one week in May. The four major reasons that walking and cycling are underestimated in the 2006 TTS are as follows:

- TTS data aims to exclude walking trips other than to work and to school. Thus, TTS severely underestimates both the number of trips and the number of people making trips. However, some trips are captured by interviewers and included in the database.
- The survey is conducted in the fall and winter season and thus fails to capture the large number of Torontonians who only cycle in the summer and / or in good weather.
- "Discretionary" trips and walking and cycling trips are generally under-reported in oral surveys such as the TTS. A comparison of trip diaries and TTS oral surveys in Toronto indicated that oral surveys exclude almost half of "non home-based" trips, and 12% of "other" home-based trips (Halcrow 2008). Modes with smaller mode shares are also generally are under-counted by oral surveys such as the TTS (Halcrow 2008). These issues are further aggravated by the TTS' reliance on a single household member to report on the trip making of all household members (Halcrow 2008).
- The sample lists used for the 2006 TTS include only persons living in households with listed telephone numbers. Residents age 18-24 have been consistently under-represented since 1996 as they substitute cell phones for landlines and do not appear on sample lists (Halcrow Consulting 2008).

2010 City of Toronto Downtown Bicycle Count. In 2010, the City conducted its first downtown bicycle count. This Count provides the first data on how many cyclists are riding on downtown streets, when and where they are riding, and other characteristics about cyclists such as helmet use, gender, sidewalk riding, and whether the cyclist is transporting a passenger. The Bicycle Count used screenlines that were selected to capture the large number of bike trips that begin and end within the central area of Toronto (Figure 1). The four screenlines that were used for the Bicycle Count are Bloor Street in the north, Spadina Avenue in the west, Queen's Quay Boulevard in the south, and Jarvis Street in the east (City of Toronto Transportation Services 2010). A map of the screenlines is included below.

Manual bicycle counts were collected by Ontario Traffic Inc. contract staff along the inside of the cordon at 34 locations (Figure 4), from September 13th to October 1st, 2010. The number of people riding a bicycle was counted at each location for one 12-hour period (7:00 AM to 7:00 PM) in 15-minute intervals, on one day when there was no precipitation. People on e-bikes were included in the count (City of Toronto Transportation Services 2010).

2008 Toronto Walking Survey and 2009 Toronto Bicycle Survey. The 2008 Walking Survey conducted by Harris-Decima (2008) included 1,000 Torontonians age 16 and over, fairly evenly divided between Toronto's districts (Central Toronto-East York, Etobicoke-York, North York and Scarborough). The 2009 Bicycle Survey conducted by Ipsos-Reid included 1,000 Torontonians age 15 and over, equally divided between Toronto's four districts.

A.3. Methodology used to estimate walking and cycling

We include three types of trips in our estimation of current levels of walking: trips to work, trips to school and trips to shopping and other destinations. Though trips to shopping and other destinations are not included in the TTS, the Walking Survey suggests that they are very important. We thus estimate walking patterns for three groups: people who walk to work and to other destinations, people who walk to school and to other destinations, and people who walk only to other destinations.

We include two types of trips in our estimation of current levels of walking: trips to work, and all other trips. We assume that only commuter cyclists travel close to year-round; for cyclists making other trips (including trips to school and occasional trips to work) we estimate a shorter cycling season. We thus estimate cycling patterns for two groups: people who cycle to work and to other destinations for a significant part of the year, and "seasonal cyclist" who cycle to a variety of destinations but only in the summer months.

We assume conservative average walking speeds of 4.8 km/h and cycling speeds of 15 km/h.

Walking and cycling to work

Number of people walking and cycling to work: The 2006 census provides information on the number of Torontonians who usually walk and cycle to work: 81,565 and 19,780, respectively.

Distance walked and cycled to work: Both the 2006 TTS and the 2008 / 2009 Walking / Bicycle surveys provide information on the average distance of walking and cycling trips to work. We take the more conservative of these two estimates. For walking, the survey-based average walking time of 12 minutes (1 km) is used rather than the TTS value of 1.3 km. For cycling, the TTS-based average cycling distance to work of 5.2 km is used rather than the survey-based average of 26 minutes (6.5 km).

Travel days per week: We assume that people who report "usually" walking and cycling to work (in the 2006 census) do so 4 days per week. This is a conservative estimate: on average, Torontonians travel to work 4.7 days per week (Harris-Decima 2008). In the City of Calgary, commuter cyclists report cycling to work 4.1 days per week, on average. We also assume that this group of commuters walks or cycles to shopping or to another destination once each week. This is further described under *Other walking trips* below.

Weeks per year of walking and cycling to work: We estimate that Torontonians who walk to work do so year-round except for two weeks of vacation (50 weeks per year). We estimate that Torontonians who cycle to work do so 36 weeks per year, based on the City of Calgary's 2006 survey of downtown cycle commuters. They report commuting 145 days per year on average, which equates to 4 days per week, 36 weeks per year.

Walking to school

Number of people walking to school and travel days per week: The 2006 TTS provides data on the number of people who walked to school on a given weekday. For the calculation of mortality reduction, we count only the Torontonians aged 20-74 who walked to school. We estimate that the students in this age group travel to school 3 days each week – less than the average of 3.9 times per week reported for all ages, as they often attend university or college (Harris-Decima 2008).

Combining this travel frequency with the 10,184 Torontonians who walk to school in a single-day based on the TTS, we calculate that 16,973 Torontonians aged 20-74 walk to school in a given week.

We also assume that this group of mature students walks to shopping or to another destination once each week. This is further described under *Other walking trips* below.

Distance walked to school: We use the average walking time of 8.9 min (0.7 km) from the walking survey rather than the TTS average distance of 1.1 km.

Weeks per year of walking to school: We estimate that all students walk to school 32 weeks per year.

Other walking trips

Number of people who walk to shopping and other destinations: The 2008 walking survey indicates that more than twice as many Torontonians use walking as the main mode of transportation to shopping as compared to work. 17% of survey respondents reported walking to shopping as their main mode of transportation, while only 7.3% reported walking to work as a main mode. Using this greater than 2:1 ratio, we estimate that 180,100 Torontonians regularly walk to shopping.

However, some of these people also walk to work and to school. We make the very conservative assumption that all of the people who walk to work and to school also walk to shopping. This still leaves over 80,000 people who only walk to shopping.

Travel frequency: We estimate that the people who walk only to shopping and for other discretionary trips travel twice each week. This is a very conservative estimate; in the Walking Survey, Torontonians report traveling to shopping 2.5 times per week on average, and to leisure activities 2.3 times per week.

We estimate that people who walk to school or to work make only one additional walking journey to shopping or leisure activities. (The second trip may be combined with a trip to or from work or school.)

Seasonal cycling (all other cycling trips)

Number of seasonal cyclists. We estimate seasonal cycling based on the Downtown Bicycle Count in combination with TTS data on relative cycling in the downtown vs. other areas.

The City of Toronto's 2010 bicycle screenline count recorded 19,162 cyclists entering the downtown core area depicted in Figure 28 below. We estimate that 40% of all City of Toronto cyclists that would be entering the downtown core using TTS data on the distribution of trips between residents of different planning districts. This is undoubtedly a conservative estimate as we assume that ALL trips made by residents of former Planning Districts 1 and 6 (map below) are to the downtown core. As seen in Figure 1, the boundaries of Planning Districts 1 and 6 extend far beyond the boundaries of the bicycle screenline count. According to the TTS, residents of Planning Districts 1 and 6 made roughly 38% of all cycling trips over a 24-hour period. (We assume that the relative number of cyclists in these planning districts is proportional to the number of trips.)

We subtract the number of regular cyclists from the citywide total to estimate the number of seasonal cyclists, and use data from the TTS to estimate that number of these seasonal cyclists that are 20-64 years of age. The bicycle count indicates that weekend cycling is about 45% lower than weekday cycling, so we estimate that the weekly volume is 6 times the single day weekday volume of 19,602. Number of "seasonal" cyclists = Total cyclists – commuter cyclists

Total cyclists on a single day = 19,162 inbound downtown / 38% = 50,426

Total seasonal cyclists on a single day = 50,426 - 19,780 = 30,646

Weekly volume = 28,582 * 6 = 183,878

Total seasonal cyclists if cyclists ride 3 days per week = approx. 61,292

We recognize that many assumptions underlie this estimate of occasional cycling. However, given the 2009 survey that suggests that 29% of all Torontonians age 16+ are utilitarian cyclists, we nonetheless feel that this is a conservative estimate of cycling in Toronto.

Travel days per week, and weeks per year. We use responses from the 2009 bicycle survey to estimate that these occasional cyclists not captured by the TTS only ride 3 days per week. This is a very conservative estimate, as Toronto cyclists report cycling to shopping alone an average of 2.6 days per week. In addition, 11% of all respondents reported cycling to work and to school and most of these cyclists are not captured by the census data on cycling to work (Ipsos-Reid 2009).

Distance cycled. We use the TTS average cycling trip distance of 3.9 km for seasonal cycling. This is a shorter distance than the average travel times suggested the 2009 Bicycle Survey.



Figure 28: Downtown bicycle screenline count (left) and Planning district 1 (right) used to estimate seasonal cycling

	Walking to work and shopping	Walking to school and shopping	Walking to shopping only	Cycling to work and shopping	Seasonal cycling	Total
Number of people	81,565	16,973	81,565	19,780	61,292	260,167
Distance per trip	1.0 km	0.7 km	1.0 km	5.2 km	3.9 km	
One-way trips per week	12	8	4	10	6	
Weeks per year	50	32	50	36	22	
Distance per year	600 km	179 km	200 km	1872 km	515 km	
Percent reduction in risk of all-cause mortality	16%	5%	5%	33%	11%	
Number of deaths prevented each year	48.5	3.2	17.1	24.9	24.4	118
Willingness to pay value ¹	\$196 million	\$13 million	\$69 million	\$101 million	\$99 million	\$478 million
Discounted future earnings value ²	\$53 million	\$4 million	\$19 million	\$27 million	\$27 million	\$130 million

Table 10: Reductions in all-cause mortality from regular walking and cycling in Toronto

1. Based on Sawyer et al.'s (2007) Value of a Statistical Life (\$4.05 million CAD)

2. Based on Vodden *et al.*'s (2007) average Discounted Future Earnings of Ontarians who are killed in motor-vehicle collisions (\$1.1 million CAD)

Appendix B. Analysis of types of collisions

Table 11: Analysis of cyclist-vehicle collisions in Toronto (5-year average, 2005-2009)Source: City of Toronto 2011b

Type of cyclist-vehicle collision	Number of collisions	% of collisions
Sideswiping collisions	169	15%
Collisions with car doors	136	12%
Rear-ending collisions	45	4%
Sub-total: Separation of cyclists and vehicles	350	31%
Motorists turning left	151	13%
Motorists turning right at signalized intersections	145	13%
Motorists turning right at un-signalized intersections	100	9%
Cyclists turning left	31	3%
Sub-total: Turns at intersections	427	37%
Total collisions	1145	100%

Table 12: Analysis of pedestrian-vehicle collisions in Toronto (5-year average, 2005-2009)**Source**: City of Toronto 2011a

Type of pedestrian-vehicle collision	Number of collisions	Fatalities (2009)	% of collisions
Vehicle turning left, pedestrian ROW	444	8	20%
Vehicle turning right, pedestrian ROW	231	2	10%
Vehicle going straight, pedestrian ROW	129	2	6%
Vehicle going straight, pedestrian without ROW	239	9	11%
Sub-total: Safety at intersections	1043	21	47%
Pedestrian hit mid-block	347	3	16%
Pedestrian hit at PXO	52	1	2%
Sub-total: Crosswalks	399	4	18%
Pedestrian on shoulder or sidewalk	114	1	5%
Total collisions	2,222	31	100%

Appendix C. Toronto policies and programs

Active transportation in the City of Toronto Official Plan

The City of Toronto's Official Plan highlights walking and cycling as a key element of the campaign for next generation transportation. The overall aim of the section entitled *A Progressive Agenda for Transportation Change* is to "provide the widest range of sustainable transportation options that are seamlessly linked, safe, convenient, affordable and economically competitive". Corresponding principles include:

- Promoting land use development and urban form that lead to fewer and shorter trips, and
- Instituting planning, traffic engineering and street design practices that encourage walking and cycling.

The Official Plan also highlights the role of walking and cycling in the vision of creating an attractive and safe city that evokes pride, passion and a sense of belonging. The Plan emphasizes the importance of walking and cycling in all parts of the City, including employment districts, avenues, centres, regeneration areas and priority neighbourhoods.

City of Toronto Walking Strategy and Pedestrian Charter

The Walking Strategy aims to build a physical and cultural environment that supports and encourages walking, including vibrant streets, parks, public squares and neighbourhoods where people will choose to walk more often:

- Encourages seamless integration with public transit and cycling
- Keep sidewalks clear, accessible and easy to navigate
- High-quality walking environment will be a consideration in public and private building projects
- Signage and maps designed to make walking easy and enjoyable
- Transforming neighbourhoods in places where people want to walk where needed
- Pedestrian-focused projects and initiatives

http://www.toronto.ca/transportation/walking/walking_strategy.htm

The Toronto Pedestrian Charter sets out six principles to ensure that walking is a safe and convenient mode of urban travel:

- accessibility
- equity
- health and well-being
- environmental sustainability
- personal and community safety
- community cohesion and vitality

http://www.toronto.ca/transportation/walking/charter.htm

City of Toronto Bike Plan

The Toronto Bike Plan sets out integrated principles, objectives and recommendations regarding safety, education and promotional programs as well as cycling related infrastructure, including a comprehensive bikeway network. "The vision for the Toronto Bike Plan is to create a safe, comfortable and bicycle friendly environment in Toronto, which encourages people of all ages to use bicycles for everyday transportation and enjoyment. The primary goals of the TBP are:

- to double the number of bicycle trips made in the City of Toronto, as a percentage of total trips by 2011; and
- to reduce the number of bicycle collisions and injuries.

The "six spokes" model used by the Toronto Bike Plan has inspired the planning efforts of numerous other cities (Interview). The six areas of focus are as follows:

- Bicycle friendly streets
- Bikeway network
- Safety and education
- Promotion
- Cycling and transit
- Implementation and evaluation

http://www.toronto.ca/cycling/bikeplan/

Active transportation in "The Big Move"

"The Big Move" is the Draft Regional Transportation Plan (RTP) for the Greater Toronto and Hamilton Area (GTHA) that provides a vision, goals and objectives for seamless, coordinated, efficient, equitable and user-centred transportation within the region. Eight of the fifteen strategic directions relate to active transportation:

- SD#1 Build a Comprehensive Regional Rapid Transit Network
- SD#2 Promote Active Transportation
- SD#4 Consider all Modes of Transportation
- SD#7 Implement an Integrated Transit Fare System
- SD#8 Build Communities That are Pedestrian, Cycling and Transit-Supportive
- SD#9 Develop a System of Mobility Hubs
- SD#10 Focus Growth and Development Along Transportation Corridors
- SD#12 Plan For Universal Access

http://www.metrolinx.com/thebigmove/en/

Bikeway network status

The Bikeway Network is a part of the City of Toronto Bike Plan. Table 13 below presents the status update available on the city's webpage.

Table 13: Status of the Bikeway Network

Bikeway type	Planned*	Existing**	Percentage of goal reached
Bike lanes	495	111	~24%
Shared roadways (local street bikeways and sharrows)	260	150	56%
Off-road paths	249	168	68%
Total	1,004	429	43%

* CITY OF TORONTO BIKE PLAN, 2001

** UPDATED NOVEMBER 30, 2010

Bicycle parking and storage infrastructure status

With currently over 16,000 parking spots, the City of Toronto's Post and Ring bike parking program is a leader in North America and new bike parking stands are installed every year.

The City of Toronto currently has 202 bicycle lockers in fifteen locations. They provide excellent, secure bicycle parking by offering protection from theft, vandalism and inclement weather. The lockers can be rented for a monthly fee and are designed to hold one bicycle plus gear such as panniers, locks, lights, etc.

Toronto's first bicycle parking station is in operation at Union Station. It offers secure, indoor bicycle parking to protect the bicycles of registered participants from theft and vandalism. The location at Union Station maximizes the opportunity for cyclists to use their bikes in combination with transit, as it is a hub for GTA transit. Two future bicycle parking stations are planned for Victoria Park Subway Station and Toronto City Hall.

http://www.toronto.ca/cycling/parking/index.htm

Toronto Green Standard

The Toronto Green Standard (TGS) establishes mandatory requirements related to sustainable site and building design for new development (Tier 1 performance measures). Developers can also receive incentives for achieving the optional (Tier 2) performance measures. Measures related to active transportation include:

- Cycling infrastructure—provision of short-term bicycle parking, long-term bicycle storage, and shower and change facilities
- Pedestrian infrastructure—provision of sun and weather protection for pedestrians, building integration with walking routes, grading and surface treatment, signage, lighting, and separation from vehicular routes and air intake or exhaust systems

• Transit accessibility—provision of building integration with transit facilities, proximity to transit service, and direct pedestrian linkage to transit

http://www.toronto.ca/planning/environment/greendevelopment.htm

Guidelines for the Design and Management of Bicycle Parking Facilities

The City of Toronto has developed these guidelines to provide specific information for planners, developers and property managers to use to improve the quality of bicycle parking that is secured through the City's development approval process. These guidelines will help new developments meet Official Plan policies and environmental targets of the Toronto Green Development Standard.

http://www.toronto.ca/planning/bicycle_parking_guide.htm

Bike boxes

The City of Toronto has recently started installing bike boxes at intersections to designate a space for cyclists to wait in front of cars at the red light, and to proceed first when the light turns green. Bike boxes increase the visibility and safety of cyclists and make left hand turns safer for cyclists.

http://www.toronto.ca/cycling/network/bikebox_faq.htm

Cycling Ambassadors

"The Cycling Ambassadors are a team of cycling experts who reach out to communities across Toronto with programs and campaigns to deliver safety messages and to encourage cycling." The Ambassadors are involved with the delivery of many cycling related initiatives, like the Bicycle Friendly Business Awards, Bike Month, the Bicycle User Group Network, and CAN-BIKE as well as providing information on cycling safety at community events and through giving safety seminars. Ambassadors participate in the evaluation process for the Toronto Bike Plan by collecting data from surveys and bicycle counts.

http://www.toronto.ca/cycling/ratsa/index.htm

Bicycle Friendly Business Awards

The Bicycle Friendly Business Awards recognize businesses and organizations that demonstrate leadership in encouraging cycling by employees and/or customers. The following are the award categories:

- Bike Parking
- Suburban Business
- Small Business
- Large Business
- Best Skills Development
- Best Overall

http://www.toronto.ca/cycling/bfba/index.htm

Toronto Bike Month

Toronto Bike Month involves dozens of events in June each year that strive to raise awareness of the benefits of cycling and encourage people to ride their bikes more. Events are hosted by the City of Toronto and other partners and include the Toronto Group Commute and pancake breakfast.

http://www.toronto.ca/cycling/bikemonth/index.htm

CAN-BIKE

CAN-BIKE courses teach cycling skills such as anticipating traffic dynamics, recognizing road hazards, and collision-avoidance techniques. Courses are offered for kids, novice cyclists, casual / recreational cyclists, and advanced / commuter cyclists. Employers are responsible for providing this safety training for all employees that are required to ride a bike to perform work duties.

http://www.toronto.ca/cycling/canbike/

BIXI Toronto

BIXI Toronto provides a network of bikes for public use throughout the downtown core. The BIXI bikes are intended to be used for one-way trips of less than 30 minutes and can be picked up or dropped off at any of the 80 bicycle docking terminals 24 hours a day, seven days a week. Users can choose to pay for single day, monthly, or annual use passes that allow free use of a bike for up to 30 minutes at a time, with fees charged for longer trips. BIXI Toronto was launched May 3, 2011 and logged over 325,000 trips, on 1,000 bikes by over 3,500 monthly or annual subscribers and over 35,000 single day 'casual users' within the first six months.

http://www.toronto.ca/cycling/bixi/

Ride the City Toronto

Ride the City Toronto is an online bike route planning tool that uses bikeway network information and user feedback to recommend safe cycling routes. It also illustrates the location of bike sharing pick-up locations and bicycle repair shops.

http://www.ridethecity.com/toronto

Green Communities Canada Active and Safe Routes to School

Green Communities Canada Active & Safe Routes to School promotes the use of active and efficient transportation for the daily trip to school, including walking and cycling by providing resources for participating schools.

Active & Safe Routes to School is a collaborative initiative between families, schools and community. This community approach helps to ensure children's safety when actively traveling to and from school.

Walk into Health Library Pedometer Lending Program

Toronto Public Health, Toronto Public Library and the City of Toronto Transportation Services offer the Walk into Health Pedometer Lending Program at 40 Toronto Public Library Branches across the City of Toronto. Residents can borrow a pedometer and receive a goal-oriented resource package to help them begin and maintain their walking behaviours. To date, approximately 12,000 pedometers have been borrowed. Plans are underway to expand the program to all library branches across the city.

Toronto Public Health's Walk into Health Program

Toronto Public Health provides agency training and support to community agencies and workplaces interested in starting a walking program or initiative within their organization. The training incorporates the health benefits of walking, training of walking leaders/guides and the provision of educational resources to help participants begin and maintain their walking behaviours. Pedometer kits are also available on loan from Toronto Public Health for organizations implementing a walking program. In addition, the program promotes both the Explore Toronto's Parks & Trails and Discovery Walks maps.
Appendix D. 2008 Toronto Road Classification System Street Name Index

http://www.toronto.ca/transportation/road_class/pdf/city-wide_index.pdf